

Scholars' Mine

Doctoral Dissertations

Student Theses and Dissertations

Summer 2021

Proper orthogonal decomposition: New approximation theory and a new computational approach

Sarah Katherine Locke

Follow this and additional works at: https://scholarsmine.mst.edu/doctoral_dissertations

Part of the Applied Mathematics Commons Department: Mathematics and Statistics

Recommended Citation

Locke, Sarah Katherine, "Proper orthogonal decomposition: New approximation theory and a new computational approach" (2021). *Doctoral Dissertations*. 3012. https://scholarsmine.mst.edu/doctoral_dissertations/3012

This thesis is brought to you by Scholars' Mine, a service of the Missouri S&T Library and Learning Resources. This work is protected by U. S. Copyright Law. Unauthorized use including reproduction for redistribution requires the permission of the copyright holder. For more information, please contact scholarsmine@mst.edu.

PROPER ORTHOGONAL DECOMPOSITION: NEW APPROXIMATION THEORY AND A NEW COMPUTATIONAL APPROACH

by

SARAH KATHERINE LOCKE

A DISSERTATION

Presented to the Graduate Faculty of the

MISSOURI UNIVERSITY OF SCIENCE AND TECHNOLOGY

In Partial Fulfillment of the Requirements for the Degree

DOCTOR OF PHILOSOPHY

in

MATHEMATICS

2021

Approved by:

John R. Singler, Advisor Jason Murphy Xiaoming He Yanzhi Zhang Lian Duan

Copyright 2021

SARAH KATHERINE LOCKE

All Rights Reserved

ABSTRACT

Proper orthogonal decomposition (POD) projection errors and error bounds for POD reduced order models of partial differential equations have been studied by many. In this research we obtain new results regarding POD data approximation theory and present a new difference quotient (DQ) approach for computing the POD modes of the data.

First, we improve on earlier results concerning POD projection errors by extending to a more general framework that allows for non-orthogonal POD projections and seminorms. We obtain new exact error formulas and convergence results for POD data approximation errors, and also prove new pointwise convergence results and error bounds for POD projections. We consider both the discrete and continuous cases of POD within this generalized framework. We also apply our results to several example problems, and show how the new results improve on previous work.

Next, we consider the relationship between POD, difference quotients (DQs), and pointwise ROM error bounds. It is known that including DQs is necessary in order to prove optimal pointwise in time error bounds for POD reduced order models of the heat equation. We introduce a new approach to including DQs in the POD procedure to further investigate the role DQs play in POD numerical analysis. Instead of computing the POD modes using all of the snapshot data and DQs, we only use the first snapshot along with all of the DQs and special POD weights. We show that this approach retains all of the numerical analysis benefits of the standard POD DQ approach, while using a POD data set that has half the number of snapshots as the standard POD DQ approach, i.e., the new approach is more computationally efficient. We illustrate our theoretical results with numerical experiments.

ACKNOWLEDGMENTS

First, I would like to thank my advisor, John Singler. He has taught me so much, and I could not have asked for a better research advisor. His hard work, dedication, and genuine support made me a better student, researcher, and person.

I would also like to thank the mathematics department at Missouri S&T and my committee members for contributing to my graduate school journey. I was able to learn a great deal from them.

I am grateful for the support of the Chancellor's Distinguished Fellowship for the first four years of my time in graduate school. Without it, I might not have attended this wonderful institution.

I would also like to extend a special thank you to the faculty, staff, and students of the University of Tennessee at Martin Mathematics department for all of their help preparing me for graduate school and their continued support throughout the last five years.

I am so appreciative of the love and encouragement from my parents, brother, and extended family. They have been so supportive of me in continuing my education.

Finally, and most importantly, I would like to thank my now husband, Jake. We might have decided to get married during a global pandemic that also corresponded to my final year in graduate school, but I wouldn't trade the experience for the world. He was there for all of the long days and late nights that led to the completion of this dissertation, and I would not have been able to do it without his love and support.

TABLE OF CONTENTS

V

AI	BSTRA	ACT		iii		
AC	CKNO	WLED	GMENTS	iv		
LI	ST OF	F ILLUS	TRATIONS	vii		
LI	ST OF	F TABL	ES	viii		
SE	CTIO	N				
1.	INTI	RODUC	TION	1		
2.	BACKGROUND					
	2.1.	FUNC	TIONAL ANALYSIS BACKGROUND	6		
	2.2.	PROPI	ER ORTHOGONAL DECOMPOSITION	9		
		2.2.1.	Discrete Case	9		
		2.2.2.	Continuous Case	11		
		2.2.3.	Optimality of POD	12		
3.	NEW POD APPROXIMATION THEORY					
	3.1.	ASSU	MPTIONS AND NOTATION	18		
		3.1.1.	Discrete Case	18		
		3.1.2.	Continuous Case	19		
	3.2.	POD P	ROPERTIES	21		
		3.2.1.	Non-orthogonal POD Projections	21		
		3.2.2.	POD Singular Values and POD Eigenvalues	25		
	3.3.	ERRO	R FORMULAS	28		
		3.3.1.	Discrete Case	31		
		3.3.2.	Continuous Case	33		

	3.4.	POINT	TWISE CONVERGENCE OF POD PROJECTIONS	35
	3.5.	EXAM	IPLES	45
		3.5.1.	Computational Example	45
		3.5.2.	Examples From [1]	47
			3.5.2.1. Example 1	48
			3.5.2.2. Example 2	51
		3.5.3.	Non-orthogonal Projection Example	54
4.	A N	EW API	PROACH TO POD WITH DIFFERENCE QUOTIENTS	56
	4.1.	REVIE	EW OF EXISTING METHODS	56
		4.1.1.	Standard POD	57
		4.1.2.	POD with Difference Quotients	58
	4.2.	A NEV	W APPROACH	61
		4.2.1.	Preliminary Computations	64
		4.2.2.	Pointwise Error Bounds	66
	4.3.	REDU	CED ORDER MODELING	69
		4.3.1.	POD Space: $L^2(\Omega)$	71
		4.3.2.	POD Space: $H_0^1(\Omega)$	76
		4.3.3.	Optimality	77
	4.4.	NUME	ERICAL RESULTS	81
		4.4.1.	ROM Comparisons for the Three POD Approaches	81
		4.4.2.	Pointwise Error Bounds	83
		4.4.3.	ROM Error Bounds	84
5.	CON	ICLUSI	ONS	87
RE	EFERI	ENCES		89
VITA				

LIST OF ILLUSTRATIONS

Figure				
4.1.	Plot of singular values of POD operators using different numbers of finite element nodes	. 65		

viii

LIST OF TABLES

Table		Page
3.1.	Error Comparison with $r = 4$	47
3.2.	Error Comparison with $r = 12$	47
4.1.	Actual Error vs. Error Formulas from Lemma 4.2.2 for New DQ POD	66
4.2.	ROM errors for the L^2 Norm at the final time	82
4.3.	ROM errors for the H_0^1 Norm at the final time	82
4.4.	ROM solution norm error Equation (4.68)	83
4.5.	Scaling factors for Theorem 4.2.4 as Δt changes with $X = L^2(\Omega)$	84
4.6.	Scaling factors for Theorem 4.2.4 as Δt changes with $X = H_0^1(\Omega)$	84
4.7.	Scaling factors as Δt changes for ROM errors	85

1. INTRODUCTION

Proper orthogonal decomposition (POD) is a model order reduction technique for partial differential equations (PDEs) and other mathematical models. With this method, modes are computed from simulation or experimental data and a Galerkin projection is used with these modes to reduce the model. Because POD reduced order models often have very low dimension, they can be used to efficiently simulate computationally demanding problems. Therefore, POD has been used in many fields of study including fluid dynamics [2, 3, 4, 5, 6, 7, 8, 9, 10] and control theory [11, 12, 13]. For more information about POD and many known results, see, e.g., [14, 15, 16, 17]. Because of the wide use of POD in many application areas, it is of great interest to study the approximation errors in POD model order reduction procedures. Numerical analysis results for POD reduced order models of PDEs were first obtained by Kunisch and Volkwein [18, 19], and then by many others; see, e.g., [1, 20, 21, 22, 23, 24, 25, 26, 27, 28, 29, 30, 31, 32, 33, 34] and the references therein.

In this work we obtain new results regarding POD data approximation theory and present a new difference quotient (DQ) approach for computing the POD modes of the data.

Understanding POD data approximation errors is typically important for these numerical analysis works. To see this, let *w* be the solution of the mathematical model, let w_r be the solution of the POD reduced order model, and let π_r be a projection onto the span of the first *r* POD modes. Split the error as

$$w - w_r = \rho_r + \theta_r, \quad \rho_r = w - \pi_r w, \quad \theta_r = \pi_r w - w_r.$$

Energy estimates can often be used to bound θ_r by quantities including various norms of ρ_r , the POD data approximation error for that projection.

In [1], exact error formulas and convergence results were proven for norms of ρ_r involving two Hilbert spaces, where one space is a subset of the other. In that work, Singler considered the continuous POD setting and proved results for different combinations of POD spaces, projections, and norms. Shortly after [1], Iliescu and Wang [25] provided analogous error formulas for the discrete POD case, and many of the recent numerical analysis works mentioned above use results from [1, 25] or extensions of these results to other scenarios.

As POD is increasingly applied in a variety of situations, it becomes more useful to have error results that can be easily applied in a wide range of scenarios. Therefore, in Section 3 we extend POD data approximation results in [1, 25] to a generalized framework that allows us to treat non-orthogonal POD projections and seminorms. We prove new error formulas and convergence results for norms of quantities involving $\rho_r = w - \pi_r w$ with various POD projections π_r . We also prove new pointwise convergence results for different POD projections. Non-orthogonal POD projections have been used in the numerical analysis for POD reduced order models [24, 27]; however, the exact POD data approximation error formulas and convergence results obtained here are new. Exact POD data approximation errors using various seminorms have been obtained in some cases (see, e.g., [35, Section 3.3], [28, Lemma 3.1]); the general extension and convergence results in this work are new. Finally, some pointwise convergence results for POD projections were obtained in [1]; we obtain new error bounds and improved convergence results here.

The POD data approximation error formulas presented here are exact and do not require the use of POD inverse inequalities. We consider both the discrete and continuous cases for POD and generalize the setting in [1, 25] to allow a linear mapping between two Hilbert spaces to act on the data. We require minimal assumptions on the data, the linear operator, and the Hilbert spaces; the assumptions we do require are naturally satisfied in many applications and allow us to obtain convergence results even in the fully continuous case when the data has infinitely many positive POD eigenvalues. Note that most of the proof strategies in this work are new; some proofs do rely on techniques from [1, 36].

As mentioned above the widespread use of POD in applications has caused many researchers to study POD ROMs from a numerical analysis perspective. In order for POD to be beneficial for applications, researchers must understand how the various errors involved behave. To fully understand this for PDEs, three types of error must be considered: spatial discretization error, time discretization error, and ROM discretization error. The optimality of these errors is of particular concern. POD numerical analysis papers tend to focus on the time discretization error and the ROM discretization error since the spatial discretization error can typically be handled using existing techniques. For more information on these three types of error and the numerical analysis of POD, see the introduction of the recent work [37]. In this thesis, we focus only on the POD ROM errors which leads to an improved understanding of the numerical analysis of POD ROMs, particularly in regards to difference quotients and pointwise error bounds.

In Section 4 we focus on various approaches to creating POD modes from the data. Two of the most common existing methods are considered, and we introduce a new method. The first existing approach is a standard method to compute the POD modes and uses only the data, and the second existing approach utilizes both the data and the DQs of the data. Researchers originally started including DQs in the POD calculations to improve the numerical analysis results for POD reduced order models as in [18]. For other numerical analysis results for POD reduced order models see [19, 38, 39, 40, 41]. Further, DQs have been used in many reduced order modeling applications including feedback control for PDEs [42], subdiffusion equations [41], modeling the dynamics of a spiking neuron [40], and partial-integro-differential equations in financial modeling [43]. For a variety of additional applications see [31, 44, 45, 46].

Researchers have been curious about the role DQs play in the behavior of the ROM and whether or not they should be included in the POD computations. In general, results on this topic were inconclusive. However in 2014, substantial progress was made by Iliescu and Wang in [38] towards understanding this. In [38] a notion of optimality was introduced and their results strongly suggested that DQs are needed to achieve optimal pointwise-in-time convergence rates. Recently in [37], further progress was made. In this work it is shown that a critical assumption often made when using standard POD without DQs is automatically guaranteed to be satisfied when DQs are included with the data. The notion of optimality introduced in [38] is extended, and it is shown that including difference quotients results in optimal POD projection errors and ROM errors. The second primary goal of the thesis is to further investigate and understand pointwise error bounds in the POD-ROM setting.

To do this, in Section 4 we introduce a new approach to deriving POD modes from the data. When using all of the data with all of the DQs, the resulting data set is linearly dependent, i.e. the data set being used contains redundant information. This also leads to more costly POD basis computations compared to standard POD without DQs. In order to improve this situation, we consider the following question: *Can we obtain all of the same numerical analysis benefits of using DQs with POD using a data set without redundancy?* We show that the answer is yes, if we choose the data set and POD weights in a correct way. For our new approach to using DQs with POD uses a data set without redundancy in the following sense: if the original set of M snapshots is linearly independent, then the data set used in our new DQ approach has dimension M and is also linearly independent. Using this new collection of data and special POD weights, we are not only able to approximate the DQs and the one regular snapshot, but all of the other regular snapshot data as well. With this method we also prove that we retain the numerical analysis benefits that come with having the DQs in the POD data set.

The material in this dissertation is mostly from the works [47] and [48]. Some small changes have been made to increase readability and unify information, notation, and results between the works.

2. BACKGROUND

In this section, we recall some functional analysis background material, the basic theory for discrete POD and continuous POD as well as results on the optimality of POD. For details and proofs for the basic discrete and continuous POD theory, see, e.g., [14, 15, 19, 49, 50, 51] and also Section 2.2.3.

2.1. FUNCTIONAL ANALYSIS BACKGROUND

Let *V* and *W* be Hilbert spaces with inner products¹ $(\cdot, \cdot)_V$ and $(\cdot, \cdot)_W$ and corresponding norms $\|\cdot\|_V$ and $\|\cdot\|_W$. Throughout this work, the scalar field \mathbb{K} for all spaces is either $\mathbb{K} = \mathbb{R}$ or $\mathbb{K} = \mathbb{C}$.

Linear Operators: Let $T : V \to W$ be a linear operator with domain $\mathcal{D}(T) \subset V$, range $\mathcal{R}(T) \subset W$, and null space ker $(T) \subset V$. The *rank* of T is the dimension of $\mathcal{R}(T)$. The operator T is *bounded* if $||Tv||_W \leq M||v||_V$ for all $v \in \mathcal{D}(T)$. Throughout this thesis, we only consider bounded operators $T : V \to W$ that are defined on the whole space, so $\mathcal{D}(T) = V$. For such a bounded operator $T : V \to W$, the usual operator norm is given by $||T|| = \sup\{||Tv||_W : v \in V, ||v||_V = 1\}$. We also consider unbounded linear operators that are not defined everywhere, so that $\mathcal{D}(T) \neq V$. The operator T is *closed* if its graph, $\mathcal{G}(T) = \{(v, w) : v \in \mathcal{D}(T), w = Tv\}$, is closed in $V \times W$. If T is bounded (and everywhere defined), then T is closed. If T is closed and invertible, then T^{-1} is closed.

Adjoint Operators: The *Hilbert-adjoint operator* $T^* : W \to V$ satisfies $(Tv, w)_W = (v, T^*w)_V$ for all $v \in \mathcal{D}(T)$ and $w \in \mathcal{D}(T^*)$. If T is bounded, then T^* exists, is unique, and is also bounded. If T is densely defined, then T^* exists, is unique, and is closed; in addition, if T is closed, then T^* is densely defined. If $T : V \to W$ is invertible, then we let

¹In this thesis, all inner products and sesquilinear forms are linear in the first argument and conjugate linear in the second argument.

 $T^{-*}: V \to W$ denote the Hilbert adjoint operator of the inverse $T^{-1}: W \to V$. We note for T^* to exist we need T bounded or densely defined, and for T^{-*} to exist we need T^{-1} bounded or densely defined. We note these assumptions when necessary.

The following basic result is important in this work.

Lemma 2.1.1. Let V and W be Hilbert spaces. If $T : V \to W$ is a bounded linear operator, then ker $(TT^*) = ker(T^*)$ and ker $(T^*T) = ker(T)$.

Proof. We only prove the first one. Let $w \in ker(TT^*)$. Then,

$$TT^*w = 0 \Rightarrow (TT^*w, w)_W = 0 \Rightarrow (T^*w, T^*w)_V = 0 \Rightarrow ||T^*w||_V^2 = 0 \Rightarrow T^*w = 0.$$

Next, let $w \in \ker(T^*)$. Then $T^*w = 0 \Rightarrow TT^*w = 0$.

Projections: A bounded linear operator $\Pi : V \to V$ is a *projection* onto $U = \mathcal{R}(\Pi)$ if $\Pi^2 = \Pi$. Then we have $\Pi v \in U$ for all $v \in V$ and $\Pi u = u$ for all $u \in U$. Also, Π is an *orthogonal projection* if $u = \Pi v \in U$ minimizes $\inf_{u \in U} ||v - u||_V$ for any $v \in V$. A nontrivial orthogonal projection Π is automatically self-adjoint, i.e., $\Pi^* = \Pi$, and satisfies $||\Pi|| = 1$. We consider non-orthogonal projections in this thesis, and therefore we do not assume a projection is orthogonal or self-adjoint unless explicitly specified. Sometimes, we assume a family of projections $\{\Pi_r\}$ is uniformly bounded in operator norm, i.e., there exists a constant *C* such that $||\Pi_r|| \leq C$ for all *r*.

The Singular Value Decomposition of a Compact Operator: If $T : V \to W$ is a compact linear operator, with separable Hilbert spaces V and W, then T has a *singular value decomposition* (SVD). The positive singular values of T are defined to be the square roots of the positive eigenvalues of the self-adjoint nonnegative compact operators $TT^* : W \to W$ and $T^*T : V \to V$. Further, the nonzero eigenvalues of these operators are equal, and we consider zero a singular value of T if either operator has a zero eigenvalue. If the ordered singular values of T are given by $\mu_1 \ge \mu_2 \ge \cdots \ge 0$ (including repetitions), the orthonormal basis of eigenvectors of TT^* is given by $\{\psi_k\} \subset W$, and the orthonormal basis

of eigenvectors of T^*T is given by $\{g_k\} \subset V$, then the singular value decomposition of T is the expansion given by

$$Tg = \sum_{k \ge 1} \mu_k(g, g_k)_V \psi_k$$

for all $g \in V$. If $\mu_k > 0$, then

$$Tg_k = \mu_k \psi_k$$
 and $T^* \psi_k = \mu_k g_k$.

Also, the rank *r* truncated SVD $T_r : V \to W$ of *T* is defined for $g \in V$ by

$$T_rg := \sum_{k=1}^r \mu_k(g,g_k)_V \psi_k.$$

For more information, see, e.g., [52, Chapters VI–VIII], [53, Section V.2.3], [54, Chapter 30], [55, Sections VI.5–VI.6].

Hilbert-Schmidt Operators: Let $T : V \to W$ be a linear operator, with separable Hilbert spaces V and W, and let $\{g_k\}$ be any orthonormal basis for V. Define the *Hilbert-Schmidt norm* of T as

$$\|T\|_{\mathrm{HS}(V,W)} = \left(\sum_{k\geq 1} \|Tg_k\|_V^2\right)^{1/2}.$$
(2.1)

If the sum converges we say the operator *T* is *Hilbert-Schmidt*. The Hilbert-Schmidt norm is independent of choice of orthonormal basis, every Hilbert-Schmidt operator is compact, $||T|| \leq ||T||_{\text{HS}(V,W)}$, *T* is Hilbert-Schmidt if and only if *T*^{*} is Hilbert-Schmidt, and *T* is Hilbert-Schmidt if and only if $\sum_{k\geq 1} \sigma_k^2 < \infty$, where $\{\sigma_k\}$ are the singular values (including repetitions) of *T*. We also have

$$\|T\|_{\mathrm{HS}(V,W)}^2 = \|T^*\|_{\mathrm{HS}(W,V)}^2 = \sum_{k \ge 1} \sigma_k^2.$$

For more, see, e.g., [52, Chapter VIII], [53, Section V.2.4], [55, Section VI.6].

Bochner Spaces: Let O be an open subset of \mathbb{R}^d , for some $d \ge 1$. For $p \in [1, \infty)$, let $L^p(O; V)$ denote the Bochner space of (equivalence classes of) Lebesgue measurable functions $v : O \to V$ satisfying $\int_O ||v(t)||_V^p dt < \infty$. For p = 2, $L^2(O; V)$ is a Hilbert space with inner product

$$(v, w)_{L^2(O;V)} = \int_O (v(t), w(t))_V dt$$

The following theorem, see, e.g., [56, Theorem III.6.20] and [57, Theorem 4.2.10], allows us to bring a closed linear operator inside an integral.

Theorem 2.1.2. Suppose $T : \mathcal{D}(T) \subset V \to W$ is a closed linear operator. If $v : O \to \mathcal{D}(T)$, $v \in L^1(O; V)$, and $Tv \in L^1(O; W)$, then

$$\int_{O} v(t) dt \in \mathcal{D}(T) \quad and \quad T \int_{O} v(t) dt = \int_{O} Tv(t) dt.$$

2.2. PROPER ORTHOGONAL DECOMPOSITION

Now we introduce proper orthogonal decomposition for both the discrete and continuous time cases.

2.2.1. Discrete Case. Let X be a separable Hilbert space. For the discrete case, let s be a positive integer and assume the POD data is given by $\{w^j\}_{j=1}^s \subset X$. Let $\mathbb{K} = \mathbb{R}$ or $\mathbb{K} = \mathbb{C}$, and define $S := \mathbb{K}_{\Gamma}^s$ with the weighted inner product given by

$$(u,v)_S = v^* \Gamma u = \sum_{j=1}^s \gamma_j u^j \overline{v^j},$$

where $u, v \in S$, $\Gamma = \text{diag}(\gamma_1, \gamma_2, ..., \gamma_s)$, and the values $\{\gamma_j\}_{j=1}^s$ are positive weights. Note these weights commonly arise from integral approximations. Define the POD operator $K: S \to X$ by

$$Kf = \sum_{j=1}^{s} \gamma_j f^j w^j, \quad f = [f^1, f^2, \dots, f^s]^T.$$
(2.2)

Since *K* has finite dimensional range, it is a compact operator and has a singular value decomposition. Let $\{\sigma_k, f_k, \varphi_k\} \subset \mathbb{R} \times S \times X$ be the singular values and orthonormal singular vectors ordered so that $\sigma_1 \ge \sigma_2 \ge \cdots \ge 0$. Thus, the singular value decomposition is given by

$$Kf = \sum_{j \ge 1} \sigma_j(f, f_j)_S \varphi_j.$$
(2.3)

When $\sigma_k > 0$, we have

$$Kf_k = \sigma_k \varphi_k$$
, and $K^* \varphi_k = \sigma_k f_k$,

where $K^* : X \to S$ is the Hilbert adjoint operator given by

$$K^*x = [(x, w^1)_X, (x, w^2)_X, \dots, (x, w^s)_X]^T.$$

For a positive integer r, define $X_r = \operatorname{span}\{\varphi_k\}_{k=1}^r$. Let $\Pi_r^X : X \to X$ be the orthogonal projection onto X_r , i.e., for $x \in X$ fixed, $\Pi_r^X x \in X_r$ minimizes the approximation error $||x - x_r||_X$ over all choices of $x_r \in X_r$. Since $\{\varphi_k\}$ is an orthonormal set in X, we have the exact representation

$$\Pi_r^X x = \sum_{k=1}^r (x, \varphi_k)_X \varphi_k.$$
(2.4)

The singular vectors $\{\varphi_k\}$ are called the POD modes of the data $\{w^k\} \subset X$. The POD modes provide the best low rank approximation to the data in the following sense: we have

$$\sum_{k=1}^{s} \gamma_k \| w^k - \Pi_r^X w^k \|_X^2 = \sum_{k>r} \sigma_k^2,$$
(2.5)

and no other choice of an orthonormal basis in (2.4) gives a smaller value for the approximation error.

Definition 2.2.1. We call the singular values $\{\sigma_k\}$ and singular vectors $\{\varphi_k\} \subset X$ of K the POD singular values and POD modes for the data $\{w^j\}_{j=1}^s$, respectively. We also call the eigenvalues $\{\lambda_k\}$ of the operator $KK^* : X \to X$ the POD eigenvalues for the data $\{w^j\}_{j=1}^s$. We let s_X denote the number of positive POD singular values (or positive POD eigenvalues) for the data $\{w^j\}_{j=1}^s$, i.e., $s_X = \operatorname{rank}(K)$.

From Section 2.1, we know $\lambda_k = \sigma_k^2$ whenever $\lambda_k > 0$. Also, we have $s_X \le s < \infty$. It is possible for data to have a zero POD singular value, but have all positive POD eigenvalues; this can happen if $s > \dim(X)$.

2.2.2. Continuous Case. Similarly to the discrete case we define the POD operator $K : S \to X$ for the continuous case, where again X is a separable Hilbert space. Let d and m be positive integers and let $O \subset \mathbb{R}^d$ be an open set. Then define $S := L^2(O; \mathbb{K}^m)$, where $\mathbb{K} = \mathbb{R}$ or $\mathbb{K} = \mathbb{C}$. We note that $L^2(O)$ is separable (see, e.g., [58, Theorem 2.5-4]), and therefore so is S. Assume the POD data is given by $\{w^j\}_{j=1}^m \subset L^2(O; X)$.

Remark 2.2.2. In POD applications the set O is frequently a time interval; however, researchers also take O to be a multidimensional parameter domain as well. Note that we could also consider multiple open sets, $O_j \subset \mathbb{R}^{d_j}$, and data $w^j \in L^2(O_j; X)$ for j = 1, ..., m. In this case, we would define $S := L^2(O_1) \times \cdots \times L^2(O_m)$. All results in this thesis hold for this case as well. The previous case is chosen to simplify notation.

Define the POD operator $K : S \rightarrow X$ by

$$Kf = \sum_{j=1}^{m} \int_{O} f^{j}(t) w^{j}(t) dt, \quad f \in S.$$
(2.6)

Since $f \in S$, note that $f = [f^1, f^2, ..., f^m]^T$, where each $f^j \in L^2(O)$. As in the discrete case, we know that K is a compact operator and has a singular value decomposition. We let $\{\sigma_k, f_k, \varphi_k\} \subset \mathbb{R} \times S \times X$ denote the singular values and orthonormal singular vectors ordered so that $\sigma_1 \ge \sigma_2 \ge \cdots \ge 0$. The SVD of K is given as in the discrete case (2.3).

Thus, when $\sigma_k > 0$, we have

$$Kf_k = \sigma_k \varphi_k$$
, and $K^* \varphi_k = \sigma_k f_k$,

where $K^* : X \to S$ is the Hilbert adjoint operator defined by

$$[K^*x](t) = [(x, w^1(t))_X, (x, w^2(t))_X, \dots, (x, w^m(t))_X]^T.$$

We define $X_r := \operatorname{span}\{\varphi_k\}_{k=1}^r$ and the orthogonal projection $\Pi_r^X : X \to X$ (2.4) as before. The data approximation error is given by

$$\sum_{j=1}^{m} \int_{O} \|w^{j}(t) - \Pi_{r}^{X} w^{j}(t)\|_{X}^{2} dt = \sum_{k>r} \sigma_{k}^{2},$$
(2.7)

and the error goes to zero as $r \to \infty$. As in the discrete case, no other orthonormal basis in (2.4) gives a smaller value for the error.

We define the POD singular values, POD modes, POD eigenvalues, and $s_X = \operatorname{rank}(K)$ as in Definition 2.2.1 for the discrete case. Again, it is possible for data to have a zero POD singular value, but have all positive POD eigenvalues; an example where X is infinite dimensional can be found in [36, Section 3.1, Example 3]. Also, if X is finite dimensional, then the data always has a zero POD singular value.

Note that while the discrete time case of POD is considered throughout the thesis, the continuous time case is only considered here and in Chapter 3.

2.2.3. Optimality of POD. Below, we present the discrete and continuous versions of a result regarding Hilbert-Schmidt norms and POD. The continuous case result is known (see, e.e., [59, Section 3.5], [60, Theorem 12.6.1], [61, Lemma 4.4]), although perhaps not exactly in this precise form. We provide a proofs of both results to be complete, and also since the results are crucial to this work. These lemmas are used to prove the optimality of POD below.

Lemma 2.2.3. For given data $\{y^j\}_{j=1}^s \subset X$ in the discrete case, the Hilbert-Schmidt norm of the POD operator $K : S \to X$ is given by

$$\|K\|_{\mathrm{HS}(S,X)}^2 = \sum_{j=1}^s \gamma_j \|y^j\|_X^2.$$

Proof. Let $\{\xi_k\}_{k\geq 1}$ be an orthonormal basis for *X*. We have

$$\begin{split} \|K\|_{\mathrm{HS}(S,X)}^2 &= \|K^*\|_{\mathrm{HS}(X,S)}^2 = \sum_{k\geq 1} \|K^*\xi_k\|_S^2 = \sum_{k\geq 1} \sum_{j=1}^s \gamma_j |(\xi_k, y^j)_X|^2 \\ &= \sum_{j=1}^s \gamma_j \sum_{k\geq 1} |(\xi_k, y^j)_X|^2 = \sum_{j=1}^s \gamma_j \|y^j\|_X^2 \end{split}$$

by Parseval's inequality.

Lemma 2.2.4. Let Z be a separable Hilbert space, and let $S = L^2(O; \mathbb{K}^m)$, where O is an open subset of \mathbb{R}^d . If $K : S \to Z$ is defined by

$$Kf = \sum_{j=1}^m \int_O f^j(t) z^j(t) dt,$$

for $\{z^j\}_{j=1}^m \subset L^2(O; Z)$, then K is Hilbert-Schmidt and

$$\|K\|_{\mathrm{HS}(S,Z)}^{2} = \sum_{j=1}^{m} \|z^{j}\|_{L^{2}(O;Z)}^{2}$$

Proof. Let $\{\chi_i\}_{i\geq 1} \subset L^2(O)$ and $\{\xi_n\}_{n\geq 1} \subset Z$ be orthonormal bases. Therefore, $\{\overline{\chi}_i\}_{i\geq 1}$ is also an orthonormal basis for $L^2(O)$, and $\{\chi_i\xi_n\}_{i,n\geq 1}$ is an orthonormal basis for $L^2(O; Z)$ (see, e.g., [60, Theorem 12.6.1]).

For $\xi \in Z$, let $[K^*\xi]^j = (\xi, z^j(t))_Z$ denote the *j*th component of $K^*\xi \in S$. Working with the Hilbert adjoint operator K^* and using Parseval's equality gives

$$\|K^*\|_{\mathrm{HS}(S,Z)}^2 = \sum_{n \ge 1} \|K^* \xi_n\|_S^2$$

$$= \sum_{j=1}^{m} \sum_{n \ge 1} \left\| \left[K^{*} \xi_{n} \right]^{j} \right\|_{L^{2}(O)}^{2}$$

$$= \sum_{j=1}^{m} \sum_{n,i \ge 1} \left| \left(\overline{\chi}_{i}, \left[K^{*} \xi_{n} \right]^{j} \right)_{L^{2}(O)} \right|^{2}$$

$$= \sum_{j=1}^{m} \sum_{n,i \ge 1} \left| \int_{O} \overline{\chi_{i}(t)} (z^{j}(t), \xi_{n})_{Z} dt \right|^{2}$$

$$= \sum_{j=1}^{m} \sum_{n,i \ge 1} \left| \int_{O} (z^{j}(t), \chi_{i}(t)\xi_{n})_{Z} dt \right|^{2}$$

$$= \sum_{j=1}^{m} \sum_{n,i \ge 1} \left| (z^{j}, \chi_{i}\xi_{n})_{L^{2}(O;Z)} \right|^{2}$$

$$= \sum_{j=1}^{m} \left\| z^{j} \right\|_{L^{2}(O;Z)}^{2}.$$

To be complete, we present a brief proof of the optimality of POD for low rank data approximation in both the discrete and continuous cases. Our problem statement and proof strongly rely on the ideas from [50] and [49].

POD optimality problem: Let *X* be a separable Hilbert space, and let $S = \mathbb{K}_{\Gamma}^{s}$ in the discrete case or $S = L^{2}(O; \mathbb{K}^{m})$ in the continuous case, where $\mathbb{K} = \mathbb{R}$ or $\mathbb{K} = \mathbb{C}$. Suppose we have given data $\{w^{j}\}_{j=1}^{s} \subset X$ in the discrete case or $\{w^{j}\}_{j=1}^{m} \subset L^{2}(O; X)$ in the continuous case. The POD optimality problem is to find coefficients $\{a_{k}\} \subset \mathbb{K}$ and basis elements $\{s_{k}\} \subset S$ and $\{\eta_{k}\} \subset X$ so that the *r*th order approximations

$$w_r^j = \sum_{k=1}^r a_k s_k^j \eta_k \qquad \text{for } j = 1, \dots, s \text{ (discrete case)},$$
$$w_r^j(t) = \sum_{k=1}^r a_k s_k^j(t) \eta_k \qquad \text{for } j = 1, \dots, m \text{ (continuous case)},$$

minimize the data approximation error

$$E_r(w_r) = \sum_{j=1}^s \gamma_j ||w^j - w_r^j||_X^2 \qquad (\text{discrete case}),$$

$$E_r(w_r) = \sum_{j=1}^m \int_O \|w^j(t) - w_r^j(t)\|_X^2 dt$$
 (continuous case).

Remark 2.2.5. In many papers on POD, the basis elements $\{\eta_k\} \subset X$ are required to be orthonormal, and w_r^j is also required to equal the orthogonal projection of w^j onto span $\{\eta_k\}_{k=1}^r$. Therefore, the POD problem above allows more general approximations. The final result is the same.

Notation: For given data $\{y^j\}_{j=1}^s \subset X$ in the discrete case or $\{y^j\}_{j=1}^m \subset L^2(O; X)$ in the continuous case, we let $K(y) : S \to X$ denote the POD operator for the data and we let $K^*(y) : X \to S$ denote the Hilbert adjoint operator of K(y).

The proof of the next result follows directly from definitions and is omitted. Lemma 2.2.6. *If the data is given by*

$$y^{j} = \sum_{k=1}^{p} \alpha_{k} s_{k}^{j} \eta_{k}$$
 (discrete case), $y^{j}(t) = \sum_{k=1}^{p} \alpha_{k} s_{k}^{j}(t) \eta_{k}$, (continuous case)

for each j with $\{\alpha_k\} \subset \mathbb{K}, \{s_k\} \subset S$, and $\{\eta_k\} \subset X$, then the POD operator $K(y) : S \to X$ is given by

$$K(y)f = \sum_{k=1}^{p} \alpha_k (f, \overline{s_k})_S \eta_k, \quad f \in S.$$

Now we prove the main optimality result. We rely on the fact that the rank r truncated SVD of K(y) is the optimal rank r approximation to K(y) in the Hilbert-Schmidt norm; see, e.g., [62, Section III.7, Theorem 7.1].

Theorem 2.2.7. Let $\{w^j\}_{j=1}^s \subset X$ in the discrete case or $\{w^j\}_{j=1}^m \subset L^2(O; X)$ in the continuous case be given data, and let $\{\sigma_i, f_i, \varphi_i\} \subset \mathbb{R} \times S \times X$ be the ordered singular values of $K(w) : S \to X$ and the corresponding orthonormal bases of singular vectors. A solution of the POD problem is given by $\{w_r^j\}_{j=1}^s \subset X$ in the discrete case or $\{w_r^j\}_{j=1}^m \subset L^2(O; X)$ in the continuous case, where

$$w_r^j = \sum_{k=1}^r \sigma_k \overline{f_k^j} \varphi_k = \sum_{k=1}^r (w^j, \varphi_k)_X \varphi_k \qquad (discrete \ case),$$

$$w_r^j(t) = \sum_{k=1}^r \sigma_k \overline{f_k^j(t)} \varphi_k = \sum_{k=1}^r \left(w^j(t), \varphi_k \right)_X \varphi_k \qquad (continuous \ case).$$

The minimum approximation error is given by

$$E_r^{\min} := E_r(w^r) = \sum_{k>r} \sigma_k^2 < \infty,$$

and $E_r^{\min} \to 0$ as r increases.

Proof. We first assume $r \leq s_X$ so that $\sigma_k > 0$ for k = 1, ..., r.

First, the equivalence of the two expressions for w_r^j comes from $K^*(w)\varphi_k = \sigma_k f_k$, $\sigma_k > 0$ for k = 1, ..., r, and the formulas for $K^*(w)$. Also, for $g \in S$, Lemma 2.2.6 implies

$$K(w_r)g = \sum_{k=1}^r \sigma_k(g, f_k)_S \varphi_k = K_r(w)g.$$

Therefore, $K(w^r) = K_r(w)$, where $K_r(w) : S \to X$ is the *r*th order truncated SVD of the POD operator $K(w) : S \to X$.

Next, by the Hilbert-Schmidt norm results Lemma 2.2.4 and Lemma 2.2.3 and since the POD operator is linear in the data we have

$$E_r(w^r) = \|K(w - w^r)\|_{\mathrm{HS}}^2 = \|K(w) - K(w^r)\|_{\mathrm{HS}}^2 = \|K(w) - K_r(w)\|_{\mathrm{HS}}^2 = \sum_{k>r} \sigma_k^2.$$

Also, since $||K(w)||_{\text{HS}} = \sum_{k \ge 1} \sigma_k^2 < \infty$, we have $\sum_{k > r} \sigma_k^2 \to 0$ as *r* increases.

Now we show that this is the smallest value possible for the error. Let coefficients $\{a_k\} \subset \mathbb{K}$ and basis elements $\{s_k\} \subset S$ and $\{\eta_k\} \subset X$ be given, and define the *r*th order approximation

$$z_r^j = \sum_{k=1}^r \alpha_k s_k^j \eta_k$$
 (discrete case), $z_r^j(t) = \sum_{k=1}^p \alpha_k s_k^j(t) \eta_k$, (continuous case).

By Lemma 2.2.6, $K(z^r)$ has rank at most *r*. Therefore, we have

$$E_r(z^r) = \|K(w - z^r)\|_{\mathrm{HS}}^2 = \|K(w) - K(z^r)\|_{\mathrm{HS}}^2 \ge \sum_{k>r} \sigma_k^2.$$

Next, if $s_X < \infty$, then the result is true for $r = s_X$. Therefore, we have $w^j = w_{s_X}^j$ for all *j*, and this proves the result for $r > s_X$.

3. NEW POD APPROXIMATION THEORY

3.1. ASSUMPTIONS AND NOTATION

In this section, assume X and Y are separable Hilbert spaces, and $L : \mathcal{D}(L) \subset X \rightarrow Y$ is a linear operator. We study POD error formulas and POD projections involving the data $\{w_i\}$ and the data $\{Lw_i\}$.

3.1.1. Discrete Case. Recall from Section 2.2.1 we consider data $\{w^j\}_{j=1}^s \subset X$ and the corresponding POD operator $K : S \to X$ defined by $Kf = \sum_{j=1}^s \gamma_j f^j w^j$, where $S = \mathbb{K}_{\Gamma}^s$ and \mathbb{K} is either \mathbb{R} or \mathbb{C} . The singular value decomposition of K is given by $Kf = \sum_{k\geq 1} \sigma_k(f, f_k)_S \varphi_k$. The set X_r is the span of $\{\varphi_k\}_{k=1}^r$, and $\Pi_r^X : X \to X$ is the orthogonal projection onto X_r .

To consider POD projections involving the data $\{Lw^j\}$, we make the following assumption:

Main assumption: For the discrete case, we assume throughout the paper that (i) $\{w^j\}_{j=1}^s \subset \mathcal{D}(L)$, and also (ii) $\sigma_r > 0$ whenever we consider the projection Π_r^X .

Assumption (i) has two important consequences. First, since $w^j \in \mathcal{D}(L)$ for each j, we know the range of K is contained in $\mathcal{D}(L)$. Second, assumption (i) allows us to consider the POD operator $K^Y : S \to Y$ for the data $\{Lw^j\}_{j=1}^s \subset Y$ defined by

$$K^{Y}f = LKf = \sum_{j=1}^{s} \gamma_{j}f^{j}Lw^{j}, \quad f = [f^{1}, f^{2}, \dots, f^{s}]^{T}.$$
 (3.1)

Note that K^Y is the result of applying *L* to the POD operator *K* for the data $\{w^j\}$, i.e., $K^Y = LK$. Since K^Y has finite rank, it is compact and has a singular value decomposition. Define $s_Y = \operatorname{rank}(K^Y)$ to be the number of positive singular values of K^Y . Note that assumption (i) is automatically satisfied if *L* is bounded. For assumption (ii), note that if $\sigma_k > 0$, then assumption (i) implies the corresponding singular vector φ_k is in $\mathcal{D}(L)$ since

$$\varphi_k = \sigma_k^{-1} K f_k \in \mathcal{D}(L). \tag{3.2}$$

Since $\sigma_r > 0$, this implies $X_r \subset \mathcal{D}(L)$ and Π_r^X maps into $\mathcal{D}(L)$.

To guarantee the boundedness of certain POD projections, in some cases of Theorem 3.2.3 we need to assume the POD modes $\{\varphi_k\}_{k=1}^r \subset \mathcal{D}(L)$ satisfy some additional regularity properties. These properties can be guaranteed by making additional regularity assumptions on the data.

First, the condition $\{\varphi_k\}_{k=1}^r \subset \mathcal{D}(L^{-*})$ is guaranteed to hold if we assume $\sigma_r > 0$ and $w^j \in \mathcal{D}(L^{-*})$ for each j. With this assumption, we know as above that $\mathcal{R}(K) \subset \mathcal{D}(L^{-*})$ and also $\varphi_k \in \mathcal{D}(L^{-*})$ whenever $\sigma_k > 0$. Since $\sigma_r > 0$, we can guarantee $\{\varphi_k\}_{k=1}^r \subset \mathcal{D}(L^{-*})$.

Next, a similar argument using (3.2) shows the condition $\{L\varphi_k\}_{k=1}^r \subset \mathcal{D}(L^*)$ is guaranteed to hold if we assume $\sigma_r > 0$ and $Lw^j \in \mathcal{D}(L^*)$ for each j.

3.1.2. Continuous Case. The continuous case requires a few more assumptions than the discrete case. Recall $K : S \to X$, where $S := L^2(O; \mathbb{K}^m)$ and \mathbb{K} is either \mathbb{R} or \mathbb{C} . In order to define the POD operator K^Y and ensure $\{\varphi_k\}_{k=1}^r \subset \mathcal{D}(L)$, we make the following assumption:

Main assumption: For the continuous case, we assume throughout the paper that (i) $\{Lw^j\}_{j=1}^m \subset L^2(O; Y)$, and for all $f \in S$ we have $Kf \in \mathcal{D}(L)$ and

$$LKf = \sum_{j=1}^{m} \int_{O} f^{j}(t) Lw^{j}(t) dt,$$

and also (ii) $\sigma_r > 0$ whenever we consider the projection Π_r^X .

As in the discrete case, assumption (i) gives $\mathcal{R}(K) \subset \mathcal{D}(L)$ and allows us to define the (compact) POD operator $K^Y = LK$ for the data $\{Lw^j\}_{j=1}^m \subset L^2(O; Y)$. As before, we let $s_Y = \operatorname{rank}(K^Y)$ be the number of positive singular values of K^Y . Also as in the discrete case, assumptions (i) and (ii) imply $\{\varphi_k\}_{k=1}^r \subset \mathcal{D}(L)$ and Π_r^X maps into $\mathcal{D}(L)$.

Remark 3.1.1. *There are three common conditions that guarantee assumption (i) holds.*

- 1. If $L : X \to Y$ is bounded, the operator L can be pulled through the integral in the definition of K and assumption (i) clearly holds.
- 2. If each $w^j \in L^2(O; X)$ takes the form

$$w^j(t) = \sum_{\ell,k=1}^{n_j} a^j_{k\ell} g^j_k(t) x_{\ell j},$$

where $a_{k\ell}^{j}$ are constants in \mathbb{K} , $g_{k}^{j}(t) \in L^{2}(O)$, and $x_{\ell j} \in \mathcal{D}(L)$, then it can be checked that assumption (i) holds. This condition is similar to the assumption made in the discrete case.

3. If $L : \mathcal{D}(L) \subset X \to Y$ is closed, $w^j \in \mathcal{D}(L)$ a.e., and $Lw^j \in L^2(O;Y)$ then Theorem 2.1.2 implies assumption (i) holds.

Again, for certain cases of Theorem 3.2.3 we need to assume the POD modes $\{\varphi_k\}_{k=1}^r \subset \mathcal{D}(L)$ satisfy some additional regularity properties. As in the discrete case, we can make additional assumptions on the data to satisfy these regularity properties.

We briefly mention conditions on the data similar to Remark 3.1.1, Item 3 that yield the needed regularity. First, if L^{-*} exists, it is closed. Therefore, $\{\varphi_k\}_{k=1}^r \subset \mathcal{D}(L^{-*})$ holds if we assume $\sigma_r > 0$, $w^j \in \mathcal{D}(L^{-*})$ a.e., and $\{L^{-*}w^j\}_{j=1}^m \in L^2(O; Y)$. Second, if L^* exists, then it is closed. Therefore, $\{L\varphi_k\}_{k=1}^r \subset \mathcal{D}(L^*)$ holds if we assume $\sigma_r > 0$, $Lw^j \in \mathcal{D}(L^*)$ a.e., and $\{L^*Lw^j\}_{j=1}^m \in L^2(O; X)$.

We also note that the condition in Remark 3.1.1, Item 2 can be modified similarly to the discrete case to yield the required regularity.

3.2. POD PROPERTIES

Recall the standard POD orthogonal projection, $\Pi_r^X : X \to X$ given by (2.4), and the known POD data approximation error given by

$$\sum_{j=1}^{m} \int_{O} \|w^{j}(t) - \Pi_{r}^{X} w^{j}(t)\|_{X}^{2} dt = \sum_{k>r} \sigma_{k}^{2}.$$

One of the goals of this dissertation is to find extensions of this error formula to other scenarios involving the linear operator $L : X \to Y$ and another sequence of projections, which need not be orthogonal.

Definition 3.2.1. For a positive integer r with $\sigma_r > 0$, we define $Y_r := LX_r = \text{span}\{L\varphi_k\}_{k=1}^r$ and we let $\Pi_r^Y : Y \to Y$ be a projection onto Y_r .

Remark 3.2.2. The condition $\sigma_r > 0$ implies $X_r \subset \mathcal{D}(L)$, so the definition makes sense. We assume throughout that $\sigma_r > 0$ whenever we consider Π_r^Y . It is important to note that unless stated otherwise we do not assume the projection Π_r^Y is orthogonal. To obtain convergence results as r increases, we sometimes need to require $\{\Pi_r^Y\}$ are uniformly bounded in operator norm. If $\{\Pi_r^Y\}$ are the orthogonal projections onto Y_r , then this condition is satisfied.

3.2.1. Non-orthogonal POD Projections. In Section 3.4, we consider pointwise convergence results for the linear operators $L^{-1}\Pi_r^Y L : X \to X$ and $L\Pi_r^X L^{-1} : Y \to Y$. Below, we give conditions that guarantee that these linear operators are bounded, or have bounded extensions, for r fixed. We note that when these operators are bounded we have $L^{-1}\Pi_r^Y L : X \to X$ is a projection onto $X_r = \operatorname{span}\{\varphi\}_{j=1}^r$ and $L\Pi_r^X L^{-1} : Y \to Y$ is a projection onto $Y_r = \operatorname{span}\{L\varphi\}_{j=1}^r$. Even if Π_r^Y is an orthogonal projection, these projections are typically non-orthogonal POD projection operators.

In the simplest case, if L and L^{-1} are bounded, then clearly $L^{-1}\Pi_r^Y L : X \to X$ and $L\Pi_r^X L^{-1} : Y \to Y$ are both bounded for each r. In this case, $\{L\Pi_r^X L^{-1}\}$ is uniformly bounded in operator norm, and $\{L^{-1}\Pi_r^Y L\}$ is also uniformly bounded when $\{\Pi_r^Y\}$ is uniformly bounded.

Below, we consider the case when either L or L^{-1} is unbounded. For each fixed r, we show $L^{-1}\Pi_r^Y L : X \to X$ is bounded when L is bounded, and $L\Pi_r^X L^{-1} : Y \to Y$ is bounded when L^{-1} is bounded. In other cases, we need certain assumptions to be satisfied to construct bounded extensions of the operators for each r. We do *not* show that these non-orthogonal POD projection operators are uniformly bounded in operator norm.

In specific cases, we need certain adjoint operators to exist, so we need the operators to be densely defined or bounded. For example, for the operator L^{-*} to exist we must assume that $\mathcal{D}(L^{-1})$ is dense in Y, or L^{-1} is bounded. These type of assumptions must be added to the second and fourth parts of the following theorem, in addition to results later in this section.

Theorem 3.2.3. Assume L is invertible and r > 0 is fixed.

- 1. If L^{-1} is bounded, then $L\Pi_r^X L^{-1} : Y \to Y$ is bounded.
- 2. If $\mathcal{D}(L^{-1})$ is dense in Y and $\{\varphi_k\}_{k=1}^r \subset \mathcal{D}(L^{-*})$, the operator $L\prod_r^X L^{-1} : Y \to Y$ can be extended to a bounded operator on Y.
- 3. If L is bounded, then $L^{-1}\Pi_r^Y L : X \to X$ is bounded.
- 4. Assume $\Pi_r^Y : Y \to Y$ is the orthogonal projection onto $span\{L\varphi_k\}_{k=1}^r$. If L^{-1} is bounded, $\mathcal{D}(L)$ is dense, and $\{L\varphi_k\}_{k=1}^r \subset \mathcal{D}(L^*)$, then $L^{-1}\Pi_r^Y L : X \to X$ can be extended to a bounded operator on X.

Remark 3.2.4. In the second and fourth items, we assume the POD modes satisfy the regularity properties $\{\varphi_k\}_{k=1}^r \subset \mathcal{D}(L^{-*})$ and $\{L\varphi_k\}_{k=1}^r \subset \mathcal{D}(L^*)$, respectively. See Section 3.1.1 and Section 3.1.2 for conditions on the data in the discrete and continuous cases that guarantee these properties hold.

Proof. 1. Note

$$L\Pi_{r}^{X}L^{-1}y = \sum_{k=1}^{r} (L^{-1}y,\varphi_{k})_{X}L\varphi_{k}.$$
(3.3)

Since L^{-1} is a bounded operator and $\varphi_k \in \mathcal{D}(L)$ for all k, the sum in (3.3) is well defined for all $y \in Y$. Also, it can be checked that

$$\|L\Pi_r^X L^{-1} y\|_Y \le c \|y\|_Y, \tag{3.4}$$

where the constant $c := \|L^{-1}\| \left(\sum_{k=1}^{r} \|L\varphi_k\|_Y^2\right)^{1/2}$ depends on *r*. This shows that the operator $L\Pi_r^X L^{-1}$ is bounded when L^{-1} is bounded.

2. The linear operator $L\Pi_r^X L^{-1} : Y \to Y$ is defined by (3.3) for all $y \in \mathcal{D}(L^{-1})$. Using the assumptions, we can rewrite (3.3) for $y \in \mathcal{D}(L^{-1})$ as

$$L\Pi_{r}^{X}L^{-1}y = \sum_{k=1}^{r} (y, L^{-*}\varphi_{k})_{X}L\varphi_{k}.$$
(3.5)

It can be checked that (3.4) holds for all $y \in \mathcal{D}(L^{-1})$ with

$$c := \sum_{k=1}^r \|L^{-*}\varphi_k\|_X \|L\varphi_k\|_Y.$$

Note that (3.5) is well-defined for all $y \in Y$, and therefore yields a bounded linear extension of $L\Pi_r^X L^{-1} : Y \to Y$ to all of *Y*.

3. Since Π_r^Y is a projection onto $Y_r = \operatorname{span}\{L\varphi\}_{j=1}^r$, we know for $y \in Y$ there exists constants $\{\alpha_j(y)\}$ depending on y such that $\Pi_r^Y y = \sum_{j=1}^r \alpha_j(y) L\varphi_j$. Then

$$\left\|\sum_{j=1}^{r} \alpha_{j}(y) L \varphi_{j}\right\|_{Y} = \|\Pi_{r}^{Y} y\|_{Y} \le \|\Pi_{r}^{Y}\| \|y\|_{Y}.$$
(3.6)

Also,

$$\left\|\sum_{j=1}^{r} \alpha_{j}(y) L \varphi_{j}\right\|_{Y}^{2} = \sum_{j,k=1}^{r} \alpha_{j}(y) (L \varphi_{j}, L \varphi_{k})_{Y} \overline{\alpha_{k}(y)} = \alpha(y)^{*} A_{r} \alpha(y),$$

where the star denotes complex conjugate, and

$$\alpha(\mathbf{y}) = [\alpha_1(\mathbf{y}), ..., \alpha_r(\mathbf{y})]^T \in \mathbb{K}^r, \quad [A_r]_{i,j} = (L\varphi_i, L\varphi_j)_Y.$$

Since *L* is invertible and $\{\varphi_j\}_{j=1}^r$ is a linearly independent set, we know $\{L\varphi_j\}_{j=1}^r$ is a linearly independent set; therefore, A_r is symmetric positive definite, which implies there exists $\beta > 0$ such that $\alpha^* A_r \alpha \ge \beta \|\alpha\|_{\mathbb{K}^r}^2$ for all $\alpha \in \mathbb{K}^r$. Note that β may depend on *r*. Together, the above implies that

$$\beta \|\alpha(y)\|_{\mathbb{K}^r}^2 \le \left\| \sum_{j=1}^r \alpha_j(y) L\varphi_j \right\|_Y^2 \le \|\Pi_r^Y\|^2 \|y\|_Y^2.$$

So,

$$\|\alpha(y)\|_{\mathbb{K}^r} \le \beta^{-1/2} \|\Pi_r^Y\| \|y\|_Y.$$
(3.7)

In this case, y = Lx and L is bounded and invertible; thus,

$$L^{-1}\Pi_r^Y(Lx) = L^{-1}\sum_{j=1}^r \alpha_j(Lx)L\varphi_j = \sum_{j=1}^r \alpha_j(Lx)\varphi_j$$

where the constants α_j now depend on Lx. Since $\{\varphi_j\} \subset X$ is orthonormal, we have

$$\begin{split} \|L^{-1}\Pi_{r}^{Y}Lx\|_{X}^{2} &= \left\|\sum_{j=1}^{r}\alpha_{j}(Lx)\varphi_{j}\right\|_{X}^{2} \\ &= \|\alpha(Lx)\|_{\mathbb{K}^{r}}^{2} \\ &\leq \beta^{-1}\|\Pi_{r}^{Y}\|^{2}\|Lx\|_{Y}^{2} \\ &\leq \beta^{-1}\|\Pi_{r}^{Y}\|^{2}\|L\|^{2}\|x\|_{X}^{2}. \end{split}$$

Therefore, for all $x \in X$ we have

$$\|L^{-1}\Pi_r^Y Lx\|_X \le c \|x\|_X, \tag{3.8}$$

where $c := \beta^{-1/2} \|\Pi_r^Y\| \|L\|$.

4. We obtain a representation of $L^{-1}\Pi_r^Y L$ as follows. First, note that the sets $\{L\varphi_k\}$ and $\{L^{-*}\varphi_k\}$ are biorthogonal, i.e., $(L\varphi_k, L^{-*}\varphi_j)_Y = \delta_{k,j}$, where $\delta_{k,j}$ is the Kronecker delta symbol. Recall from the proof of part 3 that $\Pi_r^Y y = \sum_{k=1}^r \alpha_k L\varphi_k$ for some scalars α_k that depend on y. We can calculate the values for α_k by noting

$$\left(\Pi_r^Y y, L^{-*} \varphi_j\right)_Y = \sum_{k=1}^r \alpha_k (L \varphi_k, L^{-*} \varphi_j)_Y = \alpha_j.$$

This yields

$$\Pi_{r}^{Y} y = \sum_{k=1}^{r} (\Pi_{r}^{Y} y, L^{-*} \varphi_{k})_{Y} L \varphi_{k} = \sum_{k=1}^{r} (y, \Pi_{r}^{Y} L^{-*} \varphi_{k})_{Y} L \varphi_{k},$$
(3.9)

since Π_r^Y is orthogonal and therefore $(\Pi_r^Y)^* = \Pi_r^Y$.

By assumption, $\{L\varphi_j\} \subset \mathcal{D}(L^*)$ and so (3.9) implies $\Pi_r^Y y \in \mathcal{D}(L^*)$ for all $y \in Y$. This gives the following representation for any $x \in \mathcal{D}(L)$:

$$L^{-1}\Pi_{r}^{Y}Lx = \sum_{k=1}^{r} (x, L^{*}\Pi_{r}^{Y}L^{-*}\varphi_{k})_{X}\varphi_{k}.$$
(3.10)

Also, for all $x \in \mathcal{D}(L)$, the bound (3.8) holds with $c := \left(\sum_{k=1}^{r} \|L^* \Pi_r^Y L^{-*} \varphi_k\|_X^2\right)^{1/2}$. Equation (3.10) is well-defined for all $x \in X$, and therefore defines a bounded linear extension of $L^{-1} \Pi_r^Y L : X \to X$ to all of X.

3.2.2. POD Singular Values and POD Eigenvalues. The number of nonzero singular values (or eigenvalues) of the POD operators plays an important role throughout the thesis. It is also important to note the difference between singular values and eigenvalues. For a POD operator $K : S \rightarrow Z$, recall the POD eigenvalues are the eigenvalues of $KK^* : Z \rightarrow Z$, the POD singular values are the singular values of K, and $s_Z = \operatorname{rank}(K)$,

i.e., s_Z is the number of positive POD singular values of K (or positive POD eigenvalues of KK^*). As discussed in Section 2.2, it is possible to have a zero POD singular value but to have all nonzero POD eigenvalues.

Below, we study various relationships between the POD eigenvalues and POD singular values for the data $\{w^j\}$ and the data $\{Lw^j\}$. Recall, $K : S \to X$ is the POD operator for the data $\{w^j\}$, and $K^Y = LK : S \to Y$ is the POD operator for the data $\{Lw^j\}$. Therefore, $s_X = \operatorname{rank}(K)$ is the number of nonzero POD singular values (or POD eigenvalues) for the data $\{w^j\}$, and $s_Y = \operatorname{rank}(K^Y)$ is the number of nonzero POD singular values values (or POD singular values) for the data $\{Lw^j\}$.

First, we give a relationship between the POD eigenvalues and the null space of the adjoint POD operator. Then we also give some additional information about s_X and s_Y . There is also a relationship between the number of POD eigenvalues under the linear mapping *L*.

- **Lemma 3.2.5.** 1. All of the POD eigenvalues for the data $\{w^j\}$ are nonzero if and only if ker $(K^*) = \{0\}$. In this case, $X = \overline{\mathcal{R}(K)}$. In addition, if $s_X < \infty$, then $X = \mathcal{R}(K)$ and dim $(X) = s_X$.
 - 2. All of the POD eigenvalues for the data $\{Lw^j\}$ are nonzero if and only if $ker((K^Y)^*) = \{0\}$. In this case, $Y = \overline{\mathcal{R}(K^Y)}$. In addition, if $s_Y < \infty$, then $Y = \mathcal{R}(K^Y)$ and $\dim(Y) = s_Y$.
 - 3. The number of nonzero POD eigenvalues for $\{Lw^j\}$ is less than or equal to the number of nonzero POD eigenvalues for $\{w^j\}$. That is, $s_Y \leq s_X$.
 - 4. If L is invertible, then $s_X = s_Y$.

Proof. The first two items are proven similarly. Here we show item 1.

1. Lemma 2.1.1 proves the first statement. To see the rest, note that $X = \ker(K^*) \oplus \overline{\mathcal{R}(K)}$ and $\ker(K^*) = \{0\}$ imply $X = \overline{\mathcal{R}(K)}$. Then if $s_X = \operatorname{rank}(K) = \dim(\mathcal{R}(K))$ is finite, we have $\overline{\mathcal{R}(K)} = \mathcal{R}(K)$ and therefore $X = \mathcal{R}(K)$ and $\dim(X) = s_X$. 3. First, if $s_X = \infty$, we are done. Assume $s_X < \infty$. We know

$$Kf = \sum_{j=1}^{s_X} \sigma_j(f, f_j)_S \varphi_j,$$

and therefore

$$K^{Y}f = LKf = \sum_{j=1}^{s_{X}} \sigma_{j}(f, f_{j})_{S}L\varphi_{j}.$$

Thus, $s_Y = \operatorname{rank}(K^Y) \le s_X$.

4. Because of item 3, we need only show $s_X \le s_Y$. First, if $s_Y = \infty$, we are done. Assume $s_Y < \infty$. Let the singular value decomposition of K^Y be given by

$$K^{Y}f = LKf = \sum_{j=1}^{s_{Y}} \sigma_{j}^{Y}(f, f_{j}^{Y})_{S}\varphi_{j}^{Y}.$$

Note that $\varphi_j^Y \in \mathcal{D}(L^{-1})$ whenever $\sigma_j^Y > 0$, since $\mathcal{D}(L^{-1}) = R(L)$ and

$$\varphi_j^Y = (\sigma_j^Y)^{-1} K^Y f_j^Y = (\sigma_j^Y)^{-1} L K f_j^Y.$$

Then, since L is invertible,

$$Kf = L^{-1}LKf = L^{-1}K^{Y}f = \sum_{j=1}^{s_{Y}} \sigma_{j}^{Y}(f, f_{j}^{Y})_{S}L^{-1}\varphi_{j}^{Y},$$

and therefore $s_X = \operatorname{rank}(K) \leq s_Y$.

The following lemma gives further results about the connections between the two main sets of POD eigenvalues under consideration in this paper, i.e., the POD eigenvalues for the data $\{w^j\}$ and the data $\{Lw^j\}$. With extra assumptions, we can use the fact that all the POD eigenvalues are nonzero for one set of data to obtain the same conclusion for the other set of data.
- **Lemma 3.2.6.** 1. If L is bounded, $\mathcal{R}(L)$ is dense in Y, and the POD eigenvalues for $\{w^j\}$ are all nonzero, then the POD eigenvalues for $\{Lw^j\}$ are all nonzero.
 - 2. If L^{-1} is bounded, $\mathcal{R}(L^{-1})$ is dense in X, and the POD eigenvalues for $\{Lw^j\}$ are all nonzero, then the POD eigenvalues for $\{w^j\}$ are all nonzero.

Proof. The proofs of the two items are similar; we only prove the first item.

Since $X = \ker(K^*) \oplus \overline{\mathcal{R}(K)}$ and $\ker(K^*) = \{0\}$ (Lemma 3.2.5, Item 1), we have $X = \overline{\mathcal{R}(K)}$. Let $\varepsilon > 0$ and let $y \in Y$. Since $\mathcal{R}(L)$ is dense in Y, there exists $x \in X$ such that $||y - Lx||_Y < \varepsilon/2$. Since $X = \overline{\mathcal{R}(K)}$, for this x there exists $f \in S$ such that $||x - Kf||_X < \varepsilon/(2||L||)$. This gives

$$\|y - LKf\|_{Y} < \|y - Lx\|_{Y} + \|Lx - LKf\|_{Y} < \frac{\varepsilon}{2} + \|L\|\frac{\varepsilon}{2\|L\|} < \varepsilon,$$

which shows $\overline{\mathcal{R}(K^Y)} = Y$ and ker $((K^Y)^*) = \{0\}$. Thus, the POD eigenvalues for $\{Lw^j\}$ are all nonzero by Lemma 3.2.5, Item 2.

3.3. ERROR FORMULAS

One goal of this thesis is to provide exact formulas for POD data approximation errors. The two main results of this section can be found in Theorem 3.3.2 and Theorem 3.3.4. The section is split between the discrete case, where we can use a more direct proof approach, and the continuous case, which requires more care since the data can have infinitely many nonzero POD eigenvalues.

The next lemma gives three different Hilbert-Schmidt norm approximation results involving the POD operator K for the data $\{w^j\}$ and the POD operator $K^Y = LK$ for the data $\{Lw^j\}$. The result will be of particular usefulness when discussing the continuous case in Section 3.3.2, but it applies to the discrete case as well. We also use this result throughout Section 3.4. Lemma 3.3.1. The Hilbert-Schmidt norm errors are given by

$$\|LK - L\Pi_r^X K\|_{\mathrm{HS}(S,Y)}^2 = \sum_{k>r} \sigma_k^2 \|L\varphi_k\|_Y^2,$$
(3.11)

$$\|LK - \Pi_r^Y LK\|_{\mathrm{HS}(S,Y)}^2 = \sum_{k>r} \sigma_k^2 \|L\varphi_k - \Pi_r^Y L\varphi_k\|_Y^2,$$
(3.12)

and

$$\|K - L^{-1}\Pi_r^Y LK\|_{\mathrm{HS}(S,X)}^2 = \sum_{k>r} \sigma_k^2 \|\varphi_k - L^{-1}\Pi_r^Y L\varphi_k\|_X^2.$$
(3.13)

In the case $s_X = \infty$, the following convergence results hold. For (3.11): The error tends to zero as $r \to \infty$. For (3.12): If $\{\Pi_r^Y\}$ is uniformly bounded in operator norm, then the error goes to zero as $r \to \infty$. For (3.13): If L^{-1} is bounded and $\{\Pi_r^Y\}$ is uniformly bounded in operator norm, then the error tends to zero as $r \to \infty$. For (3.13): If $\{L^{-1}\Pi_r^Y L\}$ is uniformly bounded in operator norm, then the error converges to zero as $r \to \infty$.

Proof. Let $\{f_k\}$ be an orthonormal basis of *S* of eigenvectors of K^*K and let $\mathbb{J} = \{k : f_k \notin \ker(K^*K)\}$. Note that $Kf_k = 0$ for all $k \notin \mathbb{J}$, since $\ker(K^*K) = \ker(K)$ by Lemma 2.1.1. Also, $Kf_k = \sigma_k \varphi_k$ for all $k \in \mathbb{J}$. Then,

$$\begin{split} \|LK - L\Pi_r^X K\|_{\mathrm{HS}(S,Y)}^2 &= \sum_{k \ge 1} \|(LK - L\Pi_r^X K) f_k\|_Y^2 \\ &= \sum_{k \in \mathbb{J}} \|(LK - L\Pi_r^X K) f_k\|_Y^2 \\ &= \sum_{k \in \mathbb{J}} \|L\sigma_k \varphi_k - L\Pi_r^X \sigma_k \varphi_k\|_Y^2 \\ &= \sum_{k > r, k \in \mathbb{J}} \sigma_k^2 \|L\varphi_k\|_Y^2, \end{split}$$

where the last equality holds since $\Pi_r^X \varphi_k = \varphi_k$ for $k \le r$ and $\Pi_r^X \varphi_k = 0$ for k > r. Also,

$$\sum_{k>r, k\in \mathbb{J}} \sigma_k^2 \|L\varphi_k\|_Y^2 = \sum_{k>r, k\in \mathbb{J}} \|LKf_k\|_Y^2 = \sum_{k>r, k\in \mathbb{J}} \|K^Y f_k\|_Y^2,$$

which converges to zero as $r \to \infty$ since K^Y is Hilbert-Schmidt. Next,

$$\begin{split} \|LK - \Pi_r^Y LK\|_{\mathrm{HS}(S,Y)}^2 &= \sum_{k \ge 1} \|(LK - \Pi_r^Y LK)f_k\|_Y^2 \\ &= \sum_{k \in \mathbb{J}} \|L\sigma_k \varphi_k - \Pi_r^Y L\sigma_k \varphi_k\|_Y^2 \\ &= \sum_{k > r, \, k \in \mathbb{J}} \sigma_k^2 \|L\varphi_k - \Pi_r^Y L\varphi_k\|_Y^2, \end{split}$$

where the last equality holds since $\Pi_r^Y L \varphi_k = L \varphi_k$ for $k \leq r$. For convergence, note

$$\begin{split} \|LK - \Pi_r^Y LK\|_{\mathrm{HS}(S,Y)}^2 &= \sum_{k>r,\,k\in\mathbb{J}} \|LKf_k - \Pi_r^Y LKf_k\|_Y^2 \\ &= \sum_{k>r,\,k\in\mathbb{J}} \|K^Y f_k - \Pi_r^Y K^Y f_k\|_Y^2 \\ &\leq \sum_{k>r,\,k\in\mathbb{J}} \|I - \Pi_r^Y\|^2 \,\|K^Y f_k\|_Y^2. \end{split}$$

Since $||I - \Pi_r^Y||$ is uniformly bounded and K^Y is Hilbert-Schmidt, the error converges to zero as $r \to \infty$.

Since $L^{-1}\Pi_r^Y L\varphi_k = L^{-1}L\varphi_k = \varphi_k$ for $k \le r$, for the last equality we have

$$\begin{split} \|K - L^{-1} \Pi_r^Y L K\|_{\mathrm{HS}(S,X)}^2 &= \sum_{k \in \mathbb{J}} \|\sigma_k \varphi_k - L^{-1} \Pi_r^Y L \sigma_k \varphi_k\|_X^2 \\ &= \sum_{k > r, \, k \in \mathbb{J}} \sigma_k^2 \|\varphi_k - L^{-1} \Pi_r^Y L \varphi_k\|_X^2. \end{split}$$

Assuming L^{-1} is bounded and $\{\Pi_r^Y\}$ is uniformly bounded, the convergence follows

$$\|K - L^{-1} \Pi_r^Y L K\|_{\mathrm{HS}(S,X)}^2 = \sum_{k > r, \, k \in \mathbb{J}} \|L^{-1} (I - \Pi_r^Y) L K f_k\|_X^2$$
$$\leq \sum_{k > r, \, k \in \mathbb{J}} \|L^{-1}\|^2 \|I - \Pi_r^Y\|^2 \|K^Y f_k\|_Y^2$$

from

$$\|K - L^{-1} \Pi_r^Y L K\|_{\mathrm{HS}(S,X)}^2 = \sum_{k > r, k \in \mathbb{J}} \|(I - L^{-1} \Pi_r^Y L) K f_k\|_X^2$$
$$\leq \sum_{k > r, k \in \mathbb{J}} \|I - L^{-1} \Pi_r^Y L\|^2 \|K^Y f_k\|_X^2$$

which converges to zero as $r \to \infty$.

3.3.1. Discrete Case. First we introduce several representations that will be useful in the proof of Theorem 3.3.2 below. Recall, $s_X = \operatorname{rank}(K) < \infty$ is the number of nonzero POD singular values (or POD eigenvalues) for the data $\{w^j\}$. By the known POD error formula (2.5), we have

$$w^{j} = \Pi_{s_{X}}^{X} w^{j} = \sum_{k=1}^{s_{X}} (w^{j}, \varphi_{k})_{X} \varphi_{k}$$
 and $Lw^{j} = \sum_{k=1}^{s_{X}} (w^{j}, \varphi_{k})_{X} L\varphi_{k}$.

Note that since the sums are finite, $\{\varphi_k\} \subset \mathcal{D}(L)$, and *L* is linear we can pull *L* through the sums in this section without any additional assumptions. This is one point where the discrete and continuous cases differ.

Next, from Section 2.2.1 we know for all $j \le s$ and $k \le s_X$ we have

$$(w^j, \varphi_k)_X = \overline{(\varphi_k, w^j)_X} = \overline{(K^* \varphi_k)_j} = \sigma_k \overline{f_k^j},$$

where f_k^j denotes the *j*th component of the singular vector $f^k \in \mathbb{K}^s$. This gives

$$w^{j} = \sum_{k=1}^{s_{X}} \sigma_{k} \overline{f_{k}^{j}} \varphi_{k}$$
 and $Lw^{j} = \sum_{k=1}^{s_{X}} \sigma_{k} \overline{f_{k}^{j}} L\varphi_{k}.$ (3.14)

Also, recall $\{f^j\}$ are orthonormal in S, which yields

$$\sum_{j=1}^{s} \gamma_j \overline{f_k^j} f_\ell^j = (f_\ell, f_k)_S = \delta_{\ell,k}.$$

Theorem 3.3.2. The data approximation errors are given by

$$\sum_{j=1}^{s} \gamma_j \|Lw^j - L\Pi_r^X w^j\|_Y^2 = \sum_{k=r+1}^{s_X} \sigma_k^2 \|L\varphi_k\|_Y^2,$$
(3.15)

and

$$\sum_{j=1}^{s} \gamma_j \|Lw^j - \Pi_r^Y Lw^j\|_Y^2 = \sum_{k=r+1}^{s_X} \sigma_k^2 \|L\varphi_k - \Pi_r^Y L\varphi_k\|_Y^2.$$
(3.16)

Also, if L is invertible, then

$$\sum_{j=1}^{s} \gamma_j \|w^j - L^{-1} \Pi_r^Y L w^j\|_X^2 = \sum_{k=r+1}^{s_X} \sigma_k^2 \|\varphi_k - L^{-1} \Pi_r^Y L \varphi_k\|_X^2.$$
(3.17)

Proof. We only prove (3.16). The proofs of the other two results are similar. First, note we can apply Π_r^Y to Lw^j given in (3.14) to get

$$Lw^{j} - \Pi_{r}^{Y}w^{j} = \sum_{k=1}^{s_{X}} \sigma_{k} \overline{f_{k}^{j}} (L\varphi_{k} - \Pi_{r}^{Y}L\varphi_{k}).$$

Then

$$\begin{split} \sum_{j=1}^{s} \gamma_{j} \| Lw^{j} - \Pi_{r}^{Y} Lw^{j} \|_{Y}^{2} &= \sum_{j=1}^{s} \gamma_{j} \left(\sum_{k=1}^{s_{X}} \sigma_{k} \overline{f_{k}^{j}} (L\varphi_{k} - \Pi_{r}^{Y} L\varphi_{k}), \sum_{\ell=1}^{s_{X}} \sigma_{\ell} \overline{f_{\ell}^{j}} (L\varphi_{\ell} - \Pi_{r}^{Y} L\varphi_{\ell}) \right)_{Y} \\ &= \sum_{j=1}^{s} \gamma_{j} \sum_{\ell,k=1}^{s_{X}} \sigma_{k} \sigma_{\ell} \overline{f_{k}^{j}} f_{\ell}^{j} \left(L\varphi_{k} - \Pi_{r}^{Y} L\varphi_{k}, L\varphi_{\ell} - \Pi_{r}^{Y} L\varphi_{\ell} \right)_{Y} \\ &= \sum_{\ell,k=1}^{s_{X}} \sigma_{k} \sigma_{\ell} \left(\sum_{j=1}^{s} \gamma_{j} \overline{f_{k}^{j}} f_{\ell}^{j} \right) \left(L\varphi_{k} - \Pi_{r}^{Y} L\varphi_{k}, L\varphi_{\ell} - \Pi_{r}^{Y} L\varphi_{\ell} \right)_{Y} \\ &= \sum_{k=1}^{s_{X}} \sigma_{k}^{2} \left(L\varphi_{k} - \Pi_{r}^{Y} L\varphi_{k}, L\varphi_{k} - \Pi_{r}^{Y} L\varphi_{k} \right)_{Y} \end{split}$$

$$= \sum_{k=1}^{s_X} \sigma_k^2 \| L\varphi_k - \Pi_r^Y L\varphi_k \|_Y^2.$$

Note that $\Pi_r^Y L \varphi_k = L \varphi_k$ for k = 1, ..., r since Π_r^Y is a projection onto $Y_r = \text{span}\{L \varphi_k\}_{k=1}^r$. Therefore,

$$\sum_{j=1}^{s} \gamma_j \|Lw^j - \Pi_r^Y Lw^j\|_Y^2 = \sum_{k=r+1}^{s_X} \sigma_k^2 \|L\varphi_k - \Pi_r^Y L\varphi_k\|_Y^2.$$

In Corollary 3.4.10, we focus on error bounds for approximating each individual data snapshot w^{ℓ} with various POD projections. Also, another way to prove Theorem 3.3.2 is to use the Hilbert Schmidt norm results in Lemma 3.3.1. The proof we give above requires less background. However, we do require Lemma 3.3.1 for the continuous case below.

3.3.2. Continuous Case. For the continuous case we must consider the possibility that the number of nonzero POD eigenvalues is infinite. We approach this case differently from the discrete case above. We show each of the data approximation errors we consider is equal to one of the Hilbert-Schmidt norm errors from Lemma 3.3.1. Then we use that result to prove the convergence of the errors to zero in the case of an infinite number of nonzero POD eigenvalues.

For one case, we need to make an additional assumption on L^{-1} .

The L^{-1} assumption: We assume

- 1. $s_X < \infty$, or
- 2. $L^{-1}\Pi_r^Y LKf = \sum_{i=1}^m \int_O f^i(t) L^{-1}\Pi_r^Y Lw^i(t) dt$ for all $f \in S$.

Remark 3.3.3. Note that if $s_X < \infty$, then the proof technique in Section 3.3.1 above can be used for the continuous cases, with some minor modifications to deal with the change in the space S. The second condition is similar to the main assumption made in Section 3.1.2. Any of the three common conditions in Remark 3.1.1 that guarantee the main assumption holds also imply that the second condition in the L^{-1} assumption holds.

Theorem 3.3.4. The data approximation errors are given by

$$\sum_{j=1}^{m} \|Lw^{j} - L\Pi_{r}^{X}w^{j}\|_{L^{2}(O;Y)}^{2} = \sum_{k>r} \sigma_{k}^{2} \|L\varphi_{k}\|_{Y}^{2}$$
(3.18)

and

$$\sum_{j=1}^{m} \|Lw^{j} - \Pi_{r}^{Y}Lw^{j}\|_{L^{2}(O;Y)}^{2} = \sum_{k>r} \sigma_{k}^{2} \|L\varphi_{k} - \Pi_{r}^{Y}L\varphi_{k}\|_{Y}^{2}.$$
(3.19)

Also if the L^{-1} assumption holds then

$$\sum_{j=1}^{m} \|w^{j} - L^{-1} \Pi_{r}^{Y} L w^{j}\|_{L^{2}(O;X)}^{2} = \sum_{k>r} \sigma_{k}^{2} \|\varphi_{k} - L^{-1} \Pi_{r}^{Y} L \varphi_{k}\|_{X}^{2}.$$
 (3.20)

In the case $s_X = \infty$, the following convergence results hold. For Equation (3.18): The error tends to zero as $r \to \infty$. For Equation (3.19): If $\{\Pi_r^Y\}$ is uniformly bounded in operator norm, then the error goes to zero as $r \to \infty$. For (3.20): If L^{-1} is bounded and $\{\Pi_r^Y\}$ is uniformly bounded in operator norm, then the error tends to zero as $r \to \infty$. For Equation (3.20): If $\{L^{-1}\Pi_r^Y L\}$ is uniformly bounded in operator norm, then the error converges to zero as $r \to \infty$.

Remark 3.3.5. Note that the conditions for convergence for the case $s_X = \infty$ are exactly the conditions given in Lemma 3.3.1.

Proof. We prove (3.18), and the associated convergence result. The proofs of the other equalities and convergence results are similar. We first show that the data approximation error has an integral representation, and then we use the two Hilbert-Schmidt results for POD operators to conclude.

By definition, for $f \in S$ we have

$$L\Pi_r^X Kf = \sum_{k=1}^r (Kf, \varphi_k)_X L\varphi_k$$
$$= \sum_{k=1}^r \left(\sum_{j=1}^m \int_O f^j(t) w^j(t) dt, \varphi_k \right)_X L\varphi_k$$

$$= \sum_{j=1}^{m} \int_{O} f^{j}(t) \sum_{k=1}^{r} (w^{j}(t), \varphi_{k})_{X} L \varphi_{k} dt$$
$$= \sum_{j=1}^{m} \int_{O} f^{j}(t) L w_{r}^{j}(t) dt,$$

where $w_r^j(t) = \prod_r^X w^j(t) = \sum_{k=1}^r (w^j(t), \varphi_k)_X \varphi_k$. Because of the main assumption, we can pull the operator *L* inside the integral to give

$$(LK - L\Pi_r^X K)f = \int_O \sum_{j=1}^m f^j(t) [Lw^j(t) - Lw_r^j(t)]dt.$$

Since $Lw^j - Lw^j_r \in L^2(O; Y)$ for each *j*, by Lemma 2.2.4 we have

$$\sum_{j=1}^{m} \|Lw^{j} - L\Pi_{r}^{X}w^{j}\|_{L^{2}(O;Y)}^{2} = \|LK - L\Pi_{r}^{X}K\|_{\mathrm{HS}(S,Y)}^{2}.$$

Lemma 3.3.1 proves both (3.18) and the convergence result in the case $s_X = \infty$.

Note for (3.20), for $f \in S$ the L^{-1} assumption gives

$$L^{-1}\Pi_r^Y LKf = \int_O \sum_{j=1}^m f^j(t) L^{-1}\Pi_r^Y Lw^j(t) dt, \qquad (3.21)$$

and then we proceed similarly to establish the result.

3.4. POINTWISE CONVERGENCE OF POD PROJECTIONS

Recall that $\{\varphi_k\}$ is an orthonormal basis for X, and therefore $\|\Pi_r^X x - x\|_X \to 0$ for all $x \in X$. In this section, we prove various types of pointwise convergence results for the other POD projections; namely, Π_r^Y from Section 3.2, and $L\Pi_r^X L^{-1}$ and $L^{-1}\Pi_r^Y L$ from Section 3.2.1. The majority of this section is not split into the discrete and continuous cases because the proofs are similar for both, and many of the results hold regardless of case. We do focus on the discrete case at the end of this section and address some assumptions made in the literature about approximations of each individual data snapshot using POD projections. Pointwise convergence results for these POD projections are easiest to obtain when L and L^{-1} are both bounded. We primarily focus on the case when either L or L^{-1} is unbounded.

Range conditions are an important factor in this section. When an element to be approximated by a POD projection is in the range of K or K^Y , we can often get better results. When certain conditions hold, we know these ranges exactly. Recall from Lemma 3.2.5, if all the POD eigenvalues for $\{w^j\}$ are nonzero and $s_X < \infty$, then we know $X = \mathcal{R}(K)$ and dim $(X) = s_X$. Note that in this case, the Hilbert space X must be finite dimensional. If all the POD eigenvalues for $\{w^j\}$ are nonzero and $s_X = \infty$ (i.e., X must be infinite dimensional), then Lemma 3.2.5 only gives $X = \overline{\mathcal{R}(K)}$. We do not always obtain the better convergence results in this case. Similar statements hold for the spaces Y and $\mathcal{R}(K^Y)$. Also, as in Section 3.3, we sometimes need to consider different proof techniques in the case $s_X = \infty$.

We begin with a pointwise convergence result for Π_r^Y assuming *L* is bounded. For another pointwise convergence result for Π_r^Y with different assumptions, see Theorem 3.4.5 below.

Theorem 3.4.1. Assume *L* is bounded and $\{\Pi_r^Y\}$ is uniformly bounded in operator norm. If $y \in \mathcal{R}(L)$, then $\Pi_r^Y y \to y$ as *r* increases. In addition, if $\mathcal{R}(L)$ is dense in *Y*, then $\Pi_r^Y y \to y$ for all $y \in Y$.

Proof. Let $y \in \mathcal{R}(L)$, so that y = Lx for some $x \in X$. Note that since $L\Pi_r^X x \in Y_r = \text{span}\{L\varphi_k\}_{k=1}^r$ and Π_r^Y is a projection onto Y_r , we have $\Pi_r^Y L\Pi_r^X x = L\Pi_r^X x$. Then

$$\begin{split} \|\Pi_{r}^{Y}y - y\|_{Y} &\leq \|\Pi_{r}^{Y}Lx - \Pi_{r}^{Y}L\Pi_{r}^{X}x\|_{Y} + \|\Pi_{r}^{Y}L\Pi_{r}^{X}x - Lx\|_{Y} \\ &= \|\Pi_{r}^{Y}Lx - \Pi_{r}^{Y}L\Pi_{r}^{X}x\|_{Y} + \|L\Pi_{r}^{X}x - Lx\|_{Y} \\ &\leq \|\Pi_{r}^{Y}L\|\|x - \Pi_{r}^{X}x\|_{Y} + \|L\|\|\Pi_{r}^{X}x - x\|_{Y}, \end{split}$$

which converges to zero as r increases since $\Pi_r^X x \to x$ and $\{\Pi_r^Y\}$ is uniformly bounded in operator norm. The final result follows directly from the Banach-Steinhaus theorem (i.e., the principle of uniform boundedness).

The next convergence result relies on the boundedness of either L or L^{-1} and certain range conditions involving L.

- 1. For any $y \in \mathcal{R}(L) = \mathcal{D}(L^{-1})$, if L is bounded, then $||L\Pi_r^X L^{-1} y -$ Theorem 3.4.2. $y \parallel_Y \to 0$ as r increases. In addition, if $\mathcal{R}(L)$ is dense in Y and $\{L \Pi_r^X L^{-1}\}$ is uniformly bounded, then $L\Pi_r^X L^{-1} y \rightarrow y$ for all $y \in Y$.
 - 2. For any $x \in \mathcal{D}(L) = \mathcal{R}(L^{-1})$, if L^{-1} is bounded and $\prod_r^Y y \to y$ for all $y \in Y$ as r increases, then $\|L^{-1}\Pi_r^Y Lx - x\|_X \to 0$ as r increases. In addition, if $\mathcal{D}(L)$ is dense in X and $\{L^{-1}\Pi_r^Y L\}$ is uniformly bounded, then $L^{-1}\Pi_r^Y Lx \to x$ for all $x \in X$.

Remark 3.4.3. Note that Theorem 3.4.1 and Theorem 3.4.5 give two cases where the assumption $\Pi_r^Y y \to y$ for all $y \in Y$ holds. Also, the uniform boundedness of $\{L\Pi_r^X L^{-1}\}$ and $\{L^{-1}\Pi_r^Y L\}$ is not currently known, unless L and L^{-1} are both bounded. Note that when L and L^{-1} are both bounded, Theorem 3.4.1 gives $\Pi_r^Y y \to y$ for all $y \in Y$ whenever $\{\Pi_r^Y\}$ is uniformly bounded; therefore, in this case Theorem 3.4.2 gives $L\Pi_r^X L^{-1} y \rightarrow y$ for all $y \in Y$ and $L^{-1}\Pi_r^Y Lx \to x$ for all $x \in X$.

Proof. We only prove the first result; the proof of the second is similar. Since $y \in \mathcal{R}(L)$ we have y = Lx for some $x \in X$. Then

$$\|L\Pi_r^X L^{-1} y - y\|_Y = \|L\Pi_r^X x - Lx\|_Y$$

$$\leq \|L\| \|\Pi_r^X x - x\|_X,$$

which converges to zero as r increases. The final convergence result again follows from the principle of uniform boundedness. Next, we consider how range conditions involving K and K^Y affect the convergence of POD projections. We are able to obtain convergence rates, and at most require either Lor L^{-1} to be bounded. We begin with the POD projection Π_r^Y and then consider $L\Pi_r^X L^{-1}$ and $L^{-1}\Pi_r^Y L$. We use the following simple lemma multiple times below.

Lemma 3.4.4. Assume $y \in \mathcal{R}(K^Y)$ so that $y = K^Y g = LKg$ for some $g \in S$. If

$$y_N = LK_N g = L\Pi_N^X K g, \qquad (3.22)$$

then $y_N \rightarrow y$ as N increases.

Proof. As N increases,

$$||y_N - y||_Y = ||LKg - LK_Ng||_Y \le ||LK - L\Pi_N^X K||_{\mathrm{HS}(S,Y)} ||g||_S \to 0$$

by Lemma 3.3.1.

Recall from Lemma 3.2.5 that s_Y is always less than or equal to s_X . Thus if we assume $s_X < \infty$, we know that $s_Y < \infty$. For the following proofs, we consider whether s_X is finite or infinite.

Theorem 3.4.5. Assume $\{\Pi_r^Y\}$ is uniformly bounded in operator norm whenever $s_X = \infty$. If $y = K^Y g$ for some $g \in S$, then $\Pi_r^Y y \to y$ as r increases and the following error bound holds:

$$\|\Pi_{r}^{Y} y - y\|_{Y} \le \sum_{k>r} \sigma_{k} |(g, f_{k})_{S}| \, \|\Pi_{r}^{Y} L\varphi_{k} - L\varphi_{k}\|_{Y}.$$
(3.23)

Also, if the POD eigenvalues for the data $\{Lw^j\}$ are all nonzero, then $\Pi_r^Y y \to y$ for all $y \in Y$.

Proof. First consider the case $s_X < \infty$, and fix *r*. Assume $y = K^Y g = LKg$ for some $g \in S$. Thus,

$$\Pi_r^Y y = \Pi_r^Y K^Y g = \Pi_r^Y L K g = \sum_{k=1}^{s_X} \sigma_k(g, f_k)_S \Pi_r^Y L \varphi_k, \quad \text{and} \quad y = \sum_{k=1}^{s_X} \sigma_k(g, f_k)_S L \varphi_k.$$
(3.24)

Subtracting gives

$$\Pi_r^Y y - y = \sum_{k=1}^{s_X} \sigma_k(g, f_k)_S (\Pi_r^Y L \varphi_k - L \varphi_k) = \sum_{k=r+1}^{s_X} \sigma_k(g, f_k)_S (\Pi_r^Y L \varphi_k - L \varphi_k),$$

since $\Pi_r^Y L \varphi_k = L \varphi_k$ for k = 1, ..., r. The error bound (3.23) follows directly from this representation and the triangle inequality. Furthermore, since $s_X < \infty$, clearly $\Pi_r^Y y \to y$ as r increases for each $y \in \mathcal{R}(K^Y)$.

Next, assume the POD eigenvalues for the data $\{Lw^j\}$ are all nonzero. By Item 2 of Lemma 3.2.5, since $s_Y \le s_X < \infty$ we have $Y = \mathcal{R}(K^Y)$. This gives $\prod_r^Y y \to y$ for all $y \in Y$.

Now consider the case $s_X = \infty$, and fix *r*. For $y = K^Y g = LKg$ with $g \in S$ as above, recall the definition of $y_N = L\Pi_N^X Kg$ given in (3.22). We have

$$\|\Pi_{r}^{Y}y - y\|_{Y} \leq \|\Pi_{r}^{Y}y - \Pi_{r}^{Y}y_{N}\|_{Y} + \|\Pi_{r}^{Y}y_{N} - y_{N}\|_{Y} + \|y_{N} - y\|_{Y}$$
$$\leq \left(\|\Pi_{r}^{Y}\| + 1\right)\|y - y_{N}\|_{Y} + \|\Pi_{r}^{Y}y_{N} - y_{N}\|_{Y}.$$

Note that for the second term, $\|\Pi_r^Y y_N - y_N\|_Y$, we can obtain representations for $\Pi_r^Y y_N$ and y_N similar to that in (3.24) above. Proceeding in the same way gives

$$\|\Pi_r^Y y_N - y_N\|_Y \leq \sum_{k=r+1}^N \sigma_k |(g, f_k)_S| \|\Pi_r^Y L\varphi_k - L\varphi_k\|_Y.$$

Since r is fixed and $y_N \rightarrow y$ as $N \rightarrow \infty$ (Lemma 3.4.4), the two inequalities above give

$$\|\Pi_r^Y y - y\|_Y \le \sum_{k=r+1}^{\infty} \sigma_k |(g, f_k)_S| \|\Pi_r^Y L\varphi_k - L\varphi_k\|_Y.$$

For convergence, we have

$$\|\Pi_{r}^{Y}y - y\|_{Y} \leq \left(\sum_{k>r} |(g, f_{k})_{S}|^{2}\right)^{1/2} \left(\sum_{k>r} \sigma_{k}^{2} \|\Pi_{r}^{Y}L\varphi_{k} - L\varphi_{k}\|_{Y}^{2}\right)^{1/2}$$

Since $\{f_k\}$ is an orthonormal basis for *S*, we know $\sum_{k>r} |(g, f_k)_S|^2$ goes to zero as *r* increases by Parseval's equality. Furthermore, since $\{\Pi_r^Y\}$ is uniformly bounded, Lemma 3.3.1 gives that $\sum_{k>r} \sigma_k^2 ||\Pi_r^Y L\varphi_k - L\varphi_k||_Y^2$ goes to zero as *r* increases. This gives $\Pi_r^Y y \to y$ for each $y \in \mathcal{R}(K^Y)$.

Finally, assume the POD eigenvalues for the data $\{Lw^j\}$ are all nonzero. By Item 2 of Lemma 3.2.5, we have $\mathcal{R}(K^Y)$ is dense in *Y*. Since $\{\Pi_r^Y\}$ is uniformly bounded, the principle of uniform boundedness gives $\Pi_r^Y y \to y$ for all $y \in Y$.

For the next two results we need to assume L or L^{-1} is bounded whenever $s_X = \infty$. **Theorem 3.4.6.** Assume $s_X < \infty$, or either L or L^{-1} is bounded. If $y = K^Y g$ for some $g \in S$, then

$$\|y - L\Pi_r^X L^{-1} y\|_Y \le \sum_{k>r} \sigma_k |(g, f_k)_S| \|L\varphi_k\|_Y$$
(3.25)

and the error converges to zero as r increases. Now assume $\{L\Pi_r^X L^{-1}\}$ is uniformly bounded in operator norm whenever $s_X = \infty$. If the POD eigenvalues for the data $\{Lw^j\}$ are all nonzero, then $L\Pi_r^X L^{-1} y \to y$ for all $y \in Y$.

Proof. Let y = LKg for some $g \in S$, assume $s_X < \infty$, and fix r. As in the proof of Theorem 3.4.5, it can be shown that

$$y - L\Pi_r^X L^{-1} y = LKg - L\Pi_r^X Kg$$
$$= \sum_{k=1}^{s_X} \sigma_k(g, f_k) S L\varphi_k - \sum_{k=1}^r \sigma_k(g, f_k) S L\varphi_k$$
$$= \sum_{k=r+1}^{s_X} \sigma_k(g, f_k) S L\varphi_k.$$

The triangle inequality gives the error bound (3.25). The convergence results for the case $s_X < \infty$ follow just as in the proof of Theorem 3.4.5.

Now consider the case $s_X = \infty$, assume y = LKg for some $g \in S$, and fix r. Then for $y_N = L\Pi_N^X Kg$ as in (3.22), with $N \ge r$, we have

$$\|y - L\Pi_r^X L^{-1}y\|_Y \le \|y - y_N\|_Y + \|y_N - L\Pi_r^X L^{-1}y_N\|_Y + \|L\Pi_r^X L^{-1}y_N - L\Pi_r^X L^{-1}y\|_Y.$$

Lemma 3.4.4 implies that the first term tends to zero as $N \to \infty$. For the second term, proceed as above and use $\prod_r^X \prod_N^X = \prod_r^X$ (since $N \ge r$) to show

$$\|y_N - L\Pi_r^X L^{-1} y_N\|_Y \le \sum_{k=r+1}^N \sigma_k |(g, f_k)_S| \|L\varphi_k\|_Y.$$

For the third term, first assume *L* is bounded. In this case,

$$\|L\Pi_r^X L^{-1} y_N - L\Pi_r^X L^{-1} y\|_Y \le \|L\| \|\Pi_r^X\| \|L^{-1} y_N - L^{-1} y\|_X = \|L\| \|\Pi_r^X\| \|\Pi_N^X Kg - Kg\|_X,$$

which converges to zero as $N \to \infty$, since *r* is fixed. If instead L^{-1} is bounded, then $L \prod_r^X L^{-1}$ is bounded by Theorem 3.2.3 and so

$$\|L\Pi_r^X L^{-1} y_N - L\Pi_r^X L^{-1} y\|_Y \le \|L\Pi_r^X L^{-1}\| \|y_N - y\|_Y,$$

which converges to zero as $N \to \infty$ by Lemma 3.4.4, again since *r* is fixed. Combining the above results gives

$$||y - L\Pi_r^X L^{-1} y||_Y \le \sum_{k=r+1}^{\infty} \sigma_k |(g, f_k)_S| ||L\varphi_k||_Y.$$

For convergence, we proceed as in the proof of Theorem 3.4.5. We have

$$\|y - L\Pi_r^X L^{-1} y\|_Y \le \left(\sum_{k>r} |(g, f_k)_S|^2\right)^{1/2} \left(\sum_{k>r} \sigma_k^2 \|L\varphi_k\|_Y^2\right)^{1/2}$$

We know $\sum_{k>r} |(g, f_k)_S|^2$ goes to zero as r increases by Parseval's equality. Furthermore, Lemma 3.3.1 gives that $\sum_{k>r} \sigma_k^2 ||L\varphi_k||_Y^2$ goes to zero as r increases. This implies $L\Pi_r^X L^{-1}y \to y$ for each $y \in \mathcal{R}(K^Y)$. To show convergence for all $y \in Y$, we again use Item 2 of Lemma 3.2.5 and the principle of uniform boundedness.

We omit the proof of the next result, as it is similar to the proof of the previous result, Theorem 3.4.6. Note that in Theorem 3.4.6 the error converges to zero for a fixed $y \in \mathcal{R}(K^Y)$ without any additional assumptions. In this next result, if $s_X = \infty$ we need to require additional conditions to guarantee that the error converges to zero for a fixed $x \in \mathcal{R}(K)$; these conditions come from Lemma 3.3.1.

Theorem 3.4.7. Assume $s_X < \infty$ or either L or L^{-1} is bounded. If x = Kg for some $g \in S$, then

$$\|x - L^{-1}\Pi_r^Y Lx\|_X \le \sum_{k>r} \sigma_k |(g, f_k)_S| \|\varphi_k - L^{-1}\Pi_r^Y L\varphi_k\|_X.$$
(3.26)

If $s_X < \infty$, the error converges to zero as r increases. If $s_X = \infty$, then the error goes to zero as r increases when either (i) L^{-1} is bounded and $\{\Pi_r^Y\}$ is uniformly bounded or (ii) $\{L^{-1}\Pi_r^Y L\}$ is uniformly bounded. Now assume $\{L^{-1}\Pi_r^Y L\}$ is uniformly bounded in operator norm whenever $s_X = \infty$. If the POD eigenvalues for the data $\{w^j\}$ are all nonzero, then $L^{-1}\Pi_r^Y Lx \to x$ for all $x \in X$.

To be complete, we give an exact error formula and an error bound for approximations of elements in the range of *K* using the POD projection Π_r^X . This result gives an error bound for approximating each individual data snapshot in the discrete case.

Theorem 3.4.8. If x = Kg for some $g \in S$, then

$$\|x - \Pi_r^X x\|_X = \left(\sum_{k>r} \sigma_k^2 |(g, f_k)_S|^2\right)^{1/2} \le \sigma_{r+1} \|g\|_S.$$
(3.27)

Also, in the discrete case, for each $\ell = 1, ..., s$ we have

$$\|w^{\ell} - \Pi_r^X w^{\ell}\|_X \le \gamma_{\ell}^{-1/2} \sigma_{r+1}.$$
(3.28)

Remark 3.4.9. The bound (3.28) was obtained in [40, Proposition 3.1] for $X = \mathbb{R}^n$ and $\gamma_{\ell} = 1$ for all ℓ . Recall the constants $\{\gamma_{\ell}\}$ are the positive weights in the definition of the POD operator K in the discrete case; see Section 2.2.1.

Proof. Using the SVD of K gives

$$x - \prod_r^X x = \sum_{k>r} \sigma_k(g, f_k)_S \varphi_k.$$

Since $||x - \Pi_r^X x||_X^2 = (x - \Pi_r^X x, x - \Pi_r^X x)_X$ and $\{\varphi_k\}$ is an orthonormal basis for *X*, we immediately obtain the exact error formula in (3.27). To obtain the error bound in (3.27), use $\sigma_k \leq \sigma_{r+1}$ for all k > r and also Parseval's equality.

Next, in the discrete case we have $w^{\ell} = Kg_{\ell}$ for each $\ell = 1, ..., s$, where $g_{\ell} = \gamma_{\ell}^{-1}e_{\ell}$ and e_{ℓ} is the ℓ th standard unit vector for \mathbb{K}^{s} , i.e., the ℓ th entry of e_{ℓ} is one and all other entries are zero. The error bound (3.28) follows from $||g^{\ell}||_{s} = \gamma_{\ell}^{-1/2}$ and (3.27).

In Theorem 3.4.8, note that the quantity γ_{ℓ}^{-1} appears in the error bound (3.28) for approximating the snapshot w^{ℓ} . However, in applications it is typical that each weight γ_{ℓ} tends to zero as the number *s* of snapshots increases. Next, we use the above results to prove various approximation error bounds for each individual snapshot w^{ℓ} in the discrete case that do not depend on γ_{ℓ}^{-1} . Here, the bounds are only valid if *r* is sufficiently large. We note that these type of error bounds have been *assumed* to hold in the literature; Iliescu and Wang made this type of assumption in [25, Assumption 3.2] (with $\gamma_{\ell} = s^{-1}$ for all ℓ) in their analysis of a POD reduced order model of the Navier-Stokes equations, and many others have followed their approach.

Corollary 3.4.10. In the discrete case, if r is sufficiently large, then for each $\ell = 1, ..., s$ we have

$$\|w^{\ell} - \Pi_{r}^{X} w^{\ell}\|_{X}^{2} \le \sigma_{r+1}^{2}, \tag{3.29a}$$

$$\|Lw^{\ell} - \Pi_r^Y Lw^{\ell}\|_Y^2 \le \sum_{k>r} \sigma_k^2 \|L\varphi_k - \Pi_r^Y L\varphi_k\|_Y^2,$$
(3.29b)

$$\|Lw^{\ell} - L\Pi_r^X w^{\ell}\|_Y^2 \le \sum_{k>r} \sigma_k^2 \|L\varphi_k\|_Y^2,$$
(3.29c)

$$\|w^{\ell} - L^{-1}\Pi_{r}^{Y}Lw^{\ell}\|_{X}^{2} \leq \sum_{k>r} \sigma_{k}^{2}\|\varphi_{k} - L^{-1}\Pi_{r}^{Y}L\varphi_{k}\|_{X}^{2}.$$
(3.29d)

Proof. We only prove (3.29b); the proofs of the remaining inequalities are similar. As in the proof of Theorem 3.4.8, we know $w^{\ell} = Kg_{\ell}$ for each $\ell = 1, ..., s$, where $g_{\ell} = \gamma_{\ell}^{-1}e_{\ell}$. Using the error bound (3.23) in Theorem 3.4.5, the Cauchy-Schwarz inequality on the sum, and Parseval's inequality gives

$$\|Lw^{\ell} - \Pi_{r}^{Y}Lw^{\ell}\|_{Y}^{2} \leq \|g^{\ell} - \Pi_{r}^{S}g_{\ell}\|_{S}^{2} \sum_{k>r} \sigma_{k}^{2}\|L\varphi_{k} - \Pi_{r}^{Y}L\varphi_{k}\|_{Y}^{2},$$

where $\Pi_r^S : S \to S$ is the orthogonal projection onto $S_r := \operatorname{span}\{f_k\}_{k=1}^r$. Since $\{f_k\}_{k\geq 1}$ is an orthonormal basis for S, we know $\Pi_r^S g_\ell \to g_\ell$ for $\ell = 1, \ldots, s$. Since s is fixed, for all sufficiently large r we have $||g_\ell - \Pi_r^S g_\ell||_S \le 1$ for all $\ell = 1, \ldots, s$, and this completes the proof.

3.5. EXAMPLES

In this section we present four examples illustrating these new results. Each example shows how to consider the problem in terms of our new framework. First, we consider a computational example in Section 3.5.1 to demonstrate the POD data approximation errors. Then we consider a few additional examples that relate to previous works.

For the examples in Section 3.5.2 and Section 3.5.3, we consider two separable Hilbert spaces, H and V, where V is a proper subset of H, and V is both continuously embedded¹ and dense in H. The linear operator L is a mapping between these two spaces. We assume we have the data $\{w^j\}_{j=1}^m \subset L^2(O; H) \cap L^2(O; V)$. We present the results for the continuous case only, but results for the discrete case can also be obtained using the theory in this work if desired.

3.5.1. Computational Example. Next, we briefly present numerical results for an example to demonstrate our new results. POD model order reduction is considered for this example in [29]; here, we focus on the POD data approximation errors. The new results are discussed in greater detail for other examples in Section 3.5.

Consider a nerve impulse model, the FitzHugh-Nagumo system in one dimension. The initial conditions for this model are zero and is given by

$$\begin{split} \frac{\partial u(t,x)}{\partial t} &= \mu \frac{\partial^2 u(t,x)}{\partial x^2} - \frac{1}{\mu} v(t,x) + \frac{1}{\mu} f(u) + \frac{c}{\mu}, \quad 0 < x < 1, \\ \frac{\partial v(t,x)}{\partial t} &= b u(t,x) - \gamma v(t,x) + c, \quad 0 < x < 1, \end{split}$$

where

$$f(u) = u(u - 0.1)(1 - u),$$

 $\mu = 0.015, b = 0.5, \gamma = 2$, and c = 0.05. Further, the boundary conditions are given by

$$u_x(t,0) = -50000t^3 e^{-15t}$$
 and $u_x(t,1) = 0$.

¹i.e., there exists a constant $C_V > 0$ such that $||v||_H \le C_V ||v||_V$ for all $v \in V$

For this example, we take the Hilbert spaces $X = Y = L^2(0, 1) \times L^2(0, 1)$ with the usual inner product, and define the operator $L : X \to Y$ by

$$L\begin{bmatrix} u\\ v \end{bmatrix} = \begin{bmatrix} \partial_x u\\ \partial_x v \end{bmatrix}$$

Note that here *L* is unbounded and closed, but not invertible. Thus, this operator satisfies the main assumption made for the continuous case. We let Π_r^Y be the orthogonal projection onto $Y_r = \text{span}\{L\varphi_k\}_{k=1}^r$, where $\{\varphi_k\} \subset X$ are the POD modes.

To approximate the solution of the PDE we used the interpolated coefficient finite element method with continuous piecewise linear basis functions from [29], and ode23s from MATLAB for the time stepping scheme. We approximated the solution using 100 equally spaced finite element nodes on the time interval O = (0, 10). Increasing the number of finite element nodes gave similar results below.

For the POD computations, the solution values were approximated at each time step, $w(t_k)$, where $w = [u, v]^T$, and a piecewise constant function in time was formed. The constant on each interval is given by the average of the solution at the current step and the solution at the next step, i.e., $0.5(w(t_{k+1}) + w(t_k))$. Note that for this problem we can calculate the POD eigenvalues, POD modes, and the data approximation errors exactly. Thus, comparisons between the actual approximation errors and the error formulas can be made.

In Tables 3.1 and 3.2 we present the errors from the relevant projections considered in Chapter 3 for r = 4 and r = 12. Note that errors for projections involving the inverse mapping L^{-1} are not included since L is not invertible for this example. In the tables, the actual error is the integral error measure and the error formula is the sum involving the POD singular values. The first line in the tables represents computations for the known error result (2.7). The second and third lines of the tables are computations for the new results

POD Error Equation	Actual Error	Error Formula	Difference
Equation (2.7)	6.2755×10^{-5}	6.2792×10^{-5}	3.7584×10^{-8}
Equation (3.18)	2.1584×10^{-1}	2.1593×10^{-1}	9.1863×10^{-5}
Equation (3.19)	9.8536×10^{-3}	9.8541×10^{-3}	4.7712×10^{-7}

Table 3.1. Error Comparison with r = 4

Table 3.2. Error Comparison with r = 12

POD Error Formula	Actual Error	Error Formula	Difference
Equation (2.7)	4.1453×10^{-8}	4.1487×10^{-8}	3.3661×10^{-11}
Equation (3.18)	2.2536×10^{-4}	2.2541×10^{-4}	5.2146×10^{-8}
Equation (3.19)	1.2664×10^{-5}	1.2668×10^{-5}	3.5150×10^{-9}

(3.18)-(3.19). The second line of each table gives the values for

actual error
$$= \int_O \|Lw(t) - L\Pi_r^X w(t)\|_Y^2 dt$$
, error formula $= \sum_{k>r} \sigma_k^2 \|L\varphi_k\|_Y^2$,

while the third line of each table shows computational results for

actual error
$$= \int_O \|Lw^j - \Pi_r^Y Lw^j\|_Y^2 dt$$
, error formula $= \sum_{k>r} \sigma_k^2 \|L\varphi_k - \Pi_r^Y L\varphi\|_Y^2$.

The differences in the computed values are likely due to round off errors. Note that as r increases the errors tend toward zero, as expected by the theory.

3.5.2. Examples From [1]. The first two examples are from [1]. Due to the above assumption on the data, the POD operator *K* can be viewed as a mapping into *H* or a mapping into *V*. One can obtain the SVD of $K : S \rightarrow H$ or the SVD of $K : S \rightarrow V$, i.e., one can choose X = H or X = V. The different choices for *X* give different POD singular values, POD singular vectors, POD modes, and POD projections. In [1], the author considered

both choices for X and four different POD projections between these spaces and gave exact expressions for the POD data approximation errors in the two different Hilbert space norms. We relate the notation and results for both the error formulas and pointwise convergence from the present work to [1]. We obtain better pointwise convergence results in this work. Also, O was only an interval in [1], but now we have O is an open subset of \mathbb{R}^d . For these first two examples, $Y_r = \text{span}\{L\varphi_k\}$ and $\Pi_r^Y : Y \to Y$ is the orthogonal projection onto Y_r . Note this implies $\{\Pi_r^Y\}$ is uniformly bounded in operator norm.

3.5.2.1. Example 1. For the first example, consider the case where X = H, Y = V, and $L : H \to V$ is defined by Lv = v for all $v \in \mathcal{D}(L) = V$. The operator L is clearly invertible, and $L^{-1} : V \to H$ is given by $L^{-1}v = v$ for all $v \in V$. Note that $L^{-1} : V \to H$ is bounded due to the continuous embedding assumption. Also, the inverse of a bounded operator is closed, so L is closed. Furthermore, the assumption on the data gives $\{w^j\} \subset L^2(O; X)$ and $\{Lw^j\} \subset L^2(O; Y)$. Thus, we know that both the main assumption and the L^{-1} assumption hold.

Since X = H and each set of singular vectors of the POD operator $K : S \to H$ are an orthonormal basis, we know the POD modes $\{\varphi_k\}$ are an orthonormal basis for H. Note that $X_r = \text{span}\{\varphi_k\}_{k=1}^r \subset H$, and $Y_r = \text{span}\{L\varphi_k\}_{k=1}^r = \text{span}\{\varphi_k\}_{k=1}^r \subset V$. Furthermore, the POD modes $\{\varphi_k\}$ may not be orthogonal in V. Also, the operator $K^Y = LK$ is simply the POD operator K viewed as a mapping from S to V. We take $\Pi_r^X : X \to X$ to be the orthogonal projection onto X_r , and $\Pi_r^Y : Y \to Y$ to be the orthogonal projection onto Y_r .

In order to discuss the POD projections we pay special attention to the spaces under consideration. Since $V \subset H$, the projections can be considered as mappings from V to V or from H to H. The projections considered in this work are related to the projections P_r^H and P_r^V in [1, Definition 3.2] as follows:

- $\Pi_r^X : X \to X$ is equal to the orthogonal projection $P_r^H : H \to H$.
- $\Pi_r^Y: Y \to Y$ is equal to the orthogonal projection $P_r^V: V \to V$.

- $L\Pi_r^X L^{-1}: Y \to Y$ is equal to the operator $P_r^H: V \to V$.
- $L^{-1}\Pi_r^Y L: X \to X$ is equal to the operator $P_r^V: H \to H$.

Now that we have the relationships between the projections, we compare the results. The error formulas presented here in Theorem 3.3.4 are essentially the same as the results in [1]. Again, the primary difference here is that O is an open subset of \mathbb{R}^d instead of an interval. The POD data approximation errors from Theorem 3.3.4 become the following:

$$\sum_{j=1}^{m} \int_{O} \|w^{j}(t) - P_{r}^{H} w^{j}(t)\|_{V}^{2} dt = \sum_{k>r} \sigma_{k}^{2} \|\varphi_{k}\|_{V}^{2},$$
(3.30)

$$\sum_{j=1}^{m} \int_{O} \|w^{j}(t) - P_{r}^{V} w^{j}(t)\|_{V}^{2} dt = \sum_{k>r} \sigma_{k}^{2} \|\varphi_{k} - P_{r}^{V} \varphi_{k}\|_{V}^{2},$$
(3.31)

$$\sum_{j=1}^{m} \int_{O} \|w^{j}(t) - P_{r}^{V} w^{j}(t)\|_{H}^{2} dt = \sum_{k>r} \sigma_{k}^{2} \|\varphi_{k} - P_{r}^{V} \varphi_{k}\|_{H}^{2}.$$
 (3.32)

In this example, all three sums converge to zero as r increases.

A larger improvement from [1] can be seen in the results concerning pointwise convergence of POD projections. To illustrate, we give the following result.

Proposition 3.5.1. We have

1. $||P_r^V y - y||_V \to 0$ for all $y \in \mathcal{R}(K)$, and for y = Kg we have

$$\|P_r^V y - y\|_V \le \sum_{k>r} \sigma_k |(g, f^k)_S| \, \|P_r^V \varphi_k - \varphi_k\|_V$$

- 2. If the POD eigenvalues for $\{w^j\} \subset L^2(O; V)$ are all nonzero, then $P_r^V y \to y$ in both *H* and *V* for all $y \in V$.
- 3. $||P_r^H y y||_V \to 0$ for all $y \in \mathcal{R}(K)$, and for y = Kg we have

$$\|y - P_r^H y\|_V \le \sum_{k>r} \sigma_k |(g, f^k)_S| \|\varphi_k\|_V.$$

4.
$$||P_r^V x - x||_H \to 0$$
 for all $x \in \mathcal{R}(K)$, and for $x = Kg$ we have

$$\|x - P_r^V x\|_H \le \sum_{k>r} \sigma_k |(g, f^k)_S| \|\varphi_k - P_r^V \varphi_k\|_H.$$

Note that since Π_r^Y is orthogonal, item 1 and item 2 follow from Theorem 3.4.5 and item 2 of Theorem 3.4.2. Items 3 and 4 can be obtained from Theorem 3.4.6, Theorem 3.4.7, and the fact that L^{-1} is bounded.

The pointwise convergence results above are more complete and more sharp than the results in [1, Proposition 5.5]. First, item 2 is shown in [1, Proposition 5.5] under the assumption that all the POD singular values for $\{w^j\} \subset L^2(O; V)$ are nonzero; as discussed in Section 2.2.2 this is a more restrictive assumption than the POD eigenvalues all being nonzero, as is required above. Next, the convergence result in item 3 is shown in [1, Proposition 5.5]; however, the error bound in item 3 is new. Also, items 1 and 4 are completely new.

For item 3, we note that an error bound was given in the proof of [1, Proposition 5.5]. However, that error bound does not converge to zero as fast as the error bound given in Theorem 3.4.6. Specifically, the error bound in [1] is a constant multiple of $(\sum_{k>r} |(g, f^k)_S|^2)^{1/2}$. However, the error bound in item 3 can be bounded above by

$$\|y - P_r^H y\|_V \le \left(\sum_{k>r} |(g, f^k)_s|^2\right)^{1/2} \left(\sum_{k>r} \sigma_k^2 \|\varphi_k\|_V^2\right)^{1/2},$$

and both terms in parentheses tend to zero as r increases by Parseval's equality and Lemma 3.3.1 (see the proof of Theorem 3.4.6). Therefore, the error bound in item 3 is an improvement over the error bound in [1].

Finally, we consider boundedness of the non-orthogonal POD projections $P_r^H : V \to V$ and $P_r^V : H \to H$. For each fixed r, it is shown in [1, Lemma 3.3] that $P_r^H : V \to V$ is bounded. Singler did not consider the boundedness of $P_r^V : H \to H$ in [1]. Below, we

use Theorem 3.2.3 to show $P_r^H: V \to V$ is bounded and also give a condition guaranteeing $P_r^V: H \to H$ has a bounded extension. However, we still do not know if these non-orthogonal POD projections are uniformly bounded in operator norm.

Define the linear operator $A : \mathcal{D}(A) \subset H \to H$ by

$$(Au, v)_H = (u, v)_V$$

for all $u \in \mathcal{D}(A)$ and $v \in V$ (see, e.g., [63, Section II.2]). We know A is closed. Now we apply this to our example. For all $x \in \mathcal{D}(L) = V$ and $y \in \mathcal{D}(L^*)$ we have

$$(x, L^*y)_H = (Lx, y)_V = (x, y)_V$$
$$\Rightarrow (L^*y, x)_H = (y, x)_V.$$

Thus, $L^* = A$ and $\mathcal{D}(L^*) = \mathcal{D}(A)$. For PDE solution data we often have $\{Aw^j\} \subset L^2(O; H)$ for each *j*; see [63] for examples. In this case, since $\varphi_k = \sigma_k^{-1} K f^k$ we can use the Bochner integral result in Theorem 2.1.2 to show $\varphi_k \in \mathcal{D}(A)$ whenever $\sigma_k > 0$.

Therefore, since L^{-1} is bounded, item 1 and item 4 of Theorem 3.2.3 give the following result.

Proposition 3.5.2. Let *r* be fixed. The operator $P_r^H : V \to V$ is bounded, and if $\{Aw^j\}_{j=1}^m \subset L^2(O; H)$, then the operator $P_r^V : H \to H$ can be extended to a bounded operator.

3.5.2.2. Example 2. Next, consider the case where X = V, Y = H, and $L : V \to H$ is defined by by Lv = v for all $v \in V$. Then $L^{-1} : H \to V$ is given by $L^{-1}v = v$ for all $v \in \mathcal{D}(L^{-1}) = V$. Note that in this case L is bounded by the continuous embedding property. Again, the assumption on the data gives $\{w^j\} \subset L^2(O; X)$ and $\{Lw^j\} \subset L^2(O; Y)$. Therefore, the main assumption and the L^{-1} assumption hold.

Since X = V, in this example the POD modes $\{\varphi_k\}$ are an orthonormal basis for V. We have $X_r = \text{span}\{\varphi_k\}_{k=1}^r \subset V$, and $Y_r = \text{span}\{L\varphi_k\}_{k=1}^r = \text{span}\{\varphi_k\}_{k=1}^r \subset H$. The POD modes $\{\varphi_k\}$ may not be orthogonal in H. The operator $K^Y = LK$ is the POD operator $K : S \to H$. As in Example 1, $\Pi_r^X : X \to X$ is the orthogonal projection onto X_r , and $\Pi_r^Y : Y \to Y$ is the orthogonal projection onto Y_r .

The projections in this work are related to the projections Q_r^H and Q_r^V from [1, Definition 3.2] as follows:

- $\Pi_r^X : X \to X$ is equal to the orthogonal projection $Q_r^V : V \to V$.
- $\Pi_r^Y: Y \to Y$ is equal to the orthogonal projection $Q_r^H: H \to H$.
- $L\Pi_r^X L^{-1}: Y \to Y$ is equal to the operator $Q_r^V: H \to H$.
- $L^{-1}\Pi_r^Y L: X \to X$ is equal to the operator $Q_r^H: V \to V$.

As before, the main data approximation error results in Theorem 3.3.4 become

$$\sum_{j=1}^{m} \int_{O} \|w^{j}(t) - Q_{r}^{V}w^{j}(t)\|_{H}^{2}dt = \sum_{k>r} \sigma_{k}^{2} \|\varphi_{k}\|_{H}^{2},$$

$$\sum_{j=1}^{m} \int_{O} \|w^{j}(t) - Q_{r}^{H}w^{j}(t)\|_{H}^{2}dt = \sum_{k>r} \sigma_{k}^{2} \|\varphi_{k} - Q_{r}^{H}\varphi_{k}\|_{H}^{2}$$

$$\sum_{j=1}^{m} \int_{O} \|w^{j}(t) - Q_{r}^{H}w^{j}(t)\|_{V}^{2}dt = \sum_{k>r} \sigma_{k}^{2} \|\varphi_{k} - Q_{r}^{H}\varphi_{k}\|_{V}^{2}$$

Here the first two sums converge to zero as r increases. However, we cannot show convergence of the last sum. This is because we do not know L^{-1} is bounded or $\{Q_r^H\}$ is uniformly bounded as a family of operators mapping V to V. As before, the only improvement here compared to [1] is that O is not restricted to be an interval.

We also have the following pointwise convergence results.

Proposition 3.5.3. As r increases we have

1. $||Q_r^H y - y||_H \rightarrow 0$ for all $y \in H$, and for y = Kg we have

$$\|Q_r^H y - y\|_H \le \sum_{k>r} \sigma_k |(g, f^k)_S| \|Q_r^H \varphi_k - \varphi_k\|_H$$

2. $||Q_r^V y - y||_H \rightarrow 0$ for all $y \in V$, and for y = Kg we have

$$\|Q_r^V y - y\|_H \le \sum_{k>r} \sigma_k |(g, f^k)_S| \|\varphi_k\|_H$$

3. For x = Kg we have

$$\|Q_r^H x - x\|_V \le \sum_{k>r} \sigma_k |(g, f^k)_S| \|Q_r^H \varphi_k - \varphi_k\|_V$$

If also $s_X < \infty$ or L^{-1} is bounded, then the error goes to zero as r increases.

Since *L* is bounded, item 1 follows from item 1 of Theorem 3.4.1 and also Theorem 3.4.5. Item 2 can be obtained from Theorem 3.4.6, using *L* is bounded. Theorem 3.4.7 gives item 3; note that we cannot guarantee convergence of the error without the extra assumptions since we only know *L* is bounded.

Again, these results improve on the results in [1, Proposition 5.5]. All of the error bounds are new. The convergence result in item 2 was not stated in [1], but it follows directly from the continuous embedding and $||Q_r^V y - y||_V \rightarrow 0$ for all $y \in V$. The convergence result in item 1 was given in [1, Proposition 5.5], however that work made the assumption that all the POD singular values for $\{w^j\} \subset L^2(O; V)$ are nonzero. Here, we proved the convergence result in item 1 without that assumption. Next, we use the technique from Section 3.5.2.1 to determine the boundedness of the non-orthogonal POD projections $Q_r^H : V \to V$ and $Q_r^V : H \to H$. For this example, we have $A = L^{-*} = (L^{-1})^*$. Therefore, if $\{Aw^j\} \subset L^2(O; H)$, then we have $\{\varphi_k\} \subset \mathcal{D}(L^{-*})$, just as in Section 3.5.2.1. Since *L* is bounded, items 2 and 3 of Theorem 3.2.3 give the following result.

Proposition 3.5.4. Let *r* be fixed. The operator $Q_r^H : V \to V$ is bounded, and if $\{Aw^j\}_{j=1}^m \subset L^2(O; H)$, then the operator $Q_r^V : H \to H$ can be extended to a bounded operator on *H*.

3.5.3. Non-orthogonal Projection Example. For the final example, we consider a case where Π_r^Y is not an orthogonal projection. In particular, we take Π_r^Y to be a Ritz projection, as considered in [24, 27]. All of our results for this case are new.

Consider the situation from Example 1 in Section 3.5.2.1: we have X = H, Y = V, and $L : X \to Y$ is defined by Lv = v for all $v \in \mathcal{D}(L) = Y$. Assume we have a continuous elliptic sesquilinear form² $a : V \times V \to \mathbb{K}$. Define the projection $P_r^V : V \to V$ onto $V_r := Y_r = \operatorname{span}\{L\varphi_k\} = \operatorname{span}\{\varphi_k\} \subset V$ as follows: let $P_r^V u := u_r \in V_r$ be the unique solution of

$$a(u_r, v_r) = a(u, v_r)$$
 for all $v_r \in V_r$.

The existence and uniqueness of such a solution is guaranteed by the Lax-Milgram Theorem. We take $\Pi_r^Y = P_r^V$.

Note that the main difference between this example and Example 1 is that the projection P_r^V is not the same. However, for this example it can be checked that the family of projections, $\{\Pi_r^Y\}$, is uniformly bounded. Therefore, the same pointwise convergence results and error formulas from Section 3.5.2.1 hold for this example with $P_r^V : V \to V$ defined as above. We note that these pointwise convergence results and error formulas are all new. Bounds on the POD data approximation errors can be found in Lemma 3.4 in [24]

²i.e., there exists constants C_a , $c_a > 0$ such that $|a(u, v)| \le C_a ||u||_V ||v||_V$ and $c_a ||u||_V^2 \le \operatorname{Re} a(u, u)$ for all $u, v \in V$

and Lemma 2.9 in [27] in the discrete case; however, we have the exact formulas (3.30)-(3.32) for the POD data approximation errors in the continuous case. Again, analogous error formulas can be derived for the discrete case using our results.

4. A NEW APPROACH TO POD WITH DIFFERENCE QUOTIENTS

4.1. REVIEW OF EXISTING METHODS

Before presenting the new approach to proper orthogonal decomposition with difference quotients, we present current approaches to POD: the standard POD approach and standard POD with difference quotients approach. We compare our new method to known results about these established methods throughout the section. For details on the basics of POD see, e.g., [14, 15, 16, 19, 49, 50, 51].

First we establish some general notation. Let M and N be a positive integers and recall that X and Y are Hilbert spaces where the space X is called the POD space. In this chapter it is possible for Y = X. In the examples in this chapter for these Hilbert spaces we use the standard function spaces $L^2(\Omega)$ and $H_0^1(\Omega)$, where Ω is the spatial domain. In order to consider variable weights that often arise with numerical integration, we recall the definition from Section 2.2.1 of $S := \mathbb{K}_{\Gamma}^M$ with the weighted inner product given by

$$(g,h)_S = h^* \Gamma g = \sum_{j=1}^M \gamma_j g^j \overline{h^j}$$

where $g, h \in S$, $\Gamma = \text{diag}(\gamma_1, \gamma_2, \dots, \gamma_M)$, and the values $\{\gamma_j\}_{j=1}^M$ are positive weights. In some instances it is beneficial to take the positive weights to be certain specific values in order to approximate various time integrals.

For the POD reduced order modeling in this chapter, we consider data sets consisting of approximate solution data for a time dependent partial differential equation. Throughout, we consider the time interval [0,T] with T > 0 a fixed positive constant. The approximate solution data will be given at times $t_n = (n-1)\Delta t$, for n = 1, ..., N, where the time step is given by $\Delta t = \frac{T}{N-1}$. Note that while *T* is fixed, *N* can vary. **4.1.1. Standard POD.** First we recall the standard POD approach as presented in Section 2.2.1. We present it here again with slightly different notation in order to use it for the remainder of the work.

Let $W = \{w^j\}_{j=1}^M \subset X$ be the POD data for some integer M > 0. We have the same POD operator as in Equation (2.2) which is

$$Kf = \sum_{j=1}^{M} \gamma_j f^j w^j, \quad f = [f^1, f^2, \dots, f^M]^T.$$
(4.1)

We call this operator the standard POD operator. Further recall that the POD modes provide the best low rank approximation to the data. Thus we have the following formula as in Equation (2.5):

$$E_r = \sum_{j=1}^M \gamma_j \|w^j - \Pi_r^X w^j\|_X^2 = \sum_{k=r+1}^{s_X} \lambda_k$$
(4.2)

where $\{\lambda_k\}$ are the POD eigenvalues and s_X is the number of positive POD singular values.

For certain choices of weights $\{\gamma_i\}$ the error given in Equation (2.5) approximates a time integral, or a constant multiple of a time integral, as more and more time steps are used. Using various quadrature rules to determine the appropriate POD weights will lead to different time integral approximations. Allowing the weights to vary for the standard POD problem will also allow us to apply known results for this approach to new approaches.

The following lemma provides exact formulas for POD data approximation errors using other norms and other projections. We use this result later to provide POD data approximation results for other POD approaches. The proof of Lemma 4.1.1 is very similar to the proof of Theorem 3.3.2 above so we omit it here. For another similar result and proof see [37, Lemma 2.2].

Lemma 4.1.1. Let $X_r = \text{span}\{\varphi_k\}_{k=1}^r$ and $\Pi_r^X : X \to X$ be the orthogonal projection onto X_r . Let s_X be the number of positive POD singular values for K defined in Equation (2.2). If Y is a Hilbert space with $W \subset Y$ then

$$\sum_{j=1}^{M} \gamma_j \|w^j - \Pi_r^X w^j\|_Y^2 = \sum_{i=r+1}^{s_X} \lambda_i \|\varphi_i\|_Y^2.$$
(4.3)

In addition if $\pi_r : Y \to Y$ is a bounded linear projection onto X_r then,

$$\sum_{j=1}^{M} \gamma_j \|w^j - \pi_r w^j\|_Y^2 = \sum_{i=r+1}^{s_X} \lambda_i \|\varphi_i - \pi_r \varphi_i\|_Y^2.$$
(4.4)

The standard POD approach does not have general bounds for pointwise errors, as shown in [37, Section 3].

4.1.2. POD with Difference Quotients. Another common approach to POD involves the use of difference quotients. Throughout this chapter, we refer to this method as the standard DQ approach. This approach has been studied by many including [18, 37, 38, 40, 64, 65]. We consider backward Euler for the time stepping scheme and the difference quotients.

Let $U = \{u^j\}_{j=1}^N \subset X$ be a given data set. Then the problem is to find an orthonormal basis minimizing the error

$$E_r^{DQ} = \sum_{j=1}^N \Delta t \| u^j - \Pi_r^X u^j \|_X^2 + \sum_{j=1}^{N-1} \Delta t \| \partial u^j - \Pi_r^X \partial u^j \|_X^2$$
(4.5)

where Δt is the time step and the difference quotients are given by

$$\partial u^j = \frac{u^{j+1} - u^j}{\Delta t}.\tag{4.6}$$

These difference quotients approximate the time derivative of the data in continuous time. The operator that provides the minimizing basis for the error in Equation (4.5) is K_{DQ} : $S \rightarrow X$ defined by

$$K_{DQ}f = \sum_{j=1}^{N} \Delta t f^{j} u^{j} + \sum_{j=1}^{N-1} \Delta t f^{N+j} \partial u^{j}.$$
(4.7)

This approach uses a total of M = 2N - 1 data snapshots which is nearly twice as many as the standard POD approach. Note for this operator we have $K_{DQ}f = Kf$ where $w^i = u^i$ and $\gamma_i = \Delta t$ for i = 1, ..., N, and also $w^{N+i} = \partial u^i$ and $\gamma_{N+i} = \Delta t$ for i = 1, ..., N-1. The resulting POD data set is $\{w^j\}_{j=1}^M$, where M = 2N - 1. Taking $\{\lambda_j^{DQ}\}_{j=1}^{2N-1}$ to be the POD eigenvalues and keeping the same notation $\{\varphi_j\}$ for the POD basis functions, we get similar results to those for the standard POD operator. When using this new set $\{w^j\}$ as the POD data set, we not only have a set that is nearly twice as large as the original but it can also be checked that the new set is linearly dependent. This redundancy is something we avoid with our new approach introduced in Section 4.2.

Remark 4.1.2. While the weights can be taken to be any set of positive constants, for simplicity we take them to be the constant Δt . The results in this chapter can be extended to variable weights or other choices of constant weights. One popular choice of constant weight in the literature is M^{-1} as in [37, 38] where M represents the total number of data snapshots for the standard difference quotient approach to POD. Similarly to the standard POD case, with certain choices of weights, one can approximate time integrals with various quadrature rules.

The following result is also similar to Lemma 2.4 in [37]. We provide it here for completeness.

Lemma 4.1.3. Let $X_r = \operatorname{span}\{\varphi_k\}_{k=1}^r$ and $\Pi_r^X : X \to X$ be the orthogonal projection onto X_r . Let s_X be the number of positive POD singular values for K_{DO} defined in Equation (4.7). We have the following error formula:

$$\sum_{j=1}^{N} \Delta t \| u^{j} - \Pi_{r}^{X} u^{j} \|_{X}^{2} + \sum_{j=1}^{N-1} \Delta t \| \partial u^{j} - \Pi_{r}^{X} \partial u^{j} \|_{X}^{2} = \sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ}.$$
(4.8)

If Y is a Hilbert space with $U \subset Y$ then

$$\sum_{j=1}^{N} \Delta t \| u^{j} - \Pi_{r}^{X} u^{j} \|_{Y}^{2} + \sum_{j=1}^{N-1} \Delta t \| \partial u^{j} - \Pi_{r}^{X} \partial u^{j} \|_{Y}^{2} = \sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ} \| \varphi_{i} \|_{Y}^{2}.$$
(4.9)

In addition, if $\pi_r : Y \to Y$ is a bounded linear projection onto X_r then

$$\sum_{j=1}^{N} \Delta t \| u^{j} - \pi_{r} u^{j} \|_{Y}^{2} + \sum_{j=1}^{N-1} \Delta t \| \partial u^{j} - \pi_{r} \partial u^{j} \|_{Y}^{2} = \sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ} \| \varphi_{i} - \pi_{r} \varphi_{i} \|_{Y}^{2}.$$
(4.10)

Proof. This result follows from Equation (2.5) and Lemma 4.1.1 by taking $\{\lambda_i^{DQ}\}_{i=1}^{s_X}$ as the POD eigenvalues for the POD operator in Equation (4.7).

We have the following result about the pointwise error bounds for this POD case. A similar result with a different set of constant weights can be found in [37, Theorem 3.7]. The proof of the result below is similar so we omit it here. That theorem was key to obtaining the optimal pointwise POD ROM error bounds which were a main contribution of that work. **Theorem 4.1.4.** Let $X_r = \text{span}\{\varphi_k\}_{k=1}^r$ and $\Pi_r^X : X \to X$ be the orthogonal projection onto X

$$X_r$$
. Let s_X be the number of positive POD singular values for K_{DQ} . We have

$$\max_{1 \le j \le N} \| u^j - \Pi_r^X u^j \|_X^2 \le C \left(\sum_{i=r+1}^{s_X} \lambda_i^{DQ} \right).$$
(4.11)

If Y is a Hilbert space with $U \subset Y$ then

$$\max_{1 \le j \le N} \| u^j - \Pi_r^X u^j \|_Y^2 \le C \left(\sum_{i=r+1}^{s_X} \lambda_i^{DQ} \| \varphi_i \|_Y^2 \right).$$
(4.12)

If $\pi_r: Y \to Y$ is a bounded linear projection onto X_r then

$$\max_{1 \le j \le N} \|u^{j} - \pi_{r} u^{j}\|_{Y}^{2} \le C \left(\sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ} \|\varphi_{i} - \pi_{r} \varphi_{i}\|_{Y}^{2} \right)$$
(4.13)

where $C = 2 \max\{T^{-1}, T\}$.

In Section 4.2.2 we obtain a similar pointwise POD projection error result for the new POD approach described below in Section 4.2.

4.2. A NEW APPROACH

Next, we return to the question posed in the introduction: "Can we obtain all of the same numerical analysis benefits of using DQs with POD using a data set without redundancy?" We obtain a positive answer to this question by introducing a new POD problem and operator. Instead of including all of the POD data snapshots and all of the difference quotients as in Section 4.1.2, we include the first data snapshot and all of the difference quotients. Thus for the data $U = \{u^j\}_{j=1}^N \subset X$, the new POD problem is to minimize the error given by

$$E_r^{DQ1} = \|u^1 - \Pi_r^X u^1\|_X^2 + \sum_{j=1}^{N-1} \Delta t \|\partial u^j - \Pi_r^X \partial u^j\|_X^2$$
(4.14)

where the difference quotients are defined by Equation (4.6). Note that the POD error function in Equation (4.14) does not include the weighted sum of the errors of the regular snapshots; this contrasts with the POD approaches in Section 4.1, which both include such error terms, see Equation (2.5) and Equation (4.5). Furthermore, in the POD approaches

in Section 4.1, we have exact error formulas for these error terms; see Lemma 4.1.1 and Lemma 4.1.3. In Section 4.2.2, we consider the weighted sum of the errors of the regular snapshots for this new approach and obtain an approximation error result in Corollary 4.2.5.

Note that with this approach we have a total of N - 1 difference quotients. Together with the single snapshot, we have a total of N data snapshots for this POD problem. This is an improvement from the standard DQ approach to POD which has 2N - 1 data snapshots.

The minimum error can be found using the POD operator:

$$K_1 f = f^1 u^1 + \sum_{j=1}^{N-1} \Delta t f^{j+1} \partial u^j$$
(4.15)

We have $K_1 f = K f$ where $w^1 = u^1$, $\gamma_1 = 1$, $\gamma_{i+1} = \Delta t$, and $w^{i+1} = \partial u^i$ for i = 1, ..., N - 1. We choose to use the constant time step, Δt , as the weight for simplicity throughout. The results can be extended to include variable weights as well. Recall that for certain choices of weights one can approximate a time integral and the difference quotients approximate a time derivative.

Note that in contrast to the data set created for the standard POD approach with DQs, the data set for this new approach is linearly independent if the original data set is linearly independent as shown in Lemma 4.2.1.

Lemma 4.2.1. If $\{u^i\}_{i=1}^N$ is linearly independent then $\{w^i\}_{i=1}^N$ given by $w^1 = u^1$ and $w^{i+1} = \partial u^i$ for i = 1, ..., N - 1 is linearly independent.

Proof. We show that if

$$c_1 w^1 + c_2 w^2 + \dots + c_N w^N = 0$$

then $c_i = 0$ for all i = 1, ..., N. We have

$$c_1 u^1 + c_2 \left(\frac{u^2 - u^1}{\Delta t}\right) + \dots + c_i \left(\frac{u^i - u^{i-1}}{\Delta t}\right) + \dots + c_N \left(\frac{u^N - u^{N-1}}{\Delta t}\right) = 0$$

then

$$\left(c_1 - \frac{c_2}{\Delta t}\right)u^1 + \left(\frac{c_2}{\Delta t} - \frac{c_3}{\Delta t}\right)u^2 + \dots + \left(\frac{c_{N-1}}{\Delta t} - \frac{c_N}{\Delta t}\right)u^{N-1} + \frac{c_N}{\Delta t}u^N = 0.$$

Since $\{u_i\}$ is linearly independent we know each of these coefficients must equal 0. Solving that system of equations leads to the conclusion that $c_i = 0$ for all i = 1, ..., N.

If we let $\{\lambda_j^{DQ1}\}_{j=1}^N$ be the POD eigenvalues for this new POD approach, and let $\{\varphi_k\}_{k=1}^r$ be the POD modes for this data, we get the following error formulas given in Lemma 4.2.2.

Lemma 4.2.2. Let $X_r = \text{span}\{\varphi_k\}_{k=1}^r$ and $\Pi_r^X : X \to X$ be the orthogonal projection onto X_r . Let s_X be the number of positive POD singular values for K_1 defined in Equation (4.15). We have the following formula for the data approximation error:

$$\|u^{1} - \Pi_{r}^{X}u^{1}\|_{X}^{2} + \sum_{j=1}^{N-1} \Delta t \|\partial u^{j} - \Pi_{r}^{X}\partial u^{j}\|_{X}^{2} = \sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1}.$$
(4.16)

If Y is a Hilbert space with $U \subset Y$ then

$$\|u^{1} - \Pi_{r}^{X}u^{1}\|_{Y}^{2} + \sum_{j=1}^{N-1} \Delta t \|\partial u^{j} - \Pi_{r}^{X}\partial u^{j}\|_{Y}^{2} = \sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1} \|\varphi_{i}\|_{Y}^{2}.$$
 (4.17)

If in addition $\pi_r: Y \to Y$ is a bounded linear projection onto X_r then,

$$\|u^{1} - \pi_{r}u^{1}\|_{Y}^{2} + \sum_{j=1}^{N-1} \Delta t \|\partial u^{j} - \pi_{r}\partial u^{j}\|_{Y}^{2} = \sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1} \|\varphi_{i} - \pi_{r}\varphi_{i}\|_{Y}^{2}.$$
(4.18)

Proof. This result follows from Equation (2.5) and Lemma 4.1.1 by taking $\{\lambda_j^{DQ1}\}_{j=1}^{s_X}$ as the POD eigenvalues for the POD operator in Equation (4.15).

Preliminary computational results and pointwise error estimates for this new POD approach are discussed in Section 4.2.1 and Section 4.2.2 respectively.
4.2.1. Preliminary Computations. Before moving to our main results we perform some preliminary computations to test the new POD approach with difference quotients. For all computations in this chapter we consider the following test problem.

Test Problem: Consider the one dimensional heat equation

$$u_t - v u_{xx} = 0, \text{ in } \Omega \times [0, T]$$
$$u(x, 0) = e^x \sin(\pi x)$$

with $\nu = 1$, $\Omega = [0, 1]$, T = 1, and zero Dirichlet boundary conditions.

For the data $\{u^j\}$, we compute the solution using the finite element method with linear elements, equally spaced nodes, and backward Euler with a constant time step for the time stepping. The initial condition is taken to be the linear interpolation of the initial condition with respect to the finite element nodes. For this data and with the POD space $X = L^2(\Omega)$, we can calculate the POD modes, POD singular values, and the data approximation errors exactly which allows for comparison between the errors in the formulas from Lemma 4.2.2 and the actual approximation errors. In order to compute the singular value decomposition (SVD) of the POD operator, we use the technique described in [66, Section 2.2] with a minor modification to account for the POD weights. This procedure works well and is highly accurate for smaller data sets; for larger data sets one could use an incremental SVD approach or another related algorithm instead, see e.g. [66, 67, 68, 69, 70] and the references therein.

First we plot the singular values for both the standard POD and our new POD approach that includes one data snapshot and all of the difference quotients. For the standard POD computations we use the data set consisting of only the regular snapshots $\{w^j\}_{j=1}^N = \{u^j\}_{j=1}^N$ and we choose constant weights $\gamma_j = \Delta t$ for j = 1, ..., N. The singular value plots allow for a quick comparison between the two approaches to POD and are given in Figure 4.1. For each plot, we show the first 20 POD singular values for 20, 50, 100, and



Figure 4.1. Plot of singular values of POD operators using different numbers of finite element nodes

150 finite element nodes when using 100 equally spaced time steps. The POD singular values decay at a similar rate which indicates that the POD basis for each case has a similar ability to approximate the data.

Next, we consider the data approximation error results given in Lemma 4.2.2 numerically. For these computations we take 200 equally spaced time steps and 100 equally spaced finite element nodes and compute the actual errors and the error formulas. Recall that $X = L^2(\Omega)$ is the POD space and note that here we take either $Y = H_0^1(\Omega)$ or $Y = L^2(\Omega)$. Also note the projection π_r is taken to be the Ritz projection, which we discuss in more detail in Section 4.3. The results are shown for all formulas of Lemma 4.2.2 in Table 4.1 for r = 4, r = 6, and r = 8. For example, the second row for each r value in the table gives the values for

actual error =
$$||u^1 - \prod_r^X u^1||_{H_0^1}^2 + \sum_{j=1}^{N-1} \Delta t ||\partial u^j - \prod_r^X \partial u^j||_{H_0^1}^2$$
,

and

error formula =
$$\sum_{i=r+1}^{s_X} \lambda_i^{DQ1} \|\varphi_k\|_{H_0^1}^2$$

<i>r</i> value	Projection	Norm	Actual Error	Error Formula
4	Π_r^X	$L^2(\Omega)$	9.761e-06	9.761e-06
	Π_r^X	$H_0^1(\Omega)$	6.200e-03	6.200e-03
	π_r	$L^2(\Omega)$	1.736e-05	1.736e-05
	π_r	$H^1_0(\Omega)$	3.100e-03	3.100e-03
6	Π_r^X	$L^2(\Omega)$	5.482e-09	5.482e-09
	Π_r^X	$H^1_0(\Omega)$	1.036e-05	1.036e-05
	π_r	$L^2(\Omega)$	1.133e-08	1.133e-08
	π_r	$H^1_0(\Omega)$	5.159e-06	5.159e-06
8	Π_r^X	$L^2(\Omega)$	1.557e-12	1.557e-12
	Π_r^X	$H^1_0(\Omega)$	4.772e-09	4.772e-09
	π_r	$L^2(\Omega)$	3.481e-12	3.481e-12
	π_r	$H_0^1(\Omega)$	1.659e-09	1.659e-09

Table 4.1. Actual Error vs. Error Formulas from Lemma 4.2.2 for New DQ POD

with the respective values for *r*. Round-off errors in the POD computations can cause very small imaginary parts to occur in the error formulas. Thus we report the absolute value of the computed error formulas. Note that the difference between the actual errors and the error formulas is unnoticeable to the given significant digits. These computational results verify what we show analytically. Similar results are achieved when using $X = H_0^1(\Omega)$.

Computational comparisons between all three of the methods can be found in Section 4.4.1 where we consider the errors in the reduced order models.

4.2.2. Pointwise Error Bounds. Using the technique from [37, Lemma 3.6] we establish the following lemma which allows us to directly prove the POD pointwise projection error bounds for the new DQ POD approach and an approximation error result for the weighted sum of the errors of the regular snapshots. These results are necessary to prove the reduced order model error bounds and show their optimality in Section 4.3.

Lemma 4.2.3. Let T > 0, Z be a normed space, $\{z^j\}_{j=1}^N \subset Z$, and $\Delta t = T/(N-1)$. Then

$$\max_{1 \le j \le N} \|z^j\|_Z^2 \le C \left(\|z^1\|_Z^2 + \sum_{\ell=1}^{N-1} \Delta t \|\partial z^\ell\|_Z^2 \right)$$
(4.19)

where $\partial z^{\ell} = \frac{z^{\ell+1}-z^{\ell}}{\Delta t}$ for $\ell = 1, ..., N-1$, and $C = 2 \max\{T, 1\}$.

Proof. Note using $z^j = z^1 + \sum_{\ell=1}^{j-1} \Delta t(\partial z^\ell)$, we have

$$\|z^{j}\| \leq \|z^{1}\| + \sum_{\ell=1}^{j-1} \Delta t \|\partial z^{\ell}\| \leq \|z^{1}\| + \left(\sum_{\ell=1}^{j-1} \Delta t\right)^{1/2} \left(\sum_{\ell=1}^{N-1} \Delta t \|\partial z^{\ell}\|^{2}\right)^{1/2}.$$

Then

$$\|z^{j}\|^{2} \leq 2\|z^{1}\|^{2} + 2\left(\sum_{\ell=1}^{j-1} \Delta t\right) \left(\sum_{\ell=1}^{N-1} \Delta t \|\partial z^{\ell}\|^{2}\right) \leq 2\|z^{1}\|^{2} + 2T\sum_{\ell=1}^{N-1} \Delta t \|\partial z^{\ell}\|^{2}$$

since $T = \sum_{j=1}^{N-1} \Delta t = (N-1)\Delta t$. Take the maximum over all *j* and the result follows. \Box

Next, we obtain a pointwise POD projection error result for the new POD DQ approach that is very similar to Theorem 4.1.4 for the standard POD DQ case.

Theorem 4.2.4. Let $X_r = \text{span}\{\varphi_k\}_{k=1}^r$ and $\Pi_r^X : X \to X$ be the orthogonal projection onto X_r . Let s_X be the number of positive POD eigenvalues for $U = \{u^j\}_{j=1}^N$. Then

$$\max_{1 \le j \le N} \| u^j - \Pi_r^X u^j \|_X^2 \le C \left(\sum_{i=r+1}^{s_X} \lambda_i^{DQ1} \right).$$
(4.20)

If Y is a Hilbert space with $U \subset Y$ then

$$\max_{1 \le j \le N} \| u^j - \Pi_r^X u^j \|_Y^2 \le C \left(\sum_{i=r+1}^{s_X} \lambda_i^{DQ1} \| \varphi_i \|_Y^2 \right),$$
(4.21)

and in addition if $\pi_r : Y \to Y$ is a bounded linear projection onto X_r then

$$\max_{1 \le j \le N} \|u^{j} - \pi_{r} u^{j}\|_{Y}^{2} \le C \left(\sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1} \|\varphi_{i} - \pi_{r} \varphi_{i}\|_{Y}^{2} \right)$$
(4.22)

where $C = 2 \max\{T, 1\}$.

Proof. To prove this theorem take $z^j = u^j - \prod_r^X u^j$ or $z^j = u^j - \pi_r u^j$ with Z = X or Z = Y in Lemma 4.2.3. For example, if we let $z^j = u^j - \prod_r^X u^j$ and Z = X, then

$$\max_{1 \le j \le N} \|u^j - \Pi_r^X u^j\|_X^2 \le C \left(\|u^1 - \Pi_r^X u^1\|_X^2 + \sum_{\ell=1}^{N-1} \Delta t \|\partial u^\ell - \Pi_r^X \partial u^\ell\|_X^2 \right)$$
(4.23)

Applying Lemma 4.2.2 for each of the three cases gives the result.

Next, we use the above pointwise error bounds to obtain error bounds for the weighted sum of the errors of the regular snapshots.

Corollary 4.2.5. Let $X_r = \text{span}\{\varphi_k\}_{k=1}^r$ and $\Pi_r^X : X \to X$ be the orthogonal projection onto X_r . Let s_X be the number of positive POD eigenvalues for U then

$$\sum_{j=1}^{N} \Delta t \| u^{j} - \Pi_{r}^{X} u^{j} \|_{X}^{2} \le C \left(\sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1} \right).$$
(4.24)

If Y is a Hilbert space with $U \subset Y$ then

$$\sum_{j=1}^{N} \Delta t \| u^{j} - \Pi_{r}^{X} u^{j} \|_{Y}^{2} \le C \left(\sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1} \| \varphi_{i} \|_{Y}^{2} \right).$$
(4.25)

If in addition $\pi_r: Y \to Y$ is a bounded linear projection onto X_r then

$$\sum_{j=1}^{N} \Delta t \| u^{j} - \pi_{r} u^{j} \|_{Y}^{2} \le C \left(\sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1} \| \varphi_{i} - \pi_{r} \varphi_{i} \|_{Y}^{2} \right)$$
(4.26)

where $C = 4 \max\{T^2, T\}$.

Proof. Since

$$N\Delta t = \frac{N}{N-1} [(N-1)\Delta t] = [N/(N-1)]T \le 2T,$$

we have

$$\sum_{j=1}^{N} \Delta t \| u^{j} - \Pi_{r}^{X} u^{j} \|_{X}^{2} \leq 2T \max_{1 \leq j \leq N} \| u^{j} - \Pi_{r}^{X} u^{j} \|_{X}^{2}.$$

Theorem 4.2.4 gives Equation (4.24). The proofs of Equations (4.25) and (4.26) follow in the same way. $\hfill \Box$

These results are similar to those for the standard DQ approach while keeping redundancy out of the data set.

4.3. REDUCED ORDER MODELING

In this section we establish theory to compare the reduced order model solution to the backward Euler finite element solution for the heat equation using our new POD with DQs approach. In this section all POD computations are done using the new approach and all function spaces are assumed to be real. Our analysis and proof techniques strongly rely on the approach in [37, Section 4]. We provide proofs to make the work self-contained.

Let $\Omega \subset \mathbb{R}^d$ with $d \ge 1$ be an open bounded domain with Lipschitz continuous boundary, and define $V = H_0^1(\Omega)$. The space V is a Hilbert space with inner product $(g,h)_{H_0^1} = (\nabla g, \nabla h)_{L^2}$. We consider the weak formulation of the heat equation with homogeneous Dirichlet boundary conditions:

$$(\partial_t u, v)_{L^2} + \nu (\nabla u, \nabla v)_{L^2} = (f, v)_{L^2} \quad \forall v \in V$$

$$(4.27)$$

where $u(\cdot, 0) = u^1$ is the initial condition, v is a positive constant, and f is a given forcing function. We project Equation (4.27) onto a standard conforming finite element space $V^h \subset V$ and apply backward Euler to obtain

$$\left(\frac{u^{n+1}-u^n}{\Delta t},v\right)_{L^2} + \nu(\nabla u^{n+1},\nabla v)_{L^2} = (f^{n+1},v)_{L^2} \quad \forall v \in V^h.$$
(4.28)

We use the data set $\{u^n\}_{n=1}^N \subset V^h$ to compute the POD modes $\{\varphi_j\}_{j=1}^r \subset V^h$ using the new DQ approach with respect to the Hilbert space X. Below we take X to be either $X = L^2(\Omega)$ or $X = H_0^1(\Omega)$. Let $V_r^h = \operatorname{span}\{\varphi_j\}_{j=1}^r$. Next, we develop the POD reduced order model of the heat equation by substituting u_r for the unknown u, using the Galerkin method, and projecting Equation (4.28) onto the space $V_r^h \subset V^h$. Thus we arrive at the following BE-POD-ROM:

$$\left(\frac{u_r^{n+1} - u_r^n}{\Delta t}, v_r\right)_{L^2} + \nu (\nabla u_r^{n+1}, \nabla v_r)_{L^2} = (f^{n+1}, v_r)_{L^2} \quad \forall v_r \in V_r^h.$$
(4.29)

We split the error, in the standard way, with

$$e^{n+1} = u^{n+1} - u_r^{n+1} = (u^{n+1} - \pi_r u^{n+1}) - (u_r^{n+1} - \pi_r u^{n+1}) = \eta^{n+1} - \phi_r^{n+1}$$

where π_r is a projection onto V_r^h , $\eta^{n+1} = u^{n+1} - \pi_r u^{n+1}$ is the POD projection error, and $\phi_r^{n+1} = u_r^{n+1} - \pi_r u^{n+1}$ is the discretization error. We subtract Equation (4.29) from Equation (4.28) and make the error substitution given above to get

$$\left(\frac{\phi_r^{n+1} - \phi_r^n}{\Delta t}, v_r\right)_{L^2} + \nu (\nabla \phi_r^{n+1}, \nabla v_r)_{L^2} = \left(\frac{\eta^{n+1} - \eta^n}{\Delta t}, v_r\right)_{L^2} + \nu (\nabla \eta^{n+1}, \nabla v_r)_{L^2} \quad \forall v_r \in V_r^h.$$
(4.30)

Remark 4.3.1. The approach taken here is different than the one taken in [37]. In that work the authors compare the ROM solution u_r^n to the exact solution of the PDE $u(t_n)$. We can bound the error between the ROM solution and the exact solution using the triangle

inequality

$$||u_r^n - u(t_n)||_Y \le ||u_r^n - u^n||_Y + ||u^n - u(t_n)||_Y$$

where *Y* is a Hilbert space. The current work focuses on bounding the ROM error term $||u_r^n - u^n||_Y$ with $Y = L^2(\Omega)$ or $Y = H_0^1(\Omega)$ and leaves the second term to be studied using well-known finite element theory.

For analysis and computations the initial condition is taken to be the POD projection of the given initial condition, i.e, $u_r^1 = \prod_r^X u^1$. Other initial conditions are possible and have been considered in other works for the standard POD and DQ POD approaches. For example, in [37], the Ritz projection was used for the initial condition. We now consider each POD space, $L^2(\Omega)$ and $H_0^1(\Omega)$, separately.

4.3.1. POD Space: $L^2(\Omega)$. In this section, we take *X* to be the space $L^2(\Omega)$. The orthogonal projection onto V_r^h , $\Pi_r^X : L^2(\Omega) \to L^2(\Omega)$, is given by

$$\Pi_r^X u = \sum_{i=1}^r (u, \varphi_i)_{L^2} \varphi_i \tag{4.31}$$

and the set of POD modes $\{\varphi_i\}$ are orthogonal in $L^2(\Omega)$. Define π_r to be the Ritz projection R_r which satisfies

$$(\nabla(w - R_r w), \nabla v_r)_{L^2} = 0 \tag{4.32}$$

for all $v_r \in V_r^h$. Thus, for all $w \in V^h$ we have

$$\|w - R_r w\|_{H^1_0} = \inf_{v_r \in V^h_r} \|w - v_r\|_{H^1_0}.$$

Let $\eta_{Ritz}^{n+1} = u^{n+1} - R_r u^{n+1}$. Then Equation (4.30) becomes

$$\left(\frac{\phi_r^{n+1} - \phi_r^n}{\Delta t}, v_r\right)_{L^2} + \nu (\nabla \phi_r^{n+1}, \nabla v_r)_{L^2} = \left(\frac{\eta_{Ritz}^{n+1} - \eta_{Ritz}^n}{\Delta t}, v_r\right)_{L^2} \quad \forall v_r \in V_r^h.$$
(4.33)

Using this ROM error equation, we obtain the error bounds given below. Note, the constant C can change from step to step, but it does not depend on any discretization parameter. In Section 4.4.3 we investigate the final value of C computationally.

Theorem 4.3.2. The pointwise L^2 solution error when the $L^2(\Omega)$ POD basis is used for the *BE-POD-ROM* is bounded by

$$\max_{k} \|e^{k}\|_{L^{2}}^{2} \leq C \left(\sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1} \|\varphi_{i} - R_{r}\varphi_{i}\|_{L^{2}}^{2} + \|\phi_{r}^{1}\|_{L^{2}}^{2} \right).$$
(4.34)

Proof. Taking $v_r = \phi_r^{n+1}$ in Equation (4.33) yields

$$\left(\frac{\phi_r^{n+1} - \phi_r^n}{\Delta t}, \phi_r^{n+1}\right)_{L^2} + \nu \|\nabla \phi_r^{n+1}\|_{L^2}^2 = \left(\frac{\eta_{Ritz}^{n+1} - \eta_{Ritz}^n}{\Delta t}, \phi_r^{n+1}\right)_{L^2}.$$
(4.35)

Now, apply Cauchy-Schwartz followed by Young's inequality with a constant δ to get

$$\left(\frac{\phi_r^{n+1} - \phi_r^n}{\Delta t}, \phi_r^{n+1}\right)_{L^2} + \nu \|\nabla \phi_r^{n+1}\|_{L^2}^2 \le \frac{1}{4\delta} \|\partial \eta_{Ritz}^n\|_{L^2}^2 + \delta \|\phi_r^{n+1}\|_{L^2}^2.$$
(4.36)

Finally apply a polarization identity, use the fact that $C \|\phi_r^{n+1}\|_{L^2}^2 \le \|\nabla \phi_r^{n+1}\|_{L^2}^2$, and rearrange to obtain

$$\left(\frac{1}{2\Delta t} - \delta + \frac{\nu}{C}\right) \|\phi_r^{n+1}\|_{L^2}^2 - \frac{1}{2\Delta t} \|\phi_r^n\|_{L^2}^2 \le \frac{1}{4\delta} \|\partial\eta_{Ritz}^n\|_{L^2}^2.$$
(4.37)

Taking $\delta = \frac{v}{C}$ and multiplying by $2\Delta t$ we get

$$\|\phi_r^{n+1}\|_{L^2}^2 - \|\phi_r^n\|_{L^2}^2 \le \frac{C\Delta t}{2\nu} \|\partial\eta_{Ritz}^n\|_{L^2}^2.$$
(4.38)

Summing from n = 1 to n = k - 1, using $||e^n||_{L^2}^2 \le 2(||\eta^n||_{L^2}^2 + ||\phi_r^n||_{L^2}^2)$, and rearranging once again gives

$$\|e^{k}\|_{L^{2}}^{2} \leq C\left(\Delta t \sum_{n=1}^{k-1} \|\partial \eta_{Ritz}^{n}\|_{L^{2}}^{2} + \|\eta_{Ritz}^{k}\|_{L^{2}}^{2} + \|\phi_{r}^{1}\|_{L^{2}}^{2}\right).$$
(4.39)

Then apply Lemma 4.2.2 and Theorem 4.2.4 with $Y = L^2(\Omega)$ and $\pi_r = R_r$ to get

$$\|e^{k}\|_{L^{2}}^{2} \leq C\left(\sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1} \|\varphi_{i} - R_{r}\varphi_{i}\|_{L^{2}}^{2} + \|\phi_{r}^{1}\|_{L^{2}}^{2}\right).$$
(4.40)

Take the maximum over all k to get the result.

Theorem 4.3.3. The pointwise H_0^1 solution error when the $L^2(\Omega)$ POD basis is used for the BE-POD-ROM is bounded by

$$\max_{k} \|\nabla e^{k}\|_{L^{2}}^{2} \leq C \left(\sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1} \left(\|\varphi_{i} - R_{r}\varphi_{i}\|_{L^{2}}^{2} + \|\nabla(\varphi_{i} - R_{r}\varphi_{i})\|_{L^{2}}^{2} \right) + \|\phi_{r}^{1}\|_{H^{1}_{0}}^{2} \right).$$
(4.41)

Proof. Taking $v_r = \partial \phi_r^n$ in Equation (4.33) gives

$$\left(\frac{\phi_r^{n+1} - \phi_r^n}{\Delta t}, \partial \phi_r^n\right)_{L^2} + \nu \left(\nabla \phi_r^{n+1}, \nabla \partial \phi_r^n\right)_{L^2} = \left(\frac{\eta_{Ritz}^{n+1} - \eta_{Ritz}^n}{\Delta t}, \partial \phi_r^n\right)_{L^2}.$$
(4.42)

Rearranging and using the definition of $\partial \phi_r^n$ yields

$$\nu \left(\nabla \phi_r^{n+1}, \nabla \partial \phi_r^n\right)_{L^2} = \left(\frac{\eta_{Ritz}^{n+1} - \eta_{Ritz}^n}{\Delta t}, \partial \phi_r^n\right)_{L^2} - \|\partial \phi_r^n\|_{L^2}^2.$$
(4.43)

Then applying first the Cauchy-Schwartz inequality followed by Young's inequality with the constant δ we obtain

$$\begin{split} \nu \left(\nabla \phi_r^{n+1}, \nabla \partial \phi_r^n \right)_{L^2} &\leq \left(\| \partial \eta_{Ritz}^n \|_{L^2} \| \partial \phi_r^{n+1} \|_{L^2} \right) - \| \partial \phi_r^n \|_{L^2}^2 \\ &\leq \left(\frac{1}{4\delta} \| \partial \eta_{Ritz}^n \|_{L^2}^2 + \delta \| \partial \phi_r^n \|_{L^2}^2 \right) - \| \partial \phi_r^n \|_{L^2}^2 \\ &= \frac{1}{4\delta} \| \partial \eta_{Ritz}^n \|_{L^2}^2 + (\delta - 1) \| \partial \phi_r^n \|_{L^2}^2. \end{split}$$

Taking the constant $\delta = 1$ yields

$$\nu \left(\nabla \phi_r^{n+1}, \nabla \partial \phi_r^n \right)_{L^2} \le \frac{1}{4} \| \partial \eta_{Ritz}^n \|_{L^2}^2.$$
(4.44)

Also,

$$\Delta t \left(\nabla \phi_r^{n+1}, \nabla \partial \phi_r^n \right)_{L^2} = \frac{1}{2} \left(\| \nabla \phi_r^{n+1} \|_{L^2}^2 + \| \nabla \phi_r^{n+1} - \nabla \phi_r^n \|_{L^2}^2 - \| \nabla \phi_r^n \|_{L^2}^2 \right).$$
(4.45)

Combining Equation (4.44) and Equation (4.45), summing over n = 1, ..., k - 1, and using $\|\nabla e^k\|_{L^2}^2 \le 2\left(\|\nabla \eta^k\|_{L^2}^2 + \|\nabla \phi_r^k\|_{L^2}^2\right)$ gives

$$\|\nabla e^k\|_{L^2}^2 \le C\left(\Delta t \sum_{n=1}^{k-1} \|\partial \eta_{Ritz}^n\|_{L^2}^2 + \|\eta_{Ritz}^k\|_{H^1_0}^2 + \|\phi_r^1\|_{H^1_0}^2\right).$$
(4.46)

Then apply Lemma 4.2.2 and Theorem 4.2.4 with $Y = H_0^1(\Omega)$ and $\pi_r = R_r$ to get

$$\|\nabla e^{k}\|_{L^{2}}^{2} \leq C\left(\sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1} \|\varphi_{i} - R_{r}\varphi_{i}\|_{L^{2}}^{2} + \sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1} \|\varphi_{i} - R_{r}\varphi_{i}\|_{H^{1}_{0}}^{2} + \|\phi_{r}^{1}\|_{H^{1}_{0}}^{2}\right).$$
(4.47)

Rearrange and take the maximum over all k to get the result.

Theorem 4.3.4. The pointwise solution norm error when the $L^2(\Omega)$ POD basis is used for the BE-POD-ROM is bounded by

$$\|e^{N}\|_{L^{2}}^{2} + \nu\Delta t \sum_{n=1}^{N-1} \|e^{n+1}\|_{H_{0}^{1}}^{2} \leq C \left(\sum_{i=r+1}^{s_{X}} \lambda^{DQ1} (\|\varphi_{i} - R_{r}\varphi_{i}\|_{L^{2}}^{2} + \|\varphi_{i} - R_{r}\varphi_{i}\|_{H_{0}^{1}}^{2}) + \|\phi_{r}^{1}\|_{L^{2}}^{2} \right).$$

$$(4.48)$$

Proof. Taking ϕ_r^{n+1} in Equation (4.33), yields

$$\left(\frac{\phi_r^{n+1} - \phi_r^n}{\Delta t}, \phi_r^{n+1}\right)_{L^2} + \nu \|\nabla \phi_r^{n+1}\|_{L^2}^2 = \left(\frac{\eta_{Ritz}^{n+1} - \eta_{Ritz}^n}{\Delta t}, \phi_r^{n+1}\right)_{L^2}.$$
(4.49)

Now, apply the Cauchy-Schwartz inequality, Young's inequality, and a polarization identity to get

$$\frac{1}{2\Delta t} \left(\|\phi_r^{n+1}\|_{L^2}^2 - \|\phi_r^n\|_{L^2}^2 \right) + \nu \|\nabla \phi_r^{n+1}\|_{L^2}^2 \le \frac{1}{4\delta} \|\partial \eta_{Ritz}^n\|_{L^2}^2 + \delta \|\phi_r^{n+1}\|_{L^2}^2.$$
(4.50)

Now multiply by 2, apply the Poincaré inequality to the last term, and combine the resulting like terms:

$$\frac{1}{\Delta t} \left(\|\phi_r^{n+1}\|_{L^2}^2 - \|\phi_r^n\|_{L^2}^2 \right) + 2\left(\nu - \delta C\right) \|\nabla \phi_r^{n+1}\|_{L^2}^2 \le \frac{1}{2\delta} \|\partial \eta_{Ritz}^n\|_{L^2}^2.$$
(4.51)

Take $\delta = \frac{v}{2C}$ and multiply by Δt :

$$\|\phi_r^{n+1}\|_{L^2}^2 - \|\phi_r^n\|_{L^2}^2 + \nu\Delta t \|\nabla\phi_r^{n+1}\|_{L^2}^2 \le \frac{C\Delta t}{\nu} \|\partial\eta_{Ritz}^n\|_{L^2}^2.$$
(4.52)

Sum from n = 1 to k - 1, rearrange, and take a maximum amoung the constants to obtain

$$\|\phi_r^k\|_{L^2}^2 + \nu\Delta t \sum_{n=1}^{k-1} \|\nabla\phi_r^{n+1}\|_{L^2}^2 \le C \left(\Delta t \sum_{n=1}^{k-1} \|\partial\eta_{Ritz}^n\|_{L^2}^2 + \|\phi_r^1\|_{L^2}^2\right).$$
(4.53)

Using $||e^n||_{L^2}^2 \le 2(||\eta^n||_{L^2}^2 + ||\phi_r^n||_{L^2}^2)$ and rearranging gives

$$\begin{aligned} \|e^{k}\|_{L^{2}}^{2} + \nu \Delta t \sum_{n=1}^{k-1} \|e^{n+1}\|_{H_{0}^{1}}^{2} \\ &\leq C \left(\Delta t \sum_{n=1}^{k-1} \|\partial \eta_{Ritz}^{n}\|_{L^{2}}^{2} + \|\eta_{Ritz}^{k}\|_{L^{2}}^{2} + \Delta t \sum_{n=1}^{k-1} \|\eta_{Ritz}^{n+1}\|_{H_{0}^{1}}^{2} + \|\phi_{r}^{1}\|_{L^{2}}^{2} \right). \end{aligned}$$
(4.54)

Finally, using Lemma 4.2.2, Theorem 4.2.4, and Corollary 4.2.5 with $\pi_r = R_r$ and both $Y = L^2(\Omega)$ and $Y = H_0^1(\Omega)$ yield

$$\|e^{k}\|_{L^{2}}^{2} + \nu\Delta t \sum_{n=1}^{k-1} \|e^{n+1}\|_{H_{0}^{1}}^{2} \leq C \left(\sum_{i=r+1}^{s_{X}} \lambda^{DQ1} (\|\varphi_{i} - R_{r}\varphi_{i}\|_{L^{2}}^{2} + \|\varphi_{i} - R_{r}\varphi_{i}\|_{H_{0}^{1}}^{2}) + \|\phi_{r}^{1}\|_{L^{2}}^{2} \right).$$

$$(4.55)$$

Taking k = N yields the result.

4.3.2. POD Space: $H_0^1(\Omega)$. Alternatively, we can take the POD space X to be $H_0^1(\Omega)$. The orthogonal projection onto $V_r^h, \Pi_r^X : H_0^1(\Omega) \to H_0^1(\Omega)$, is given by

$$\Pi_{r}^{X} u = \sum_{i=1}^{r} (u, \varphi_{i})_{H_{0}^{1}} \varphi_{i}$$
(4.56)

and the set of POD modes $\{\varphi_i\}$ are orthogonal in $H_0^1(\Omega)$. Note that

$$\Pi_r^X \varphi_k = \sum_{i=1}^r (\varphi_k, \varphi_i)_{H_0^1} \varphi_i = 0$$

for k > r since $\{\varphi_k\}$ are orthogonal in $H_0^1(\Omega)$. Further, since Π_r^X is orthogonal we have

$$(w - \Pi_r^X w, v_r)_X = 0 \quad \forall v_r \in V_r^h$$
$$\implies (w - \Pi_r^X w, v_r)_{H_0^1} = 0 \quad \forall v_r \in V_r^h$$
$$\implies (\nabla (w - \Pi_r^X w), \nabla v_r)_{L^2} = 0 \quad \forall v_r \in V_r^h.$$

Similarly to before we take π_r to be the Ritz projection, but in this case the Ritz projection is the orthogonal projection Π_r^X , and Equation (4.30) becomes

$$\left(\frac{\phi_r^{n+1} - \phi_r^n}{\Delta t}, v_r\right)_{L^2} + \nu (\nabla \phi_r^{n+1}, \nabla v_r)_{L^2} = \left(\frac{\eta_{Ritz}^{n+1} - \eta_{Ritz}^n}{\Delta t}, v_r\right)_{L^2} \quad \forall v_r \in V_r^h \tag{4.57}$$

where $\eta_{Ritz}^{n+1} = u^{n+1} - R_r u^{n+1} = u^{n+1} - \Pi_r^X u^{n+1}$. As before we can show error bounds for the L^2 and H_0^1 norm errors and the solution norm error.

Theorem 4.3.5. The ROM solution errors when the $H_0^1(\Omega)$ POD basis is used for the *BE-POD-ROM* are bounded as follows:

$$\max_{k} \|e^{k}\|_{L^{2}}^{2} \leq C \left(\sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1} \|\varphi_{i}\|_{L^{2}}^{2} + \|\phi_{r}^{1}\|_{L^{2}}^{2} \right),$$
(4.58)

$$\max_{k} \|\nabla e^{k}\|_{L^{2}}^{2} \leq C\left(\sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1}(\|\varphi_{i}\|_{L^{2}}^{2} + 1) + \|\phi_{r}^{1}\|_{H^{1}_{0}}^{2}\right),$$
(4.59)

and

$$\|e^{N}\|_{L^{2}}^{2} + \nu\Delta t \sum_{n=1}^{N-1} \|\nabla e^{n+1}\|_{L^{2}}^{2} \le C \left(\sum_{i=r+1}^{s_{X}} \lambda^{DQ1} (1 + \|\varphi_{i}\|_{L^{2}}^{2}) + \|\phi_{r}^{1}\|_{L^{2}}^{2} \right).$$
(4.60)

Proof. Proceed as in the proofs of Theorem 4.3.2, Theorem 4.3.3, and Theorem 4.3.4 respectively, but now use $\eta_{Ritz} = u - \prod_{r}^{X} u$ and the appropriate choices for *Y* in Theorem 4.2.4, Corollary 4.2.5, and Lemma 4.2.2.

4.3.3. Optimality. In this section we investigate the optimality of this new approach to POD with difference quotients. To do so we follow the approach given in [37] but modify the optimality definitions given there to include the ROM error for the initial condition. For this portion of the thesis we focus on both of the POD ROM discretization errors and assume the time and spatial discretization errors are optimal. Thus, we ignore the latter errors in the equation given below. In this section we consider only the new approach as the standard difference quotient approach is discussed in [37]. The optimality of each error depends on both the POD space *X* and the error norm space *Y*. To give a precise definition of pointwise ROM error optimality, assume there exists a constant C > 0 that is independent of the discretization parameters such that the ROM errors $e^k = u^k - u^k_r$ for k = 1, ..., N satisfy

$$\max_{1 \le k \le N} \|e^k\|_Y^2 \le C\left(\Lambda_r + \Lambda_r^1\right) \tag{4.61}$$

where

- Λ_r is the ROM discretization error, and depends only on *r*, the POD eigenvalues, and the POD modes;
- Λ_r^1 is the ROM discretization error for the initial condition, and depends only on *r*, the POD eigenvalues, and the POD modes.

Let $X_r \subset X$ be the span of the first *r* POD modes, and assume $X_r \subset Y$. Let $\Pi_r^X : X \to X$ be the orthogonal projection onto X_r and $\Pi_r^Y : Y \to Y$ be the *Y*-orthogonal projection onto X_r . Further let s_X be the number of positive POD eigenvalues. In Definition 4.3.6 we extend the definitions of optimality provided in [37] to include the ROM discretization error for the initial condition.

Definition 4.3.6. We say the total ROM discretization error, $\Lambda_r + \Lambda_r^1$, is

• optimal-I if there exists a constant C such that

$$\Lambda_r + \Lambda_r^1 \le C \sum_{i=r+1}^{s_X} \lambda_i \|\varphi_i\|_Y^2, \tag{4.62}$$

• optimal-II if there exists a constant C such that

$$\Lambda_r + \Lambda_r^1 \le C \sum_{i=r+1}^{s_X} \lambda_i \|\varphi_i - \Pi_r^Y \varphi_i\|_Y^2.$$
(4.63)

The constant C can depend on the solution data and the problem data, but does not depend on any discretization parameter.

Remark 4.3.7. For a detailed discussion on the optimality types see [37]. Note that, as shown in [37, Proposition 4.8], optimal-II is stronger than optimal-I, and the two are equivalent if X = Y.

First we consider the ROM error from the choice of initial condition by noting $\Lambda_r^1 = \|\phi_r^1\|_Y^2$ in Lemma 4.3.8 below.

Lemma 4.3.8. Let the initial condition be the POD projection of the given initial condition, i.e. $u_r^1 = \prod_r^X u^1$ so that $\phi_r^1 = u_r^1 - R_r u^1 = \prod_r^X u^1 - R_r u^1$. If $X = L^2(\Omega)$ then

$$\|\phi_r^1\|_{L^2}^2 \le 2\sum_{i=r+1}^{s_X} \lambda_i + 2\sum_{i=r+1}^{s_X} \lambda_i \|\varphi_i - R_r \varphi_i\|_{L^2}^2$$
(4.64)

and

$$\|\phi_r^1\|_{H_0^1}^2 \le 2\sum_{i=r+1}^{s_X} \lambda_i \|\varphi_i\|_{H_0^1}^2 + 2\sum_{i=r+1}^{s_X} \lambda_i \|\varphi_i - R_r \varphi_i\|_{H_0^1}^2.$$
(4.65)

If $X = H_0^1(\Omega)$, then

$$\phi_r^1 = 0 \tag{4.66}$$

for either choice of space Y.

Proof. First consider the case $X = L^2(\Omega)$. We have

$$\begin{split} \|\phi_r^1\|_Y^2 &= \|u_r^1 - R_r u^1\|_Y^2 \\ &= \|\Pi_r^X u^1 - R_r u^1\|_Y^2 \\ &\leq 2\|\Pi_r^X u^1 - u^1\|_Y^2 + 2\|u^1 - R_r u^1\|_{L^2}^2 \end{split}$$

Using Lemma 4.2.2 with $Y = X = L^2(\Omega)$ and $\pi_r = R_r$ gives Equation (4.64) and with $Y = H_0^1(\Omega)$ and $\pi_r = R_r$ gives Equation (4.65). Finally if $X = H_0^1(\Omega)$, then $\Pi_r^X = R_r$ and Equation (4.66) follows by definition of π_r .

This lemma allows us to investigate the total ROM discretization error in the following two theorems with the POD space taken to be L^2 and H_0^1 respectively.

Theorem 4.3.9. If the L^2 POD basis is used, i.e. $X = L^2(\Omega)$, then the following hold:

• The pointwise ROM error in Equation (4.34) with the error norm $Y = L^2$ is optimal-I if there exists a constant C such that

$$\|\varphi_i - R_r \varphi_i\|_{L^2} \le C \tag{4.67}$$

for $r + 1 \le i \le s_X$. Note that in this case optimal-I is equivalent to optimal-II.

• The pointwise ROM error in Equation (4.41) with error norm $Y = H_0^1$ is optimal-I.

Remark 4.3.10. In [37, Theorem 4.10 (iv)], when using the L^2 POD basis with $Y = H_0^1(\Omega)$ but not considering the initial condition error the ROM error was optimal-II. When including the initial condition as the POD projection of the given initial condition, we obtain the

Proof. For the first item

$$\begin{split} \Lambda_{r} + \Lambda_{r}^{1} &= \sum_{i=r+1}^{s_{X}} \lambda_{i} \|\varphi_{i} - R_{r}\varphi_{i}\|_{L^{2}}^{2} + \|\phi_{r}^{1}\|_{L^{2}}^{2} \\ &\leq \sum_{i=r+1}^{s_{X}} \lambda_{i} \|\varphi_{i} - R_{r}\varphi_{i}\|_{L^{2}}^{2} + 2\left(\sum_{i=r+1}^{s_{X}} \lambda_{i} + \sum_{i=r+1}^{s_{X}} \lambda_{i} \|\varphi_{i} - R_{r}\varphi_{i}\|_{L^{2}}^{2}\right) \\ &\leq C \sum_{i=r+1}^{s_{X}} \lambda_{i} \\ &= C \sum_{i=r+1}^{s_{X}} \lambda_{i} \|\varphi_{i}\|_{L^{2}}^{2} \end{split}$$

using Lemma 4.3.8, the assumption in Equation (4.67), Theorem 4.2.4, and the L^2 orthonormality of the POD basis.

For the second item use the Poincaré inequality to get,

$$\begin{split} \Lambda_r + \Lambda_r^1 &= \sum_{i=r+1}^{s_X} \lambda_i \|\varphi_i - R_r \varphi_i\|_{L^2}^2 + \sum_{i=r+1}^{s_X} \lambda_i \|\nabla(\varphi_i - R_r \varphi_i)\|_{L^2}^2 + \|\phi_r^1\|_{H_0^1}^2 \\ &\leq C \sum_{i=r+1}^{s_X} \lambda_i \|\varphi_i - R_r \varphi_i\|_{H_0^1}^2 + \|\phi_r^1\|_{H_0^1}^2. \end{split}$$

Lemma 4.3.8 and the orthogonality of the Ritz projection $R_r: H_0^1 \to H_0^1$ then yields

$$\Lambda_r + \Lambda_r^1 \le C \sum_{i=r+1}^{s_X} \lambda_i \|\varphi_i - R_r \varphi_i\|_{H_0^1}^2 + C \sum_{i=r+1}^{s_X} \lambda_i \|\varphi_i\|_{H_0^1}^2 \le C \sum_{i=r+1}^{s_X} \lambda_i \|\varphi_i\|_{H_0^1}^2.$$

Thus the error given in Equation (4.41) is optimal-I.

Theorem 4.3.11. If the H_0^1 POD basis is used, i.e. $X = H_0^1$, then the following results hold.

• The pointwise ROM error in Equation (4.58) with the error norm $Y = L^2$ is optimal-I.

• The pointwise ROM error in Equation (4.58) with error norm $Y = L^2$ is also optimal-II when

$$\|\varphi_i\|_Y \le C \|\varphi_i - \Pi_r^Y \varphi_i\|_Y \quad \text{for } r+1 \le i \le s_X.$$

• The pointwise ROM error in Equation (4.59) with the error norm $Y = H_0^1$ is optimal-I. Note that in this case optimal-I is equivalent to optimal-II.

The proof of this result is similar to that of [37, Theorem 4.10] since from Lemma 4.3.8 we know $\phi_r^1 = 0$ for both $Y = L^2$ and $Y = H_0^1$. We omit the details. We note that the optimality results for the new POD DQ approach in Theorem 4.3.9 and Theorem 4.3.11 are very similar to [37, Theorem 4.10] for the standard POD DQ approach. In fact, the only fundamental difference in the results here was caused by our choice of the initial condition and including this in the optimality definition, as discussed in Remark 4.3.10.

4.4. NUMERICAL RESULTS

We now turn our attention to computational results. In this section we use the same test problem as in Section 4.2.1 to compute ROM solution errors in order to compare the new POD approach to the existing standard approaches. We also compute the scaling factors present in the bounds of the theoretical results from Theorem 4.2.4 and Section 4.3.

4.4.1. ROM Comparisons for the Three POD Approaches. First we find the ROM errors for the new POD DQ approach at the final time, i.e., the errors at T = 1. We compute both the L^2 norm error and the $H_0^1(\Omega)$ norm error. For this computation we use 100 finite element nodes, 100 time steps and 3 different values for r. Note that in all tables below we report the square of the norms for consistency with previous results. For comparison we also show the final time errors for the standard POD approach and the standard DQ POD approach. First we consider the case where the POD space is taken to be $X = L^2(\Omega)$. The results for the L^2 and H_0^1 errors can be found in Table 4.2 and Table 4.3

POD Space	r value	Standard POD	Standard DQ POD	New DQ POD
L^2	4	1.639e-08	1.206e-07	1.141e-07
	6	1.257e-10	1.148e-09	1.090e-09
	8	9.486e-13	9.144e-12	8.722e-12
H_0^1	4	1.851e-08	1.350e-07	1.276e-07
	6	1.324e-10	1.218e-09	1.156e-09
	8	9.999e-13	9.778e-12	9.323e-12

Table 4.2. ROM errors for the L^2 Norm at the final time

Table 4.3. ROM errors for the H_0^1 Norm at the final time

POD Space	r value	Standard POD	Standard DQ POD	New DQ POD
L^2	4	1.936e-07	1.340e-06	1.269e-06
	6	2.615e-09	2.364e-08	2.245e-08
	8	2.264e-11	2.123e-10	2.024e-10
H_0^1	4	2.171e-07	1.494e-06	1.413e-06
	6	2.755e-09	2.506e-08	2.379e-08
	8	2.396e-11	2.263e-10	2.157e-10

respectively. In Table 4.4, we compute the solution norm squared errors for this reduced order model using the same computation parameters. This solution norm error at the final time is given by Equation (4.68):

$$\|e^{N}\|_{L^{2}}^{2} + \Delta t \sum_{n=1}^{N-1} \|\nabla e^{n+1}\|_{L^{2}}^{2}.$$
(4.68)

We also wish to compare the different POD approaches when we take the POD space to be $H_0^1(\Omega)$, i.e., $X = H_0^1(\Omega)$. All other parameters of the computation stay the same and the results can also be found in Table 4.2, Table 4.3, and Table 4.4. For both cases of the chosen POD space the errors behave similarly in all three approaches but are particularly close in the two approaches that utilize the difference quotients.

POD Space	r value	Standard POD	Standard DQ POD	New DQ POD
L^2	4	1.172e-07	2.441e-06	2.008e-06
	6	1.472e-11	4.373e-10	3.576e-10
	8	7.032e-16	2.227e-14	1.810e-14
H_0^1	4	1.135e-07	3.049e-06	2.525e-06
	6	1.467e-11	4.861e-10	3.996e-10
	8	6.984e-16	2.479e-14	2.032e-14

Table 4.4. ROM solution norm error Equation (4.68)

4.4.2. Pointwise Error Bounds. We want to compute the ratios generated by the theoretical results in Section 4.2.2. For these results we vary the number of time steps, while keeping everything else constant. This allows us to verify that the scaling factors are not dependent on the time step chosen. We performed similar experiments by varying other parameters and obtained similar results. We use 100 finite element nodes and an *r* value of 4. Both $X = L^2(\Omega)$ and $X = H_0^1(\Omega)$ are considered in this section. We use the data generated by the finite element method as described in Section 4.2.1 and compute the POD modes and POD singular values.

First, we consider the pointwise error bounds as in Theorem 4.2.4. The tables show the projection and the norm space Y that is used for each of the computations. Note that for these we have $\pi_r = R_r$ and either $Y = L^2(\Omega)$ or $Y = H_0^1(\Omega)$. For example, for the second result in Theorem 4.2.4 with $X = L^2(\Omega)$ and $Y = H_0^1(\Omega)$ we have

scaling factor =
$$\left(\max_{j} \|u^{j} - \Pi_{r}^{X} u^{j}\|_{Y}^{2}\right) / \left(\sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1} \|\varphi_{i}\|_{Y}^{2}\right).$$

Projection	Y	1/40	1/50	1/100	1/200	1/300
Π_r^X	$L^2(\Omega)$	6.0e-02	5.4e-02	4.0e-02	3.2e-02	2.9e-02
Π_r^X	$H_0^1(\Omega)$	5.6e-02	4.9e-02	3.3e-02	2.3e-02	1.8e-02
R_r	$L^2(\Omega)$	5.8e-02	5.2e-02	3.7e-02	2.9e-02	2.5e-02
R_r	$H^1_0(\Omega)$	5.5e-02	4.9e-02	3.2e-02	2.1e-02	1.6e-02

Table 4.5. Scaling factors for Theorem 4.2.4 as Δt changes with $X = L^2(\Omega)$

Table 4.6. Scaling factors for Theorem 4.2.4 as Δt changes with $X = H_0^1(\Omega)$

Projection	Y	1/40	1/50	1/100	1/200	1/300
Π_r^X	$L^2(\Omega)$	6.5e-02	5.8e-02	4.2e-02	3.1e-02	2.6e-02
Π_r^X	$H_0^1(\Omega)$	6.8e-02	6.2e-02	4.8e-02	4.0e-02	3.6e-02
R_r	$L^2(\Omega)$	6.5e-02	5.9e-02	4.3e-02	3.3e-02	2.8e-02
R_r	$H_0^1(\Omega)$	7.0e-02	6.4e-02	5.1e-02	4.4e-02	4.1e-02

The results for $X = L^2(\Omega)$ and $X = H_0^1(\Omega)$ are given in Table 4.5 and Table 4.6 respectively. We present the results with fixed values of r = 4 and 100 finite element nodes and varying values of Δt . Theoretically, we showed that the scaling factor should always be less that or equal to 2 (since T = 1 here). The computational results show that the scaling factor can be much less than that value.

4.4.3. ROM Error Bounds. Next we consider the reduced order model error bounds from Section 4.3. Again we use 100 finite element nodes and r = 4 with varying values of Δt . First define the following values:

$$\operatorname{err}_{1} = \max_{k} \|e^{k}\|_{L^{2}}^{2},$$
$$\operatorname{err}_{2} = \max_{k} \|e^{k}\|_{H^{1}_{0}}^{2},$$

and

$$\operatorname{err}_{3} = \|e^{N}\|_{L^{2}}^{2} + \Delta t \sum_{n=1}^{N-1} \|\nabla e^{n+1}\|_{L^{2}}^{2}.$$

Δt	1/40	1/50	1/100	1/200	1/300
C_1	5.8e-02	5.1e-02	3.7e-02	2.8e-02	2.5e-02
C_2	1.0e-05	1.4e-05	2.9e-05	4.0e-05	4.3e-05
C_3	1.9e-06	2.4e-06	4.1e-05	5.2e-06	5.3e-06
C_4	6.7e-02	6.1-e02	4.7e-02	4.0e-02	3.6e-02
C_5	6.5e-02	5.8e-02	4.1e-02	3.0e-02	2.6e-02
C_6	1.1e-02	9.3e-03	5.7e-03	3.8e-03	3.1e-03

Table 4.7. Scaling factors as Δt changes for ROM errors

For the first set of scaling factors defined in Equation (4.69), Equation (4.70), and Equation (4.71) below we have $X = L^2(\Omega)$. We also compute the scaling factors for the results that use $X = H_0^1(\Omega)$ as the POD space. These scaling factors for the $X = H_0^1(\Omega)$ case are given by Equations (4.72) to (4.74). For these computations we once again vary the time steps and keep all other parameters constant. The results for both cases of POD basis space can be found in Table 4.7.

$$C_{1} = \operatorname{err}_{1} / \left(\sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1} \|\varphi_{i} - R_{r}\varphi_{i}\|_{L^{2}}^{2} + \|\phi_{r}^{1}\|_{L^{2}}^{2} \right)$$
(4.69)

$$C_{2} = \operatorname{err}_{2} / \left(\sum_{i=r+1}^{s_{X}} \lambda_{i}^{DQ1} \left(\left\| \varphi_{i} - R_{r} \varphi_{i} \right\|_{L^{2}}^{2} + \left\| \nabla (\varphi_{i} - R_{r} \varphi_{i}) \right\|_{L^{2}}^{2} \right) + \left\| \phi_{r}^{1} \right\|_{H^{1}_{0}}^{2} \right)$$
(4.70)

$$C_{3} = \operatorname{err}_{3} / \left(\sum_{i=r+1}^{s_{X}} \lambda^{DQ1} (\|\varphi_{i} - R_{r}\varphi_{i}\|_{L^{2}}^{2} + \|\nabla(\varphi_{i} - R_{r}\varphi_{i})\|_{L^{2}}^{2}) + \|\phi_{r}^{1}\|_{L^{2}}^{2} \right)$$
(4.71)

$$C_4 = \operatorname{err}_1 / \left(\sum_{i=r+1}^{s_X} \lambda_i^{DQ1} \|\varphi_i\|_{L^2}^2 + \|\phi_r^1\|_{L^2}^2 \right)$$
(4.72)

$$C_5 = \operatorname{err}_2 / \left(\sum_{i=r+1}^{s_X} \lambda_i^{DQ1} (\|\varphi_i\|_{L^2}^2 + 1) + \|\phi_r^1\|_{H_0^1}^2 \right)$$
(4.73)

$$C_6 = \operatorname{err}_3 / \left(\sum_{i=r+1}^{s_X} \lambda_i^{DQ1} (1 + \|\varphi_i\|_{L^2}^2) + \|\phi_r^1\|_{L^2}^2 \right)$$
(4.74)

Note that changing the number of finite element nodes and keeping the number of time steps constant yields similar results. Theoretically we showed the scaling factors should remain bounded, and these computational results support that. For the computations in Section 4.4.2 and Section 4.4.3, we note that for larger values of r some of the projection errors, ROM errors, and error formulas become extremely small. In such cases some of the computed scaling factors are very large, but we believe this is caused by round off errors.

5. CONCLUSIONS

In this dissertation, we first prove new generalized error formulas for POD data approximation errors for both the discrete and continuous cases. We also show convergence of these errors under certain conditions and obtain new pointwise convergence results for POD projections. We demonstrate the application of our results to several example problems. We leave the application of these results to the numerical analysis of POD model order reduction methods for PDEs to be considered elsewhere.

Some open questions remain. When L^{-1} is unbounded, we have to assume uniform boundedness of the POD projections $\{L^{-1}\Pi_r^Y L\}$ to show that the error formula in (3.20) converges to zero as *r* increases. We do not know if there is a simpler condition that yields convergence of the approximation error. If *L* or L^{-1} is unbounded, we also do not know if the POD projections $\{L\Pi_r^X L^{-1}\}$ and $\{L^{-1}\Pi_r^Y L\}$ are uniformly bounded. Both of these issues have been discussed in the context of Example 2 in Section 3.5.2.2 in [1, 21]. The second issue has also been discussed in the context of Example 1 in Section 3.5.2.1 in [30, 39]; in these works, the H^1 stability of the L^2 POD projection is of interest.

Then, we introduce a new approach to POD using the difference quotients of the snapshot data. Specifically, we derive the POD modes from a data set that includes only the first snapshot and the regular difference quotients. This data set has approximately half the number of snapshots as the standard POD with DQ approach; also, this data set does not include redundant data when the snapshots are linearly independent. For this new approach to POD with DQs, we prove an approximation result for the weighted sum of the POD projection errors of the regular snapshots, and we also prove that we retain all of the numerical analysis benefits of using DQs that were shown in [37] for the standard POD with DQ approach. Our numerical experiments for a heat equation test problem show that this new approach produces similar reduced order model errors to other known POD approaches.

The latter part of this work focuses on numerical analysis results concerning pointwise POD projection errors and the optimality of pointwise in time ROM errors for our new approach to POD with DQs. Further investigation is needed to determine how this new approach compares to other standard POD methods in practical computations. The size of the ROM errors is not considered here and more research is needed to compare the size of the ROM error in this case with those of standard POD and standard DQ POD.

Also, we consider only one PDE and one choice of difference quotient. We focus on the heat equation and leave the Navier-Stokes equations and other more complicated PDEs to be considered elsewhere. Additionally, in this work we consider the difference quotients obtained when using backward Euler for time stepping. Other difference quotients are possible and have been used for snapshot collection in [31, 41, 46]. It is possible that results in this thesis can be extended to these other difference quotients, but we leave that to be considered elsewhere.

REFERENCES

- Singler, J. R., 'New POD error expressions, error bounds, and asymptotic results for reduced order models of parabolic PDEs,' SIAM J. Numer. Anal., 2014, 52(2), pp. 852–876, ISSN 0036-1429, doi:10.1137/120886947.
- [2] Eroglu, F. G., Kaya, S., and Rebholz, L. G., 'A modular regularized variational multiscale proper orthogonal decomposition for incompressible flows,' Comput. Methods Appl. Mech. Engrg., 2017, **325**, pp. 350–368, ISSN 0045-7825, doi: 10.1016/j.cma.2017.07.017.
- [3] Fahl, M., 'Computation of POD basis functions for fluid flows with Lanczos methods,' Mathematical and Computer Modelling, 2001, 34(1-2), pp. 91–107, ISSN 0895-7177, doi:10.1016/S0895-7177(01)00051-6.
- [4] Gräßle, C., Hinze, M., Lang, J., and Ullmann, S., 'POD model order reduction with space-adapted snapshots for incompressible flows,' Adv. Comput. Math., 2019, 45(5-6), pp. 2401–2428, ISSN 1019-7168, doi:10.1007/s10444-019-09716-7.
- [5] Higham, J. E., Shahnam, M., and Vaidheeswaran, A., 'Using a proper orthogonal decomposition to elucidate features in granular flows,' Granular Matter, 2020, 22(4), doi:10.1007/s10035-020-01037-7.
- [6] Gunzburger, M., Iliescu, T., and Schneier, M., 'A Leray regularized ensembleproper orthogonal decomposition method for parameterized convection-dominated flows,' IMA Journal of Numerical Analysis, April 2020, 40(2), pp. 886–913, doi: 10.1093/imanum/dry094.
- [7] Rowley, C. W., 'Model reduction for fluids, using balanced proper orthogonal decomposition,' Internat. J. Bifur. Chaos Appl. Sci. Engrg., 2005, 15(3), pp. 997–1013, ISSN 0218-1274.
- [8] Mohebujjaman, M., Rebholz, L. G., Xie, X., and Iliescu, T., 'Energy balance and mass conservation in reduced order models of fluid flows,' J. Comput. Phys., 2017, 346, pp. 262–277, ISSN 0021-9991.
- [9] Nguyen, V. B., Dou, H.-S., Willcox, K., and Khoo, B.-C., 'Model order reduction for reacting flows: Laminar Gaussian flame applications,' in '30th International Symposium on Shock Waves 1,' pp. 337–343, Springer International Publishing, 2017, doi:10.1007/978-3-319-46213-4_57.
- [10] Zerfas, C., Rebholz, L. G., Schneier, M., and Iliescu, T., 'Continuous data assimilation reduced order models of fluid flow,' Comput. Methods Appl. Mech. Engrg., 2019, 357, p. 112596, ISSN 0045-7825, doi:10.1016/j.cma.2019.112596.
- [11] Bergmann, M., Cordier, L., and Brancher, J.-P., 'Optimal rotary control of the cylinder wake using proper orthogonal decomposition reduced-order model,' Physics of Fluids, 2005, 17(9), p. 097101, doi:10.1063/1.2033624.

- [12] Freya, B., Dennis, B., Jianjie, L., and Stefan, V., 'POD-Based Mixed-Integer Optimal Control of the Heat Equation,' J. Sci. Comput., 2019, 81(1), pp. 48–75, ISSN 0885-7474, doi:10.1007/s10915-019-00924-3.
- [13] Alla, A. and Volkwein, S., 'Asymptotic stability of POD based model predictive control for a semilinear parabolic PDE,' Adv. Comput. Math., 2015, 41(5), pp. 1073–1102, ISSN 1019-7168, doi:10.1007/s10444-014-9381-0.
- [14] Gubisch, M. and Volkwein, S., 'Proper orthogonal decomposition for linear-quadratic optimal control,' in 'Model reduction and approximation,' volume 15 of *Comput. Sci. Eng.*, pp. 3–63, SIAM, Philadelphia, PA, 2017.
- [15] Holmes, P., Lumley, J. L., Berkooz, G., and Rowley, C. W., *Turbulence, coherent structures, dynamical systems and symmetry*, Cambridge Monographs on Mechanics, Cambridge University Press, Cambridge, second edition, 2012, ISBN 978-1-107-00825-0.
- [16] Liang, Y. C., Lee, H. P., Lim, S. P., Lin, W. Z., Lee, K. H., and Wu, C. G., 'Proper orthogonal decomposition and its applications. I. Theory,' J. Sound Vibration, 2002, 252(3), pp. 527–544, ISSN 0022-460X, doi:10.1006/jsvi.2001.4041.
- [17] Willcox, K. and Peraire, J., 'Balanced model reduction via the proper orthogonal decomposition,' AIAA Journal, 2002, 40(11), pp. 2323–2330, doi:10.2514/2.1570.
- [18] Kunisch, K. and Volkwein, S., 'Galerkin proper orthogonal decomposition methods for parabolic problems,' Numerische Mathematik, 2001, 90(1), pp. 117–148, ISSN 0029-599X, doi:10.1007/s002110100282.
- [19] Kunisch, K. and Volkwein, S., 'Galerkin proper orthogonal decomposition methods for a general equation in fluid dynamics,' SIAM J. Numer. Anal., 2002, 40(2), pp. 492–515, ISSN 0036-1429, doi:10.1137/S0036142900382612.
- [20] Alla, A., Falcone, M., and Volkwein, S., 'Error analysis for POD approximations of infinite horizon problems via the dynamic programming approach,' SIAM J. Control Optim., 2017, 55(5), pp. 3091–3115, ISSN 0363-0129, doi:10.1137/15M1039596.
- [21] Chapelle, D., Gariah, A., and Sainte-Marie, J., 'Galerkin approximation with proper orthogonal decomposition: new error estimates and illustrative examples,' ESAIM Math. Model. Numer. Anal., 2012, 46(4), pp. 731–757, ISSN 0764-583X, doi: 10.1051/m2an/2011053.
- [22] Gräßle, C. and Hinze, M., 'POD reduced-order modeling for evolution equations utilizing arbitrary finite element discretizations,' Adv. Comput. Math., 2018, 44(6), pp. 1941–1978, ISSN 1019-7168, doi:10.1007/s10444-018-9620-x.
- [23] Gunzburger, M., Jiang, N., and Schneier, M., 'An ensemble-proper orthogonal decomposition method for the nonstationary Navier-Stokes equations,' SIAM J. Numer. Anal., 2017, 55(1), pp. 286–304, ISSN 0036-1429.

- [24] Iliescu, T. and Wang, Z., 'Variational multiscale proper orthogonal decomposition: convection-dominated convection-diffusion-reaction equations,' Math. Comp., 2013, 82(283), pp. 1357–1378, ISSN 0025-5718, doi:10.1090/S0025-5718-2013-02683-X.
- [25] Iliescu, T. and Wang, Z., 'Variational multiscale proper orthogonal decomposition: Navier-Stokes equations,' Numer. Methods Partial Differential Equations, 2014, 30(2), pp. 641–663, ISSN 0749-159X, doi:10.1002/num.21835.
- [26] Li, R., Wu, Q., and Zhu, S., 'Proper orthogonal decomposition with SUPG-stabilized isogeometric analysis for reduced order modelling of unsteady convection-dominated convection-diffusion-reaction problems,' J. Comput. Phys., 2019, 387, pp. 280–302, ISSN 0021-9991, doi:10.1016/j.jcp.2019.02.051.
- [27] Rubino, S., 'A streamline derivative POD-ROM for advection-diffusion-reaction equations,' ESAIM: Proceedings and Surveys, 2018, 64, pp. 121–136, doi: 10.1051/proc/201864121.
- [28] Shen, J., Singler, J. R., and Zhang, Y., 'HDG-POD reduced order model of the heat equation,' J. Comput. Appl. Math., 2019, 362, pp. 663–679, ISSN 0377-0427, doi:10.1016/j.cam.2018.09.031.
- [29] Wang, Z., 'Nonlinear model reduction based on the finite element method with interpolated coefficients: semilinear parabolic equations,' Numer. Methods Partial Differential Equations, 2015, 31(6), pp. 1713–1741, ISSN 0749-159X.
- [30] Xie, X., Wells, D., Wang, Z., and Iliescu, T., 'Numerical analysis of the Leray reduced order model,' J. Comput. Appl. Math., 2018, 328, pp. 12–29, ISSN 0377-0427.
- [31] Zhu, S., Dedè, L., and Quarteroni, A., 'Isogeometric analysis and proper orthogonal decomposition for the acoustic wave equation,' ESAIM. Mathematical Modelling and Numerical Analysis, 2017, 51(4), pp. 1197–1221, ISSN 0764-583X, doi: 10.1051/m2an/2016056.
- [32] Kunisch, K. and Volkwein, S., 'Proper orthogonal decomposition for optimality systems,' M2AN Math. Model. Numer. Anal., 2008, 42(1), pp. 1–23, ISSN 0764-583X, doi:10.1051/m2an:2007054.
- [33] Luo, Z., Chen, J., Navon, I. M., and Yang, X., 'Mixed finite element formulation and error estimates based on proper orthogonal decomposition for the nonstationary Navier-Stokes equations,' SIAM J. Numer. Anal., 2008/09, 47(1), pp. 1–19, ISSN 0036-1429, doi:10.1137/070689498.
- [34] Rathinam, M. and Petzold, L. R., 'A new look at proper orthogonal decomposition,' SIAM Journal on Numerical Analysis, 2003, 41(5), pp. 1893–1925, ISSN 0036-1429, doi:10.1137/S0036142901389049.

- [35] Giere, S., Iliescu, T., John, V., and Wells, D., 'SUPG reduced order models for convection-dominated convection-diffusion-reaction equations,' Comput. Methods Appl. Mech. Engrg., 2015, 289, pp. 454–474, ISSN 0045-7825, doi: 10.1016/j.cma.2015.01.020.
- [36] Singler, J. R., 'Optimality of balanced proper orthogonal decomposition for data reconstruction II: further approximation results,' J. Math. Anal. Appl., 2015, 421(2), pp. 1006–1020, ISSN 0022-247X, doi:10.1016/j.jmaa.2014.07.059.
- [37] Koc, B., Rubino, S., Schneier, M., Singler, J. R., and Iliescu, T., 'On optimal pointwise in time error bounds and difference quotients for the proper orthogonal decomposition,' SIAM Journal on Numerical Analysis, 2020, to appear, arXiv:2010.03750.
- [38] Iliescu, T. and Wang, Z., 'Are the snapshot difference quotients needed in the proper orthogonal decomposition?' SIAM J. Sci. Comput., 2014, 36(3), pp. A1221–A1250, ISSN 1064-8275, doi:10.1137/130925141.
- [39] Kean, K. and Schneier, M., 'Error analysis of supremizer pressure recovery for POD based reduced-order models of the time-dependent Navier-Stokes equations,' SIAM Journal on Numerical Analysis, 2020, 58(4), pp. 2235–2264, ISSN 0036-1429, doi: 10.1137/19M128702X.
- [40] Kostova-Vassilevska, T. and Oxberry, G. M., 'Model reduction of dynamical systems by proper orthogonal decomposition: error bounds and comparison of methods using snapshots from the solution and the time derivatives,' J. Comput. Appl. Math., 2018, 330, pp. 553–573, ISSN 0377-0427.
- [41] Jin, B. and Zhou, Z., 'An analysis of Galerkin proper orthogonal decomposition for subdiffusion,' ESAIM. Mathematical Modelling and Numerical Analysis, 2017, 51(1), pp. 89–113, ISSN 0764-583X, doi:10.1051/m2an/2016017.
- [42] Leibfritz, F. and Volkwein, S., 'Numerical feedback controller design for PDE systems using model reduction: techniques and case studies,' in 'Real-time PDE-constrained optimization,' volume 3 of *Comput. Sci. Eng.*, pp. 53–72, SIAM, Philadelphia, PA, 2007, doi:10.1137/1.9780898718935.ch3.
- [43] Sachs, E. W. and Schu, M., 'A priori error estimates for reduced order models in finance,' ESAIM. Mathematical Modelling and Numerical Analysis, 2013, 47(2), pp. 449–469, ISSN 0764-583X, doi:10.1051/m2an/2012039.
- [44] Hömberg, D. and Volkwein, S., 'Control of laser surface hardening by a reducedorder approach using proper orthogonal decomposition,' Mathematical and Computer Modelling, 2003, 38(10), pp. 1003–1028, ISSN 0895-7177, doi:10.1016/S0895-7177(03)90102-6.
- [45] Hijazi, S., Stabile, G., Mola, A., and Rozza, G., 'Data-driven POD-Galerkin reduced order model for turbulent flows,' Journal of Computational Physics, 2020, 416, pp. 109513, 30, ISSN 0021-9991, doi:10.1016/j.jcp.2020.109513.

- [46] Herkt, S., Hinze, M., and Pinnau, R., 'Convergence analysis of Galerkin POD for linear second order evolution equations,' Electronic Transactions on Numerical Analysis, 2013, 40, pp. 321–337.
- [47] Locke, S. and Singler, J., 'New proper orthogonal decomposition approximation theory for PDE solution data,' SIAM Journal on Numerical Analysis, 2020, 58(6), pp. 3251– 3285, ISSN 0036-1429, doi:10.1137/19M1297002.
- [48] Locke, S. and Singler, J., 'A new approach to proper orthogonal decomposition with difference quotients,' arXiv:2106.10224, 2021.
- [49] Djouadi, S. M., 'On the optimality of the proper orthogonal decomposition and balanced truncation,' in 'Proceedings of the 47th IEEE Conference on Decision and Control,' 2008 pp. 4221 –4226.
- [50] Volkwein, S., 'Interpretation of proper orthogonal decomposition as singular value decomposition and HJB-based feedback design,' in 'Proceedings of the Sixteenth International Symposium on Mathematical Theory of Networks and Systems (MTNS),' 2004.
- [51] Quarteroni, A., Manzoni, A., and Negri, F., *Reduced basis methods for partial differential equations*, volume 92 of *Unitext*, Springer, Cham, 2016, ISBN 978-3-319-15430-5; 978-3-319-15431-2.
- [52] Gohberg, I., Goldberg, S., and Kaashoek, M. A., *Classes of Linear Operators. Vol. I*, volume 49 of *Operator Theory: Advances and Applications*, Birkhäuser Verlag, Basel, 1990, ISBN 3-7643-2531-3, doi:10.1007/978-3-0348-7509-7.
- [53] Kato, T., Perturbation theory for linear operators, Classics in Mathematics, Springer-Verlag, Berlin, 1995, ISBN 3-540-58661-X, reprint of the 1980 edition.
- [54] Lax, P. D., *Functional analysis*, Pure and Applied Mathematics (New York), Wiley-Interscience [John Wiley & Sons], New York, 2002, ISBN 0-471-55604-1.
- [55] Reed, M. and Simon, B., *Methods of modern mathematical physics I: Functional analysis*, Academic Press, Inc., New York, second edition, 1980, ISBN 0-12-585050-6.
- [56] Dunford, N. and Schwartz, J. T., *Linear Operators. I. General Theory*, With the assistance of W. G. Bade and R. G. Bartle. Pure and Applied Mathematics, Vol. 7, Interscience Publishers, Inc., New York; Interscience Publishers, Ltd., London, 1958.
- [57] Miklavčič, M., Applied functional analysis and partial differential equations, World Scientific Publishing Co., Inc., River Edge, NJ, 1998, ISBN 981-02-3535-6, doi: 10.1142/9789812796233.
- [58] Ciarlet, P. G., *Linear and nonlinear functional analysis with applications*, Society for Industrial and Applied Mathematics, Philadelphia, PA, 2013, ISBN 978-1-611972-58-0.

- [59] Balakrishnan, A. V., *Applied functional analysis*, Springer-Verlag, New York-Heidelberg, 1976.
- [60] Aubin, J.-P., *Applied functional analysis*, Pure and Applied Mathematics (New York), Wiley-Interscience, New York, second edition, 2000, ISBN 0-471-17976-0, doi: 10.1002/9781118032725.
- [61] Singler, J. R., 'Convergent snapshot algorithms for infinite-dimensional Lyapunov equations,' IMA J. Numer. Anal., 2011, 31(4), pp. 1468–1496, ISSN 0272-4979, doi:10.1093/imanum/drq028.
- [62] Gohberg, I. C. and Kreĭn, M. G., *Introduction to the theory of linear nonselfadjoint operators*, American Mathematical Society, Providence, R.I., 1969.
- [63] Temam, R., Infinite-dimensional dynamical systems in mechanics and physics, volume 68 of Applied Mathematical Sciences, Springer-Verlag, New York, second edition, 1997, ISBN 0-387-94866-X, doi:10.1007/978-1-4612-0645-3.
- [64] Hinze, M. and Volkwein, S., 'Proper orthogonal decomposition surrogate models for nonlinear dynamical systems: Error estimates and suboptimal control,' in 'Lecture Notes in Computational Science and Engineering,' pp. 261–306, Springer-Verlag, 2005, doi:10.1007/3-540-27909-1_10.
- [65] Gu, H., Xin, J., and Zhang, Z., 'Error Estimates for a POD Method for Solving Viscous G-Equations in Incompressible Cellular Flows,' SIAM Journal on Scientific Computing, 2021, 43(1), pp. A636–A662, ISSN 1064-8275, doi:10.1137/19M1241854.
- [66] Fareed, H., Singler, J. R., Zhang, Y., and Shen, J., 'Incremental proper orthogonal decomposition for PDE simulation data,' Computers & Mathematics with Applications. An International Journal, 2018, 75(6), pp. 1942–1960, ISSN 0898-1221, doi:10.1016/j.camwa.2017.09.012.
- [67] Baker, C. G., Gallivan, K. A., and Van Dooren, P., 'Low-rank incremental methods for computing dominant singular subspaces,' Linear Algebra and its Applications, 2012, 436(8), pp. 2866–2888, ISSN 0024-3795, doi:10.1016/j.laa.2011.07.018.
- [68] Brand, M., 'Fast low-rank modifications of the thin singular value decomposition,' Linear Algebra and its Applications, 2006, 415(1), pp. 20–30, ISSN 0024-3795, doi:10.1016/j.laa.2005.07.021.
- [69] Himpe, C., Leibner, T., and Rave, S., 'Hierarchical approximate proper orthogonal decomposition,' SIAM Journal on Scientific Computing, 2018, 40(5), pp. A3267– A3292, ISSN 1064-8275, doi:10.1137/16M1085413.
- [70] Fareed, H. and Singler, J. R., 'Error analysis of an incremental proper orthogonal decomposition algorithm for PDE simulation data,' Journal of Computational and Applied Mathematics, 2020, 368, pp. 112525, 14, ISSN 0377-0427, doi: 10.1016/j.cam.2019.112525.

Sarah Katherine Locke was born in Tennessee. She graduated from Richland High School in Lynnville, Tennessee in May of 2011. She started at the University of Tennessee-Martin in Martin, Tennessee in August of 2011 and graduated with a Bachelor of Science degree in Mathematics with a minor in Agriculture in May of 2016. That fall she attended Missouri University of Science and Technology as a Chancellor's Distinguished Fellow. She graduated with her Doctor of Philosophy in Mathematics in July 2021 from Missouri University of Science and Technology.